UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report: March 22, 2010 (Date of earliest event reported)

Timberline Resources Corporation

(Exact name of registrant as specified in its charter)

Commission File Number: 001-34055

Delaware

(State or other jurisdiction of incorporation)

82-0291227

(IRS Employer Identification No.)

101 East Lakeside Avenue Coeur d'Alene, Idaho 83814

(Address of principal executive offices, including zip code)

(208) 664-4859

(Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

[] Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
[] Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
[] Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
[] Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01 Entry into a Material Definitive Agreement

On March 22, 2010, the Registrant, Timberline Resources Corporation ("Timberline"), entered into a definitive agreement (the "Agreement"), whereby Timberline will acquire, through a court-approved plan of arrangement, all of the issued and outstanding common shares of Staccato Gold Resources Ltd. ("Staccato"), a Canadian corporation, by means of a share exchange (the "Arrangement").

Under the Agreement, Staccato shareholders will receive one share of Timberline common stock and \$0.0001 for every seven shares of Staccato common stock held (the "Exchange Ratio"). Pursuant to the Arrangement, the holders of a portion of the outstanding Staccato options and the holders of all Staccato warrants will be entitled to receive Timberline options and warrants at the Exchange Ratio and with corresponding changes to the exercise price based on the Exchange Ratio. The balance of the outstanding Staccato options will either be exchanged for Staccato shares prior to the closing or cancelled at closing.

Upon completion of the Arrangement, Timberline will have approximately 56 million common shares issued and outstanding, with 73 million shares on a fully diluted basis. Current Timberline shareholders will own approximately 74% of the shares outstanding and former Staccato shareholders will own approximately 26%. On a fully diluted basis, Timberline will be owned 71% by current Timberline shareholders and 29% by current Staccato shareholders. In conjunction with this Arrangement, Timberline will seek a listing of its shares of commons stock on the TSX Venture Exchange with the intent to have a listing in Toronto in addition to the NYSE Amex listing.

The closing is subject to a favorable vote of at least two-thirds of Staccato common stockholders voting at a special meeting of shareholders to be held on May 14, 2010; a favorable vote of a majority of Timberline common stockholders at a special meeting of shareholders to be held on May 21, 2010; as well as the approval of the NYSE Amex and the TSX Venture Exchange, and other customary closing conditions, including the receipt of all necessary court and regulatory approvals.

The Arrangement has been unanimously approved by the Board of Directors of both Staccato and Timberline and the management and directors of both companies have entered into support agreements. The Agreement contains a commitment from Staccato not to solicit or initiate discussions concerning alternative transactions to the proposed Arrangement. There will be no change in the management or Board of Directors of Timberline following the completion of the Arrangement.

The Agreement provides for the payment of reciprocal break fees or expense reimbursements under certain conditions. Timberline will pay a break fee to Staccato should the Agreement be terminated by Staccato as a result of: (i) a material breach of covenants contained in the Agreement or (ii) the Timberline Board of Directors withdrawing its recommendation of the Arrangement, and Timberline will reimburse Staccato's expenses if the Agreement is terminated by Staccato as a result of the shareholders of Timberline not approving the Arrangement. Staccato will pay a break fee to Timberline should the Agreement be terminated by Timberline as a result of: (i) a material breach of covenants contained in the Agreement , (ii) the Staccato Board of Directors withdraws its recommendation of the Arrangement, (iii) the Staccato Board of Directors recommends a competing proposal, (iv) the Board of Staccato fails to re-affirm the Arrangement following a request by Timberline to do so, (v) Staccato enters into a letter of intent or a term sheet for a competing proposal, (vi) Staccato breaches its non-solicitation obligations under the Agreement, or (vii) Staccato fails to recommend rejection for a tender offer or exchange offer relating to its securities, or should the Agreement be terminated by Staccato as a result of the Staccato Board of Directors accepting a competing proposal to the Arrangement, and Staccato will reimburse Timberline's expenses if the Agreement is terminated by Timberline as a result of the shareholders of Staccato not approving the Arrangement.

The foregoing is a description of the material terms of the Agreement and is qualified in its entirety by the actual terms and conditions of the Agreement, which is attached hereto as Exhibit 10.1 and is incorporated herein by reference.

Item 5.07 Submission of Matters to a Vote of Security Holders

On Monday, March 22, 2010, the Company held its 2010 Annual Meeting and the following are the final voting tallies for the Annual Meeting:

Proposal #1 – Election of Directors	Voted For	Withheld	
The election of the Nominees to the Company's Board to s			
Company's 2011 Annual Meeting of Shareholders or until	successors are		
duly elected and qualified:			
Paul Dircksen		15,756,659	29,277
Randal Hardy		15,506,159	279,777
Vance Thornsberry		11,689,008	4,096,928
Eric Klepfer		15,673,644	112,292
Ron Guill		15,739,344	46,592
James Moore		15,741,159	44,777
Robert Martinez		15,746,660	39,276
Proposal #2	For	Against	Abstain
To ratify the appointment of the Company's Independent	27,465,898	25,491	44,004
Registered Public Accounting Firm for the 2010 fiscal year			

Proxies were solicited under the proxy statement filed with the Securities and Exchange Commission on January 25, 2010. All nominees for director were elected.

Item 7.01 Regulation FD Disclosure.

On March 23, 2010, Timberline Resources Corporation (the "Company") issued a press release entitled "Timberline Agrees to Acquire Staccato Gold with Potential Near-Term Nevada Gold Production at Lookout Mountain". A copy of the press release is attached as Exhibit 99.1 and incorporated herein by reference.

In accordance with General Instruction B.2 of Form 8-K, the information in this report, including the exhibits attached hereto, shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended ("the Exchange Act"), nor shall such information be deemed incorporated by reference in any filing under the Securities Act of 1933, as amended, or the Exchange Act, except as shall be expressly set forth by specific reference in such a filing.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

<u>Exhibit No.</u>	<u>Description</u>
10.1	Arrangement Agreement between Timberline and Staccato dated March 22, 2010
10.2	Form of Support Agreement between Timberline and officers and directors of Staccato
99.1	Press Release of Timberline Resources Corporation dated March 23, 2010.*

^{*} Furnished to not filed with the SEC pursuant to Item 7.01 above.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

TIMBERLINE RESOURCES CORPORATION

Date: March 24, 2010

By: /s/ Randal Hardy

Randal Hardy

Chief Executive Officer, Chief Financial Officer

and Director

EXHIBIT INDEX

Exhibit No.	<u>Description</u>
10.1	Arrangement Agreement between Timberline and Staccato dated March 22, 2010
10.2	Form of Support Agreement between Timberline and officers and directors of Staccato
99.1	Press Release of Timberline Resources Corporation dated March 23, 2010.*

^{*} Furnished to not filed with the SEC pursuant to Item 7.01 above.

Exhibit 10.1

ARRANGEMENT AGREEMENT DATED for reference the 22nd day of March, 2010.

BETWEEN:

<u>TIMBERLINE RESOURCES CORPORATION</u>, a corporation duly organized under the laws of Delaware and having an office at 101 E. Lakeside Avenue, Coeur d'Alene, Idaho, 83814

(hereinafter called "Timberline")

OF THE FIRST PART

AND:

STACCATO GOLD RESOURCES LTD., a company duly organized under the laws of the Province of British Columbia and having an office at 2060-777 Hornby Street, Vancouver, British Columbia, V6Z 1T7

(hereinafter called "Staccato")

OF THE SECOND PART

WHEREAS THE PARTIES HAVE AGREED THAT:

- A. The respective boards of directors of Timberline and Staccato have approved the combination of Timberline and Staccato pursuant to the Plan of Arrangement (as defined below), providing for, among other things, the purchase by Timberline of all of the issued and outstanding Staccato Securities (as defined below).
- B. The board of directors of Timberline has unanimously determined that the Arrangement is in the best interests of Timberline and is fair to the Timberline Shareholders (as defined below).
- C. The board of directors of Staccato has unanimously determined that the Arrangement is in the best interests of Staccato and is fair to the Staccato Securityholders (as defined below).
- D. Subject to the provisions hereof, the board of directors of Timberline has unanimously recommended that the Timberline Shareholders approve the Arrangement at the Timberline Meeting (as defined below).
- E. Subject to the provisions hereof, the board of directors of Staccato has unanimously recommended that the Staccato Securityholders approve the Arrangement at the Staccato Meeting (as defined below).
- F. Pursuant to a plan of arrangement under Section 288 of the BCBCA (the "Plan of Arrangement"), the securityholders of Staccato will receive Timberline Securities (as defined below) in consideration of the acquisition by Timberline of their Staccato Securities (the "Arrangement").
- G. Timberline will apply to have the Timberline Shares issued pursuant to the Arrangement listed for trading on the NYSE Amex Equities ("Amex") and Timberline will apply to have its shares of common stock listed on the TSX Venture Exchange (the "TSXV").

THEREFORE THIS AGREEMENT WITNESSES that in consideration of the premises and of the mutual covenants and agreements hereinafter set forth, the parties hereto agree each with the other as follows:

1.0 <u>DEFINITIONS</u>

- 1.1 In this Agreement:
- (a) all capitalized terms which are not otherwise defined in this Agreement shall have the meaning ascribed to them in the Plan of Arrangement;
- (b) "Acquisition Proposal" means, other than the business combination transaction contemplated by this Agreement, any offer or proposal for a merger, amalgamation, arrangement, reorganization, share exchange, consolidation, recapitalization, liquidation, dissolution or other business combination involving Staccato or the acquisition or purchase of 20% or more of any class of equity securities of Staccato, or any take-over bid or tender offer (including issuer bids and self-tenders) or exchange offer that if consummated would result in any Person beneficially owning 20% or more of any class of any equity securities of Staccato, or any transaction involving the sale, lease, license or other disposition (by sale, merger or otherwise) of 20% or more of the book or market value of assets (including, without limitation, securities of any subsidiary of Staccato) of Staccato and its subsidiaries, taken as a whole;
- (c) "Amex" has the meaning ascribed thereto in Recital G;
- (d) "**Arrangement**" has the meaning ascribed thereto in Recital F;
- (e) "BCBCA" means the *Business Corporations Act*, S.B.C. 2002, c. 57, including all regulations made thereunder, as amended;
- (f) "Benefit Plan" means a pension, retirement, profit sharing, bonus, savings, deferred compensation, stock option, purchase, appreciation, group insurance or other material employee or retiree benefit plan, programme or arrangement, formal or informal, registered, unregistered or supplementary, oral or written, maintained or contributed to, or required to be contributed to, in the case of Timberline, by or on behalf of Timberline or a subsidiary of Timberline or in respect of which Timberline or a subsidiary of Timberline has an actual, actuarial or contingent liability and, in the case of Staccato, by or on behalf of Staccato or a subsidiary of Staccato or in respect of which Staccato or a subsidiary of Staccato has an actual, actuarial or contingent liability;
- (g) "business day" means a day that is not a Saturday, Sunday or civic or statutory holiday in Vancouver, British Columbia;
- (h) "Canadian Securities Administrators" means, collectively, the securities regulators in each of the provinces of Canada;
- (i) "Canadian GAAP" means accounting principles generally accepted in Canada from time to time;
- (j) "Canadian Securities Legislation" means, collectively, and as the context may require, the securities legislation of each of the provinces and territories of Canada, and the rules, regulations and policies published and/or promulgated thereunder, as such may be amended from time to time prior to the Effective Date;
- (k) "Code" means the United States Internal Revenue Code of 1986, as amended;
- (l) **"Competition Act"** means the *Competition Act* (Canada), as amended;

- (m) "Closing" has the meaning ascribed thereto in Section 6.3;
- (n) "Closing Date" has the meaning ascribed thereto in Section 6.3;
- (o) "Court" means the Supreme Court of British Columbia;
- (p) "Effective Date" has the meaning thereto ascribed in the Plan of Arrangement;
- (q) "Effective Time" means 12:01 a.m. (Pacific Time) on the Effective Date;
- (r) "**Final Order**" means the final order to be made by the Court approving the Arrangement as provided for in Section 2.2;
- (s) "Governmental Entity" means any (i) multinational, federal, provincial, territorial, state, regional, municipal, local or other government, governmental or public department, central bank, court, tribunal, arbitral body, commission, board, bureau or agency, domestic or foreign, (ii) subdivision, agent, commission, board, or authority of any of the foregoing, or (iii) quasi-governmental or private body exercising any regulatory, expropriation or taxing authority under or for the account of any of the foregoing;
- (t) "HSR Act" means the United States Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended
- (u) "Law" means all statutes, regulations, statutory rules, policies, orders, and terms and conditions of any grant of approval, permission, authority or license of any court, Governmental Entity, statutory body or regulatory authority (including the SEC, AMEX, and TSXV), and the term "applicable" with respect to such Law and in the context that refers to one or more Persons, means that such Law applies to such Person or Persons or its or their business, undertaking, property or securities and emanates from a Governmental Entity, statutory body or regulatory authority having jurisdiction over the Person or Persons or its or their business, undertaking, property or securities;
- (v) "Lien" means any mortgage, hypothec, prior claim, lien, pledge, assignment for security, security interest, lease, option, right of third parties or other charge or encumbrance, including the lien or retained title of a conditional vendor, and any easement, servitude, right of way or other encumbrance on title to real or immovable property or personal or movable property;
- (w) "Material Adverse Change" means any change (or any condition, event or development involving a prospective change) in the business, operations, results of operations, assets, capitalization, financial condition, licenses, permits, concessions, rights, liabilities, prospects or privileges, whether contractual or otherwise, of Timberline or Staccato (as applicable) which has a Material Adverse Effect on Timberline or Staccato (as applicable);
- "Material Adverse Effect" means any effect that is, or would reasonably be expected to be, material and adverse to the business, operations, results of operations, assets, prospects, capitalization or financial condition of Timberline or Staccato (as applicable) or that would prevent or materially impair or delay the ability of Timberline or Staccato (as applicable) to consummate the transactions contemplated by this Agreement (including the Plan of Arrangement) and any agreements contemplated hereunder; but "Material Adverse Effect" shall not include an effect resulting from (i) an action taken by Timberline or Staccato (as applicable) to which the other party consented to in writing; (ii) conditions affecting the mineral exploration industry generally in the jurisdictions in which Timberline or Staccato (as applicable) holds its assets, taken as a whole including, without limitation, changes in commodity prices or Taxes; or (iii) general economic, financial, currency exchange, securities or commodity market conditions

- in Canada and the United States including, without limitation, changes in currency exchange rates;
- (y) "material fact", "material change" and "misrepresentation" have the meanings ascribed to them by the *Securities Act* (British Columbia);
- (z) "Mineral Properties" means, with respect to a Party, the natural or mineral resource or exploration properties of the Party or its subsidiaries, and for greater certainty includes any mines or development projects in which the Party or its subsidiaries has an interest;
- (aa) "Owned Real Property" means, with respect to a Party, the real property owned by the Party or its subsidiaries that is material to the conduct of the business of the Party or its subsidiaries;
- (bb) "Party" means either Timberline or Staccato, as the case may be;
- (cc) "**Permitted Liens**" means, with respect to a Party and its subsidiaries:
 - (a) conflicts and overlaps of unpatented mining claims on fee lands; conflicting unpatented mining claims for which no maps were filed as required by law; the possibility that the boundaries of unpatented mining claims may not be marked on the ground; the possibility that unpatented mining claims may not have been located on open public lands; determinations whether unpatented mining claims should be located in the form of lodes or placers; rights granted to third parties by governmental entities or by statute to use the surface of lands covered by unpatented mining claims; and rights that would have been shown by a careful physical inspection or by an accurate survey of unpatented mining claims:
 - (b) any rights-of-way or other rights granted by Governmental Entities;
 - rights vested in or exercisable by Indigenous peoples including such rights arising out of the designation of lands in the U.S. as Indian Country;
 - (d) inchoate or statutory liens for Taxes not at the time overdue, and inchoate or statutory liens for overdue Taxes the validity of which the Party or its Subsidiaries owing such Taxes is contesting in good faith but only for so long as such contestation effectively postpones enforcement of any such liens or Taxes;
 - (e) statutory liens incurred or deposits made in the ordinary course of the business of the Party and its subsidiaries in connection with workers' compensation, unemployment insurance and similar legislation, but only to the extent that each such statutory lien or deposit relates to amounts not yet due;
 - (f) liens and privileges arising out of any judgment with respect to which the Party or any of its subsidiaries intends to prosecute an appeal or proceedings for review but only for so long as there is a stay of execution pending the determination of such appeal or proceedings for review;
 - (g) security given by the Party or any of its Subsidiaries to a public utility or any Governmental Entity when required in the ordinary course of business of the Party and its subsidiaries;
 - (h) undetermined or inchoate construction or repair or storage liens arising in the ordinary course of the business of the Party and its subsidiaries, a claim for which has not been

- filed or registered pursuant to law or of which notice in writing has not been given to the Party or its subsidiaries;
- (i) any reservations or exceptions contained in the original grants from the Crown or government, as applicable, and the paramount title of the government to lands covered by unpatented mining claims;
- (j) easements, including rights of way for, or reservations or rights of others relating to, sewers, water lines, gas lines, pipelines, electric lines, telegraph and telephone lines and other similar products or services and any registered restrictions or covenants that run with the land, provided that there has been compliance with the provisions thereof and that they do not in the aggregate materially detract from the value of the Owned Real Property and will not materially and adversely affect the ability of the Party and its subsidiaries to carry on their business as it has been carried on in the past; and
- (k) zoning by laws, ordinances or other restrictions as to the use of real property, and agreements with other Persons registered against title to the Owned Real Property, provided that they do not in the aggregate materially detract from the value of the Owned Real Property and will not materially and adversely affect the ability of the Party and its subsidiaries to carry on their business as it has been carried on in the past.
- (dd) "**Person**" includes any individual, firm, partnership, joint venture, venture capital fund, limited liability company, unlimited liability company, association, trust, trustee, executor, administrator, legal personal representative, estate, group, body corporate, corporation, unincorporated association or organization, Governmental Entity, syndicate or other entity, whether or not having legal status;
- (ee) "Plan of Arrangement" means the plan of arrangement under Section 288 of the BCBCA attached hereto as Schedule "A":
- (ff) "Real Property Leases" means with respect to a Party the leases, subleases, easements and other agreements under which the Party or any of its subsidiaries uses or occupies or has the right to use or occupy, now or in the future, any real property that is material to the conduct of the business of the Party or its subsidiaries;
- (gg) "Registrar" means the Registrar of Companies appointed pursuant to the BCBCA;
- (hh) "SEC" means the United States Securities and Exchange Commission;
- (ii) "Section 3(a)(10) Exemption" has the meaning ascribed thereto in Section 2.3;
- (jj) "Securities Administrators" means, collectively, the Canadian Securities Administrators and the U.S. Securities Administrators:
- (kk) "Securities Legislation" means the Canadian Securities Legislation, U.S. Securities Law and the published instruments and rules of any Governmental Entity administering those statutes, as well as the rules, regulations, by laws and policies of all applicable Securities Administrators, the TSXV and AMEX;
- (ll) "Staccato Break Fee" has the meaning ascribed thereto in Section 8.2;
- (mm) "Staccato Expense Reimbursement Fee" has the meaning ascribed thereto in Section 8.5;

- (nn) "Staccato Information Circular" means the information circular to be sent to the Staccato Securityholders in connection with the Staccato Meeting;
- (oo) "Staccato Material Subsidiary" has the meaning ascribed thereto in Subsection 3.2(a);
- (pp) "Staccato Meeting" has the meaning ascribed thereto in the Plan of Arrangement;
- (qq) "Staccato Options" means options to purchase Staccato Shares granted under Staccato's stock option plan;
- (rr) "Staccato Public Record" has the meaning ascribed thereto in Subsection 3.2(1);
- (ss) "Staccato Securities" means the Staccato Options, Staccato Shares, Staccato Warrants and other securities convertible into Staccato Shares;
- (tt) "Staccato Securityholders" means the holders of Staccato Securities;
- (uu) "Staccato Shares" means the common shares without par value in the capital of Staccato;
- (vv) "Staccato Shareholders" means the holders of Staccato Shares;
- (ww) "Staccato Warrants" means the issued and outstanding warrants entitling the holders thereof to purchase Staccato Shares;
- (xx) "**Superior Proposal**" has the meaning ascribed thereto in Section 8.7;
- (yy) "Support Agreements" means the agreements dated as of the date hereof between the relevant Party and each of the Supporting Shareholders, respectively, pursuant to which the Supporting Shareholders have agreed to support the Arrangement and to vote in favour of the Arrangement;
- (zz) "Supporting Shareholders" means the officers and directors of the relevant Party;
- (aaa) "**Tax Act**" means the *Income Tax Act* (Canada) R.S.C. 1985 (5th Supp.) c. 1, and the regulations promulgated thereunder, as now in effect and as it may be amended from time to time prior to the Effective Date;
- (bbb) "Taxes" includes any taxes, duties, fees, premiums, assessments, imposts, levies and other charges of any kind whatsoever imposed by any Governmental Entity, including all interest, penalties, fines, additions to tax or other additional amounts imposed by any Governmental Entity in respect thereof, and including those levied on, or measured by, or referred to as, income, gross receipts, profits, capital, transfer, land transfer, sales, goods and services, harmonized sales, use, value-added, excise, stamp, withholding, business, franchising, property, development, occupancy, employer health, payroll, employment, health, social services, education and social security taxes, all surtaxes, all customs duties and import and export taxes, countervail and anti-dumping duties, all license, franchise and registration fees and all employment insurance, health insurance and Canada and other government pension plan premiums or contributions;
- (ccc) "Tax Returns" includes all returns, reports, declarations, elections, notices, filings, forms, statements and other documents (whether in tangible, electronic or other form) and including any amendments, schedules, attachments, supplements, appendices and exhibits thereto, made, prepared, filed or required to be made, prepared or filed by law in respect of Taxes;
- (ddd) "Third Party" means any person other than the parties hereto and their affiliates;

- (eee) "**Third Party Claim**" has the meaning ascribed thereto in Section 7.2;
- (fff) "**Timberline Break Fee**" has the meaning ascribed thereto in Section 8.3;
- (ggg) "Timberline Expense Reimbursement Fee" has the meaning ascribed thereto in Section 8.4;
- (hhh) "**Timberline Information Circular**" means the information circular to be sent to holders of Timberline Shareholders in connection with the Timberline Meeting;
- (iii) "**Timberline Material Subsidiaries**" has the meaning ascribed thereto in Subsection 3.1(a);
- (jjj) "**Timberline Meeting**" means the special meeting of the Timberline Shareholders to be held to consider and, if thought fit, to approve the Arrangement, among other things;
- (kkk) "**Timberline Options**" means options to purchase Timberline Shares granted under Timberline's stock option plan;
- (III) "**Timberline Public Record**" has the meaning ascribed thereto in Subsection 3.1(1);
- (mmm) "**Timberline Securities**" means the Timberline Shares, Timberline Options, Timberline Warrants and other securities convertible into Timberline Shares;
- (nnn) "Timberline Securityholders" means the holders of Timberline Securities;
- (000) "**Timberline Shares**" means the shares of common stock with par value of \$0.001 in the capital of Timberline;
- (ppp) "**Timberline Shareholders**" means the holders of Timberline Shares;
- (qqq) "**Timberline Warrants**" means the issued and outstanding warrants entitling the holders thereof to purchase Timberline Shares;
- (rrr) "**Third Party**" has the meaning ascribed thereto in Section 8.1;
- (sss) "TSXV" means the TSX Venture Exchange;
- (ttt) "U.S. Exchange Act" means Securities Exchange Act of 1934, as amended, of the United States;
- (uuu) "U.S. GAAP" means accounting principles generally accepted in the United States of America from time to time and which meet the standards established by the Public Company Accounting Oversight Board (United States); and
- (vvv) "U.S. Securities Act" has the meaning ascribed thereto in Section 2.3.
- (www) "U.S. Securities Administrators" means, collectively, the SEC and any state securities commission or similar regulatory authority of any state of the United States;
- (xxx) "U.S. Securities Law" means all applicable securities legislation in the United States, including without limitation, the U.S. Securities Act and the U.S. Exchange Act, and the rules and regulations promulgated thereunder, including judicial and administrative interpretations thereof, and the securities laws of the states of the United States.

1.2 Interpretation Not Affected by Headings, etc.

The division of this Agreement into Articles, Sections and other portions and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Agreement. Unless otherwise indicated, all references to an "Article", "Section" or "Schedule" followed by a number and/or a letter refer to the specified Article or Section of or Schedule to this Agreement. The terms "this Agreement" and, unless otherwise specified, the terms "hereof", "herein" and "hereunder" and similar expressions, refer to this Agreement (including the Schedules hereto) and not to any particular Article, Section or other portion hereof.

1.3 Number and Gender

In this Agreement, unless the context otherwise requires, words used herein importing the singular include the plural and vice versa and words importing gender include all genders.

1.4 Date of Any Action

In the event that any date on which any action is required to be taken hereunder by any of the Parties is not a Business Day, such action shall be required to be taken on the next succeeding day which is a Business Day.

1.5 Subsidiaries

Notwithstanding any other provision hereof, to the extent any covenants contained herein relate, directly or indirectly, to a subsidiary of either Timberline or Staccato, each such provision will be construed as a covenant by Timberline or Staccato, as the case may be, to cause (to the fullest extent to which it is legally capable) that subsidiary to perform the required action.

1.6 Currency

Unless otherwise stated, all references in this Agreement to sums of money are expressed in lawful money of Canada.

1.7 Knowledge

Each reference herein to the knowledge of a Party means, unless otherwise specified, the existing knowledge of every vice president or more senior officer of such Party.

1.8 Statutory References

Except as expressly stated otherwise, any reference in this Agreement to a statute includes all rules and regulations made thereunder, all amendments to that statute or the rules and regulations made thereunder in force from time to time, and any statute or rule or regulation that supplements or supersedes that statute or the rules or regulations made thereunder.

1.9 Entire Agreement

This Agreement and the other agreements and documents referred to herein, constitute the entire agreement between the Parties pertaining to the terms of the Arrangement and ancillary arrangements and supersede all other prior agreements, understandings, negotiations and discussions, whether oral or written, between the Parties with respect to the terms of the Arrangement and such arrangements.

1.10 Accounting Matters

Unless otherwise stated, all accounting terms used in this Agreement:

(a) In respect of Timberline and its subsidiaries shall have the meanings attributable thereto under U.S. GAAP and all determinations of an accounting nature required to be made hereunder in

respect of Timberline shall be made in a manner consistent with U.S. GAAP as historically applied by Timberline; and

(b) in respect of Staccato and its subsidiaries shall have the meanings attributable thereto under Canadian GAAP and all determinations of an accounting nature required to be made hereunder in respect of Staccato shall be made in a manner consistent with Canadian GAAP as historically applied by Staccato.

1.11 Schedules

The following are the Schedules are annexed to this Agreement, and are hereby incorporated by reference to this Agreement and form a part hereof:

Schedule A: Plan of Arrangement

Schedule B: Outstanding Staccato Options, Staccato Warrants and

Staccato Securities

Schedule C: Outstanding Timberline Options, Timberline

Warrants and Timberline Securities

2.0 **ARRANGEMENT**

2.1 The Arrangement

The parties agree to carry out the Arrangement on the terms as set out in the Plan of Arrangement.

From and after the Effective Date, the Plan of Arrangement will have all of the effects provided therein and by applicable Laws, including the BCBCA.

2.2 Court Proceedings

Staccato shall, as soon as reasonably practicable, but in any event not later than March 31, 2010 or such later date as the parties may agree, apply to the Court pursuant to Section 288 of the BCBCA for the interim order of the Court (the "Interim Order") providing for, among other things, the calling and holding of the Staccato Meeting for the purpose of considering and, if deemed advisable, approving the Arrangement, and for the form of approval by Staccato Securityholders of the Arrangement. If the approval of the Arrangement as set forth in the Interim Order is obtained, Staccato shall take the necessary steps to submit the Arrangement to the Court and apply for the final order (the "Final Order") in such fashion as the Court may direct and, as soon as practicable thereafter, and subject to satisfaction or waiver of any other conditions provided for in this Agreement, Staccato shall file with the Registrar, pursuant to Section 292 of the BCBCA, a certified copy of the Final Order and all other necessary documents to give effect to the Arrangement.

- 2.3 The parties agree that the Arrangement will be carried out with the intention that all Timberline Shares and other Timberline Securities issued on completion of the Arrangement to the Staccato Securityholders will be issued by Timberline in reliance on the exemption from the registration requirements of the United States *Securities Act of 1933*, as amended (the "U.S. Securities Act"), provided by Section 3(a)(10) of the U.S. Securities Act (the "Section 3(a)(10) Exemption"). In order to ensure the availability of the Section 3(a)(10) Exemption, the parties agree that the Arrangement will be carried out on the following basis:
- (a) the Arrangement will be subject to the approval of the Court;

- (b) the Court will be advised as to the intention of the parties to rely on the Section 3(a)(10) Exemption prior to the hearing required to approve the Arrangement;
- (c) the Court will be required to satisfy itself as to the fairness of the Arrangement to the Staccato Securityholders subject to the Arrangement;
- (d) the Final Order approving the Arrangement that is obtained from the Court will expressly state that the Arrangement is approved by the Court as being fair to the Staccato Securityholders;
- (e) Staccato will ensure that each Staccato Securityholder entitled to receive Timberline Securities on completion of the Arrangement will be given adequate notice advising them of their right to attend the hearing of the Court to give approval of the Arrangement and providing them with sufficient information necessary for them to exercise that right;
- (f) the Staccato Securityholders will be advised that the Timberline Securities issued in the Arrangement have not been registered under the U.S. Securities Act and will be issued by Timberline in reliance on the exemption from the registration requirements of the U.S. Securities Act provided by Section 3(a)(10) of the U.S. Securities Act and may be subject to restrictions on resale under the securities laws of the United States, including, as applicable, Rule 144 under the U.S. Securities Act with respect to affiliates of Timberline;
- (g) the Interim Order approving the Staccato Meeting will specify that each Staccato Securityholder will have the right to appear before the Court at the hearing of the Court to give approval of the Arrangement so long as they enter an appearance within a reasonable time; and
- (h) the Final Order shall include a statement to the following effect:

"This Order will serve as a basis of a claim to an exemption, pursuant to section 3(a)(10) of the *United States Securities Act of 1933, as amended*, from the registration requirements otherwise imposed by that act, regarding the distribution of securities of Timberline, pursuant to the Plan of Arrangement."

3.0 REPRESENTATIONS AND WARRANTIES

Representations and Warranties of Timberline

- 3.1 As at the date of this Agreement and as at the Closing Date, Timberline represents and warrants to Staccato, and acknowledges that Staccato is relying thereon, that, except as disclosed in the Timberline Public Record:
- (a) Timberline's only material subsidiaries are Timberline Drilling Inc. and World Wide Exploration S.A. de C.V (collectively, the "**Timberline Material Subsidiaries**"), all of which are directly or indirectly 100% owned by Timberline, and Timberline does not at present own shares in and is not a party to any agreement of any nature to acquire any shares in any other corporation or entity which is material to its business and operations, other than pursuant to this Agreement and is not a party to any agreement to acquire or lease any other business operations;
- (b) each of Timberline and the Timberline Material Subsidiaries, is duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation, continuance or amalgamation and has the corporate power to own or lease its property and assets and to carry on its business as now conducted by it, is duly licensed or qualified as a foreign corporation in each jurisdiction in which the character of the property and assets now owned by it or the nature of its business as now conducted by it requires it to be so licensed or qualified (save where failure to have such licence or qualification would not in the aggregate have a Material Adverse Effect);

- (c) Timberline has the corporate power to enter into this Agreement and, subject to obtaining the requisite approvals contemplated hereby, to carry out its respective obligations hereunder;
- (d) as of the date hereof:
 - (i) the authorized capital of Timberline consists of 100,000,000 shares of common stock with par value of \$0.001 and 10,000,000 shares of preferred stock with par value of \$0.01 of which a total of 41,244,398 shares of common stock and nil shares of preferred stock were issued and outstanding as of March 22, 2010; and
 - (ii) other than as set forth in Schedule "C" hereto, Timberline has no unexercised Timberline Options, no unexercised Timberline Warrants outstanding and no other Timberline Securities issued or outstanding which entitle the holder to purchase any other Timberline Securities up to the Effective Date of the Arrangement;
- (e) the Timberline Shares issuable pursuant to the Arrangement will, upon their issuance, be validly issued and outstanding, fully paid and non-assessable shares of common stock of Timberline and will form part of a class of shares of common stock that is listed and posted for trading on Amex;
- (f) neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated by this Agreement will conflict with, result in a breach or default of, accelerate the performance required by any agreement to which Timberline or any of the Timberline Material Subsidiaries is a party, or create a state of facts, which after notice or lapse of time or both, will result in a breach by Timberline or any of the Timberline Material Subsidiaries of (i) any Law applicable to Timberline or the Timberline Material Subsidiaries, subject to obtaining necessary regulatory approvals; (ii) the constating documents, articles or resolutions of the directors or shareholders of Timberline or any of the Timberline Material Subsidiaries which are in effect at the date hereof; (iii) any mortgage, note, indenture, contract, agreement, instrument, lease or other document to which Timberline or the Timberline Material Subsidiaries is a party, except such as would not result in a Material Adverse Effect; or (iv) any judgment, decree or order binding Timberline or the Timberline Material Subsidiaries or the property or assets of Timberline or the Timberline Material Subsidiaries;
- (g) there are no agreements, covenants, undertakings or other commitments of Timberline or the Timberline Material Subsidiaries, including partnerships or joint ventures of which Timberline or the Timberline Material Subsidiaries is a partner or member, under which the consummation of the Arrangement would:
 - (i) have the effect of imposing restrictions or obligations on Timberline or the Timberline Material Subsidiaries materially greater than those imposed upon Timberline or the Timberline Material Subsidiaries or any such partnership or joint venture at the date hereof; or
 - (ii) give a third party a right to terminate any material agreement to which Timberline or the Timberline Material Subsidiaries or any such partnership or joint venture is a party or to purchase any of their respective assets; or
 - (iii) impose material restrictions on the ability of Timberline or the Timberline Material Subsidiaries to carry on any business which they might choose to carry on within any geographical area, to acquire property or dispose of their property and assets in their entirety or to change their corporate status; or

- (iv) impose material restrictions on the ability of Timberline or the Timberline Material Subsidiaries to pay any dividends or make other distributions to their shareholders or to borrow money and to mortgage and pledge their property as security therefor;
- (h) there are no actions, suits or proceedings, pending or, to the knowledge of Timberline, threatened against or affecting Timberline or the Timberline Material Subsidiaries, or any of their principals, at law or in equity, or before or by any federal, provincial, state, municipal or other governmental department, commission, board, bureau or agency, domestic or foreign, and Timberline is not aware of any existing grounds on which any such action, suit or proceeding might be commenced with any reasonable likelihood of success against Timberline or the Timberline Material Subsidiaries, except where such would not result in a Material Adverse Effect;
- (i) this Agreement has been duly authorized, executed and delivered by Timberline and constitutes a legal, valid and binding obligation, enforceable against Timberline in accordance with its terms subject to bankruptcy, insolvency and other applicable laws affecting creditors' rights generally and to general principles of equity;
- (j) the latest audited financial statements of Timberline for the fiscal year ended September 30, 2009 to be disclosed in the Staccato Information Circular are true and correct in every material respect, and have been prepared in accordance with US GAAP and fairly reflect the consolidated financial position of Timberline as at the date of such financial statements and the results of its operations for the period then ended;
- (k) the unaudited financial statements of Timberline for the interim period ended December 31, 2009 to be disclosed in the Staccato Information Circular are true and correct in every material respect, and have been prepared in accordance with US GAAP and fairly reflect the consolidated financial position of Timberline as at the date of such financial statements and the results of its operations for the period then ended;
- (l) Timberline has filed with all applicable Securities Administrators (including exchanges and markets) all material information and documents required to be filed with such authorities under the requirements of the U.S Exchange Act (the "Timberline Public Record") and the statements set forth in the Timberline Public Record are true, correct and complete and do not contain any misrepresentation as of the dates on which they were made under the circumstances in which they were made:
- (m) except as disclosed in the Timberline Public Record, the Timberline Shares are listed on Amex and Timberline is in compliance with all rules, regulations and policies of the Amex in all material respects;
- (n) Timberline is not in default in any material respect of any requirement of any Laws or Governmental Authority having jurisdiction over any securities of Timberline, except where such would not result in a Material Adverse Effect;
- (o) Timberline is WTO Investor for the purposes of the Investment Canada Act, R.S.C. 1985, c.28 (1st Supplement), as amended;
- (p) the description of the business of Timberline, its financial condition, assets and properties as provided to Staccato for inclusion in the Staccato Information Circular will not contain any untrue statement of a material fact or omit to state any material fact necessary to make such description not misleading in light of the circumstances in which they are made and will contain all information required by all applicable Laws;

- (q) there are no known or anticipated material liabilities of Timberline or the Timberline Material Subsidiaries of any kind whatsoever (including absolute, accrued or contingent liabilities) nor any commitments whether or not determined or determinable, in respect of which Staccato is or may become liable other than the liabilities disclosed on, reflected in or provided for in the financial statements referred to in paragraphs (j) and (k) of this Section 3.1, the Timberline Public Record, or to be reflected in the Staccato Information Circular or incurred in the ordinary course of business;
- (r) the corporate records and minute books of Timberline or the Timberline Material Subsidiaries as required to be maintained by them under the laws of their respective jurisdictions of incorporation are up to date and contain complete and accurate minutes of all meetings of its directors, any committees of the board of directors and shareholders held and all resolutions consented to in writing;
- (s) each of Timberline and the Timberline Material Subsidiaries owns good and marketable title to its property and assets free and clear of any and all mortgages, liens, pledges, charges, security interests, encumbrances, actions, claims or demands of any nature whatsoever or howsoever arising which would have a Material Adverse Effect on the property or assets of Timberline or the Timberline Material Subsidiaries except as disclosed in the Staccato Information Circular and all agreements by which Timberline and the Timberline Material Subsidiaries hold an interest in a property, business or assets are in good standing according to their terms and the properties are in good standing under the applicable laws of the jurisdictions in which they are situated and all filings and work commitments required to maintain the properties are in good standing, have been properly recorded and filed in a timely manner with the appropriate regulatory body and there are no mortgages, charges, encumbrances or any other interests in or on such properties other than as disclosed in the Timberline Public Record;
- (t) each of Timberline and the Timberline Material Subsidiaries has duly filed on a timely basis all Tax Returns required to be filed by it and has paid or withheld and remitted, as applicable, all Taxes which are due and payable or required to be withheld or remitted, and has paid all assessments and reassessments, and all other Taxes due and payable by it on or before the date hereof; adequate provision has been made for Taxes payable for the current period for which Tax Returns are not yet required to be filed; there are no agreements, waivers or other arrangements providing for an extension of time with respect to the filing of any Tax Return by, or payment of any Tax by Timberline or any Timberline Material Subsidiary; there are no actions, suits, proceedings, investigations or claims commenced or to the knowledge of Timberline, threatened or contemplated against Timberline or any Timberline Material Subsidiary in respect of Taxes or any matters under discussion with any Governmental Entity relating to Taxes asserted by any such Governmental Entity;
- (u) each of Timberline and the Timberline Material Subsidiaries is in compliance in all respects with each license and permit held by it and is not in any respect in violation of, or default under, the applicable statutes, ordinances, rules, regulations, orders or decrees (including, without limitation, Environmental Laws) of any governmental entities, regulatory agencies or bodies having, asserting or claiming jurisdiction over it or over any part of its operations or assets, except in all cases above where such would not result in a Material Adverse Effect;
- (v) except where such would not result in a Material Adverse Effect, each of Timberline and the Timberline Material Subsidiaries: (i) is in compliance with any and all applicable Environmental Laws; (ii) has received all permits, licenses or other approvals required of it under applicable Environmental Laws to conduct its business as currently conducted; (iii) is in compliance with all terms and conditions of each such permit, license or approval; (iv) confirms that there have been no past, and, to the knowledge of Timberline, there are no pending or threatened claims, complaints, notices or requests for information received by Timberline or the Timberline Material

Subsidiaries with respect to any alleged material violation of any Environmental Law; and (v) confirms that no conditions exist at, on or under any property now or previously owned, leased or occupied by Timberline or the Timberline Material Subsidiaries which, with the passage of time, or the giving of notice or both, would give rise to liability under any Environmental Law;

- (w) neither Timberline or the Timberline Material Subsidiaries, nor to Timberline's knowledge, any other person, has ever caused or permitted hazardous or toxic waste to be placed, held, located or disposed of on, under or at any lands or premises owned, leased or occupied by Timberline or the Timberline Material Subsidiaries otherwise than in compliance with applicable Environmental Laws and no notice has been received by Timberline of any action or potential liability in respect thereof and, to the knowledge of Timberline, no civil, criminal or enforcement actions or complaints in respect thereof are threatened, pending or have been commenced against Timberline or the Timberline Material Subsidiaries;
- (x) there are no environmental audits, evaluations, assessments, studies or tests that were commissioned by Timberline respecting the business, operations, properties or facilities of Timberline or the Timberline Material Subsidiaries:
- (y) there is no agreement, judgment, injunction, order or decree binding upon Timberline or any of the Timberline Material Subsidiaries that has or could reasonably be expected to have the effect of prohibiting, restricting or materially impairing any business practice of Timberline or the Timberline Material Subsidiaries, any acquisition of property by Timberline or the Timberline Material Subsidiaries or the conduct of business by Timberline or the Timberline Material Subsidiaries as currently conducted other than such agreements, judgments, injunctions, orders or decrees which would not, individually or in the aggregate, have a Material Adverse Effect on Timberline:
- (z) the board of directors of Timberline has:
 - (i) determined unanimously that, as of the date of this Agreement, the Arrangement is fair to the Timberline Shareholders and is in the best interests of Timberline; and
 - (ii) determined unanimously, as of the date of this Agreement, to recommend that the Timberline Shareholders vote in favour of the Timberline Resolution;
- (aa) Management of Timberline has established and maintained a system of disclosure controls and procedures designed to provide reasonable assurance that information required to be disclosed by Timberline in its annual filings, interim filings or other reports filed, furnished or submitted by it under applicable Securities Legislation is recorded, processed, summarized and reported within the time periods specified in such legislation, laws and rules. Such disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed by Timberline in its annual filings, interim filings or other reports filed, furnished or submitted under applicable Securities Legislation is accumulated and communicated to Timberline's management, including its chief executive officer and chief financial officer (or Persons performing similar functions), as appropriate to allow timely decisions regarding required disclosure
- (bb) except as disclosed in the Timberline Public Record, (i) Timberline maintains internal control over financial reporting (as defined in Rule 13a-15 under the U.S. Exchange Act, (ii) such internal control over financial reporting is effective in providing reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles and includes policies and procedures that (a) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the assets of Timberline, (b) provide reasonable

assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with US GAAP, and that receipts and expenditures of Timberline are being made only in accordance with authorizations of management and directors of Timberline, and (c) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of Timberline's assets that could have a material effect on its financial statements and (iii) Timberline has disclosed, based on the most recent evaluation of its chief executive officer and its chief financial officer prior to the date hereof, to Timberline's auditors and the audit committee of Timberline's board of directors (a) any significant deficiencies in the design or operation of its internal controls over financial reporting that are reasonably likely to adversely affect Timberline's ability to record, process, summarize and report financial information and has identified for Timberline's auditors and Timberline's board of directors any material weaknesses in internal control over financial reporting and (b) any fraud, whether or not material, that involves management or other employees who have a significant role in Timberline's internal control over financial reporting;

- (cc) the books, records and accounts of Timberline and the Timberline Material Subsidiaries, in all material respects, (i) have been maintained in accordance with good business practices and on a basis consistent with prior years, except as otherwise disclosed in Timberline Public Documents), (ii) are stated in reasonable detail and accurately and fairly reflect the transactions and dispositions of the assets of Timberline and the Timberline Material Subsidiaries and (iii) accurately and fairly reflect the basis for the Timberline financial statement in paragraphs (j) and (k) of this Section 3.1;
- (dd) except for:
 - (i) the transaction contemplated hereby, or
 - (ii) any change, condition, event or circumstance disclosed in the Timberline Public Documents,

Timberline has not been subject to any Material Adverse Change since September 30, 2009, and since such date, there has not been any damage, loss or other change of any kind whatsoever in circumstances materially affecting the business, assets, rights or capacity of Timberline or the Timberline Material Subsidiaries to conduct its respective business, such business having been conducted in the ordinary course;

- all material contracts and agreements of Timberline and the Timberline Material Subsidiaries (ee) have been disclosed in Timberline's Public Documents and true and correct copies of each or in the case of oral contracts, summaries of the material terms thereof) have been provided to Staccato. Timberline and the Timberline Material Subsidiaries are in compliance in all material respects with all material contracts, agreements, indentures, leases, policies, instruments and licences in connection with the conduct of their respective businesses and all such material contracts, agreements, indentures, leases, policies, instruments and licences are valid and binding in accordance with their terms (subject, however, to limitations with respect to enforcement imposed by Law in connection with bankruptcy or similar proceedings, the equitable power of the courts to stay proceedings before them and the execution of judgments and to the extent that equitable remedies such as specific performance and injunction are in the discretion of the courts from which they are sought) and in full force and effect, and no breach or default by Timberline or the Timberline Material Subsidiaries or event which, with notice or lapse of time or both, could constitute a material breach or material default by Timberline or the Timberline Material Subsidiaries, exists with respect thereto;
- (ff) Timberline maintains, for itself and the Timberline Material Subsidiaries, policies of insurance in force as at the date hereof that adequately cover all those risks reasonably and prudently

foreseeable in the current operation and conduct of their respective businesses which, having regard to the nature of such risk and the relative costs of obtaining insurance, it is reasonable to seek rather than to provide for self-insurance, and all of the policies in respect of such insurance coverage are in good standing in all respects and not in default in any material respect, except where the failure to obtain such insurance coverage would not reasonably be expected to have a Material Adverse Effect on Timberline;

(gg) Benefit Plans

- (i) Timberline has provided to Staccato copies of all Benefit Plans of Timberline and the Timberline Material Subsidiaries:
- (ii) Timberline and the Timberline Material Subsidiaries have complied, in all material respects, with all of the terms of their respective Benefit Plans, including the provisions of any collective agreements, funding and investment contracts or obligations applicable thereto, arising under or relating to each Benefit Plan, whether written or oral, which are maintained by or binding upon Timberline or the Timberline Material Subsidiaries. All such Benefit Plans have been administered in accordance with the documents governing the Benefit Plans and all reports and filings with Governmental Entities required in connection with each Benefit Plan have been timely made;
- (iii) All Benefit Plans of Timberline and the Timberline Material Subsidiaries are fully funded and in good standing with such Governmental Entities as may be applicable and no notice of underfunding, non-compliance, failure to be in good standing or otherwise has been received by Timberline or the Timberline Material Subsidiaries from any such Governmental Entities. No action has been taken, no event has occurred and no condition or circumstance exists that has resulted in, or could reasonably be expected to result in, any Benefit Plan of Timberline or the Timberline Material Subsidiaries, being ordered or required to be terminated or wound up in whole or in part or having its registration under applicable legislation refused or revoked, or being placed under the administration of any trustee or receiver or regulatory authority;

(hh) Employment Matters

- (i) None of Timberline or the Timberline Material Subsidiaries:
 - (a) is a party to any collective bargaining agreement or letter of understanding, letter of intent or other written communication with any trade union, council of trade unions, employee bargaining agency or affiliated bargaining agent which may qualify as a trade union, which would apply to any employees of Timberline or the Timberline Material Subsidiaries, nor has any trade union, council of trade unions, employee bargaining agency or affiliated bargaining agent applied or threatened to apply for certification as bargaining agent for the employees of Timberline or the Timberline Material Subsidiaries, nor is there a threatened or apparent union-organizing campaign for employees not covered under a collective bargaining agreement; or
 - (b) subject to any current, pending or threatened strike or lockout;
- (ii) Timberline or the Timberline Material Subsidiaries have been and are being operated in compliance in all material respects with all applicable Laws relating to employees including, but not limited to, employment and labour standards, labour risk prevention measures, social security and other contributions, occupational health and safety, employment equity, pay equity, pay practices, workers' compensation, equal employment

opportunity, human rights and labour relations and there are no current, pending or, to Timberline's knowledge, threatened, complaints, charges, orders and its investigations, prosecutions, litigation, proceedings or claims against Timberline or the Timberline Material Subsidiaries before any federal, state, municipal or other Governmental Entity, commission, board, bureau, agency or arbitrator, arbitration tribunal or instrumentality, whether domestic or foreign based on, arising out of, in connection with, or otherwise relating to unfair labour practices, the employment application for employment of or termination of employment of any individual by Timberline or the Timberline Material Subsidiaries respecting employment or labour standards, including but not limited to employment equity, pay equity, labour relations, workers' compensation or workplace safety and insurance, occupational health and safety, privacy, wrongful dismissal or human rights laws;

- (iii) There are no material notices of assessment, provisional assessment, reassessment, supplementary assessment, penalty assessment or increased assessment or any other communications related thereto which Timberline or the Timberline Material Subsidiaries has received from any workers' compensation or workplace safety and insurance board or similar authorities in any jurisdictions where the business is carried on which are unpaid on the date hereof or which will be unpaid at the Effective Date and there are no facts or circumstances which may result in a material increase in liability from any applicable workers' compensation or workplace safety and insurance legislation, regulations or rules after the Effective Date. The accident cost experience of Timberline or the Timberline Material Subsidiaries is such that there are no such material pending or possible assessments and there are no claims or potential claims which may materially affect the accident cost experience of Timberline or the Timberline Material Subsidiaries;
- (iv) To the knowledge of Timberline, without investigation, no employee of Timberline or the Timberline Material Subsidiaries is bound by any non-compete agreement or other agreement or restriction of any kind that would impede in any way the ability of such employee to carry out fully all activities of such employee in furtherance of the business of Timberline or the Timberline Material Subsidiaries;

(ii) Mineral Properties

- (i) The disclosure made in Timberline's Public Documents (including without limitation technical reports) concerning Timberline's Mineral Properties is complete and accurate in all material respects and all known facts of a scientific or technical nature are fully and accurately disclosed in Timberline's most recent public technical report, related to such Mineral Properties, and nothing has come to the attention of Timberline to indicate that any of the foregoing statements are or may be inaccurate in any material respect;
- (ii) Timberline has provided to Staccato all material information regarding all Mineral Properties owned, leased, or otherwise held by Timberline or the Timberline Material Subsidiaries that are material to the conduct of the business of Timberline or the Timberline Material Subsidiaries, and all such information as made available to Staccato is true and correct in all material respects and no material fact or facts have been omitted therefrom which would make such information misleading;
- (iii) Except as disclosed in Timberline's Public Documents, any and all of the agreements and other documents and instruments pursuant to which Timberline or the Timberline Material Subsidiaries hold their Mineral Properties and/or their interests and rights therein (including any interest in, or right to earn an interest in, any of their Mineral Properties) are valid and subsisting agreements, documents or instruments in full force and effect, enforceable in accordance with the terms thereof and neither Timberline or the

Timberline Material Subsidiaries is in default of any of the material provisions of any such agreements, documents or instruments nor has any such default been alleged. All leases, licences and claims pursuant to which Timberline or the Timberline Material Subsidiaries hold their Mineral Properties and/or their interests and rights therein are in good standing in all material respects and neither Timberline or the Timberline Material Subsidiaries is in default of any of the material provisions of any such leases, licences and claims nor has any such default been alleged;

- (iv) Other than as disclosed in Timberline's Public Record:
 - (a) all interests and rights in Timberline's Mineral Properties are (i) owned or held by Timberline or the Timberline Material Subsidiaries as owner thereof with good and marketable title, (ii) in good standing, valid, subsisting and enforceable under the applicable Laws of the jurisdictions in which the Mineral Properties are located, (iii) sufficient to permit Timberline or the Timberline Material Subsidiaries to carry on the business currently carried on by them with respect to the Mineral Properties, and (iv) free and clear of any title defects or Liens, other than Permitted Liens:
 - (b) none of Timberline's Mineral Properties or any mineral rights therein is subject to any option, pre-emption right, right of first refusal or purchase, or acquisition right;
 - (c) no royalty or other payment is payable in respect of any of Timberline's Mineral Properties and all work required to be performed in connection therewith has been performed;
 - (d) there are no restrictions on the ability of Timberline or its Subsidiaries to use, transfer or otherwise exploit any of their interests and rights in their Mineral Properties, and Timberline does not know of any claim or basis for any claim that may adversely affect such rights or interests; and
 - (e) Timberline has no knowledge of any claim or the basis for any claim that might or could materially and adversely affect the right thereof to use, transfer or otherwise exploit their interests and rights in Timberline's Mineral Properties;
- (v) Timberline or the Timberline Material Subsidiaries have all surface rights, access rights and other property rights and interests relating to Timberline's Mineral Properties necessary to permit Timberline or the Timberline Material Subsidiaries to carry on the business currently carried on by them with respect to those Mineral Properties, and no other property rights are necessary for the conduct of Timberline's or the Timberline Material Subsidiaries' business;

(jj) Real Property

- (i) Timberline has provided to Staccato all material information regarding all Owned Real Property of Timberline and its Subsidiaries and all such information as made available to Staccato is true and correct in all material respects and no material fact or facts have been omitted therefrom which would make such information misleading. Each of Timberline or the Timberline Material Subsidiaries has good and marketable title to all of its Owned Real Property, free and clear of all Liens, other than Permitted Liens;
- (ii) Timberline has provided to Staccato all material information regarding all Real Property Leases of Timberline and its Subsidiaries. Each such Real Property Lease constitutes is valid and legally binding on the parties thereto, and is in full force and effect. All rents

and other sums and charges payable by Timberline and the Timberline Material Subsidiaries as tenants under such Real Property Leases are current and no termination event or condition or uncured default on the part of Timberline or the Timberline Material Subsidiaries or, to Timberline's knowledge, the landlord, exists under any such Real Property Lease. Timberline and the Timberline Material Subsidiaries have a good and valid interest in each parcel of such leased real property, free and clear of all Liens, other than Permitted Liens;

(kk) Condition of Property

All real and tangible personal property of Timberline and the Timberline Material Subsidiaries necessary for the conduct of the business currently carried on by Timberline and the Timberline Material Subsidiaries is in good repair and is operational and usable in the manner in which it is currently being utilized, subject to normal wear and tear and technical obsolescence, repair or replacement, except for such property whose failure to be in such condition does not, and could not be reasonably expected to have, a Material Adverse Effect on Timberline;

(ll) Commissions and fees

Timberline has not entered into any agreement that would entitle any Person to any valid claim against Timberline for a broker's commission, finder's fee, expense reimbursement or any like payment in respect of the Arrangement or any other matter contemplated by this Agreement;

(mm) Winding-up

There is no bankruptcy, liquidation, winding-up or other similar proceeding pending or in progress or, to the knowledge of Timberline threatened against Timberline or any of the Timberline Material Subsidiaries before any Governmental Entity;

(nn) Solvency

There are reasonable grounds for believing that (i) Timberline is able to pay its liabilities as they become due, (ii) the realizable value of the assets of Timberline are not less than the aggregate of the liabilities thereof and the stated capital of all classes of shares thereof, and (iii) no creditor of Timberline will be prejudiced by the Arrangement;

(oo) Shareholder Rights Plan

No shareholder rights plan is in force in respect of Timberline;

(pp) Auditors

The auditors of Timberline are independent public accountants as required by applicable Laws and there is not now, and there has never been, any reportable disagreement with the present or any former auditors of Timberline;

(qq) Intellectual Property

None of Timberline or the Timberline Material Subsidiaries own or license any patents, patent rights, trademarks, trade names, service marks, copyrights, know how or other proprietary intellectual property rights that are material to the conduct of the business of Timberline or the Timberline Material Subsidiaries and, with respect to its mineral property databases, Timberline

or the relevant Timberline Material Subsidiary holds all such rights as are necessary to enable it to continue to use these databases, consistent with their past use or in the ordinary course of its business;

(rr) Restrictions on Business

None of Timberline or the Timberline Material Subsidiaries is a party to or bound by any non-competition agreement or any other agreement, obligation, judgment, injunction, order or decree which purports to (i) limit the manner or the localities in which all or any material portion of the business of Timberline or the Timberline Material Subsidiaries are conducted, (ii) limit any business practice of Timberline or the Timberline Material Subsidiaries in any material respect, or (iii) restrict any acquisition or disposition of any property by Timberline or the Timberline Material Subsidiaries in any material respect;

(ss) Cease Trade Orders

No order (or the equivalent) ceasing or suspending the trading of its securities or prohibiting the sale of securities by Timberline or the Timberline Material Subsidiaries has been issued and is in force as of the date hereof and, to the knowledge of Timberline, no proceedings for this purpose have been instituted or are pending, contemplated or threatened;

(tt) Indebtedness to Directors, Officers and Securityholders

None of Timberline or the Timberline Material Subsidiaries is indebted to any of its directors or officers or any of their associates, to Timberline's knowledge, to any Timberline securityholder;

(uu) Indebtedness of Directors, Officers and Securityholders

None of the directors or officers of Timberline or any of their associates or, to Timberline's knowledge, Timberline securityholders is indebted or under obligation to Timberline or to any of its Subsidiaries on any account whatsoever;

(vv) Issuance of Timberline Shares

The Timberline Shares to be issued pursuant to the Arrangement will be duly and validly issued by Timberline on the date on which such shares are to be issued, and will, upon such issuance, be issued as fully paid and non-assessable shares;

(ww) None of Timberline, the Timberline Material Subsidiaries or, to Timberline's knowledge, any directors or officers, agents or employees of Timberline or the Timberline Material Subsidiaries, has (i) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses related to political activity; (ii) made any unlawful payment to foreign or domestic government officials or employees or to foreign or domestic political parties or campaigns or violated any provision of the Foreign Corrupt Practices Act of 1977, as amended; or (iii) made any payment in the nature of criminal bribery;

(xx) Information

All written information, data and materials made available by Timberline to Staccato in connection with the transactions contemplated by this Agreement is true, complete and correct in all material respects as at the respective dates of which it was prepared; and

(yy) none of the representations, warranties or statements of fact made in this Section contain any untrue statement of a material fact or omit to state any material fact necessary to make any such warranty or representation not misleading.

Representations and Warranties of Staccato

- 3.2 As at the date of this Agreement and as at the Closing Date, Staccato represents and warrants to the other parties, and acknowledges that the other parties are relying thereon, that, except as disclosed on the Staccato Disclosure Schedule and in the Staccato Public Record:
- (a) Staccato's only material subsidiary is BH Minerals USA, Inc. (the "Staccato Material Subsidiary"), which is directly or indirectly 100% owned by Staccato, and Staccato does not at present own shares in and is not a party to any agreement of any nature to acquire any shares in any other corporation or entity which is material to its business and operations, other than pursuant to this Agreement and is not a party to any agreement to acquire or lease any other business operations;
- (b) each of Staccato and the Staccato Material Subsidiary is a company duly organized, validly existing and in good standing under the laws of its jurisdiction of amalgamation and has the corporate power to own or lease its property and assets and to carry on its business as now conducted by it, is duly licensed or qualified as a foreign corporation in each jurisdiction in which the character of the property and assets now owned by it or the nature of its business as now conducted by it requires it to be so licensed or qualified (save where failure to have such licence or qualification would not in the aggregate have a Material Adverse Effect);
- (c) Staccato has the corporate power to enter into this Agreement and, subject to obtaining the requisite approvals contemplated hereby, to carry out its obligations hereunder;
- (d) neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated by this Agreement will conflict with, result in a breach or default of, accelerate the performance required by any agreement to which Staccato or the Staccato Material Subsidiary is a party or create a state of facts, which after notice or lapse of time or both, will result in a breach by Staccato or the Staccato Material Subsidiary of (i) any Law applicable to Staccato or the Staccato Material Subsidiary; (ii) the constating documents, by-laws or resolutions of the directors or shareholders of Staccato or the Staccato Material Subsidiary which are in effect at the date hereof; (iii) any mortgage, note, indenture, contract, agreement, instrument, lease or other document to which Staccato or the Staccato Material Subsidiary is a party, except as such would not result in a Material Adverse Effect; or (iv) any judgment, decree or order binding Staccato or the Staccato Material Subsidiary;
- (e) as of the date hereof:
 - (i) the authorized capital of Staccato consists of an unlimited number of common shares without par value, of which a total of 99,842,793 common shares were issued and outstanding as of March 22, 2010; and
 - (ii) other than as set forth in Schedule "B" hereto, Staccato has no unexercised Staccato Options, no unexercised Staccato Warrants outstanding and no other Staccato Securities issued or outstanding which entitle the holder to purchase any other Staccato Securities up to the Effective Date of the Arrangement;
- (f) there are no agreements, covenants, undertakings or other commitments of Staccato or the Staccato Material Subsidiary, including partnerships or joint ventures of which Staccato or the

Staccato Material Subsidiary is a partner or member, under which the consummation of the Arrangement would:

- (i) have the effect of imposing restrictions or obligations on Staccato or the Staccato Material Subsidiary materially greater than those imposed upon Staccato or the Staccato Material Subsidiary or any such partnership or joint venture at the date hereof; or
- (ii) give a third party a right to terminate any material agreement to which Staccato or the Staccato Material Subsidiary or any such partnership or joint venture is a party or to purchase any of their respective assets; or
- (iii) impose material restrictions on the ability of Staccato or the Staccato Material Subsidiary to carry on any business which it might choose to carry on within any geographical area, to acquire property or dispose of its property and assets in their entirety or to change its corporate status; or
- (iv) impose material restrictions on the ability of Staccato or the Staccato Material Subsidiary to pay any dividends or make other distributions to its shareholders or to borrow money and to mortgage and pledge its property as security therefor;
- (g) there are no actions, suits or proceedings, pending or, to the knowledge of Staccato, threatened against or affecting Staccato or the Staccato Material Subsidiary or any of its principals, at law or in equity, or before or by any federal, provincial, state, municipal or other governmental department, commission, board, bureau or agency, domestic or foreign, and Staccato is not aware of any existing grounds on which any such action, suit or proceeding might be commenced with any reasonable likelihood of success against Staccato or the Staccato Material Subsidiary, except where such would not result in a Material Adverse Effect:
- (h) this Agreement has been duly authorized, executed and delivered by Staccato and constitutes a legal, valid and binding obligation, enforceable against it in accordance with its terms subject to bankruptcy, insolvency and other applicable laws affecting creditors' rights generally and to general principles of equity;
- (i) Staccato is a "reporting issuer" within the meaning of the securities laws of British Columbia and Alberta, has no reporting requirements under any other jurisdiction, is not on a list of defaulting issuers maintained by the securities commissions in these jurisdictions and no regulatory authority having jurisdiction has issued any order preventing or suspending trading of any securities of Staccato which is currently outstanding;
- (j) the latest audited financial statements of Staccato for the year ended April 30, 2009 are true and correct in every material respect, and have been prepared on a consolidated basis in accordance with Canadian GAAP and fairly reflect the consolidated financial position of Staccato as at the date of such financial statements and the results of its operations for the period then ended;
- (k) the unaudited financial statements of Staccato for the interim period ended October 31, 2009 to be disclosed in the Staccato Information Circular are true and correct in every material respect, and have been prepared in accordance with Canadian GAAP and fairly reflect the consolidated financial position of Staccato as at the date of such financial statements and the results of its operations for the period then ended;
- (l) Staccato has filed with all applicable Securities Administrators (including exchanges and markets) all material information and documents required to be filed with such authorities under the securities legislation in the jurisdictions in which it is a reporting issuer (the "Staccato Public Record") and the statements set forth in the Staccato Public Record are true, correct and

complete and do not contain any misrepresentation as of the dates on which they were made under the circumstances in which they were made and Staccato has not filed any confidential material change reports which currently remain confidential or similar reports;

- (m) the Staccato Shares trade on the TSXV and Staccato is in compliance with all rules, regulations and policies of the TSXV in all material respects;
- (n) Staccato is not in default in any material respect of any requirement of any applicable Laws or Governmental Authority having jurisdiction over any securities of Staccato, except where such would not result in a Material Adverse Effect;
- (o) the description of the business of Staccato, its financial condition, assets and properties in the Staccato Information Circular will not contain any untrue statement of a material fact or omit to state any material fact necessary to make such description not misleading in light of the circumstances in which they are made and will contain all information required by all applicable Law and the rules of the TSXV:
- (p) Staccato has not incurred any liability for brokerage fees, finder's fees, agent's commissions or other similar forms of compensation in connection with this Agreement or the Arrangement except as disclosed in the Staccato Information Circular;
- (q) there are no known or anticipated material liabilities of Staccato or the Staccato Material Subsidiary of any kind whatsoever (including absolute, accrued or contingent liabilities) nor any commitments whether or not determined or determinable, in respect of which Staccato is or may become liable other than the liabilities disclosed on, reflected in or provided for in the financial statements referred to in paragraphs (j) and (k) of this Section 3.2, the Staccato Public Record, or to be reflected in the Staccato Information Circular or incurred in the ordinary course of business;
- (r) the corporate records and minute books of Staccato or the Staccato Material Subsidiary as required to be maintained by them under the laws of their respective jurisdictions of incorporation are up to date and contain complete and accurate minutes of all meetings of its directors, any committees of the board of directors and shareholders held and all resolutions consented to in writing;
- (s) each of Staccato and the Staccato Material Subsidiary owns good and marketable title to its property and assets free and clear of any and all mortgages, liens, pledges, charges, security interests, encumbrances, actions, claims or demands of any nature whatsoever or howsoever arising which would have a Material Adverse Effect on the property or assets of Staccato or the Staccato Material Subsidiary except as disclosed in the Staccato Information Circular and all agreements by which Staccato and the Staccato Material Subsidiary hold an interest in a property, business or assets are in good standing according to their terms and the properties are in good standing under the applicable laws of the jurisdictions in which they are situated and all filings and work commitments required to maintain the properties are in good standing, have been properly recorded and filed in a timely manner with the appropriate regulatory body and there are no mortgages, charges, encumbrances or any other interests in or on such properties other than as disclosed in the Staccato Public Record;
- (t) each of Staccato and the Staccato Material Subsidiary has duly filed on a timely basis all Tax Returns required to be filed by it and has paid or withheld all Taxes which are due and payable or required to be withheld, and has paid all assessments and reassessments, and all other Taxes due and payable by it on or before the date hereof; adequate provision has been made for Taxes payable for the current period for which Tax Returns are not yet required to be filed; there are no agreements, waivers or other arrangements providing for an extension of time with respect to the filing of any Tax Return by, or payment of any Tax by Staccato or the Staccato Material

Subsidiary; there are no actions, suits, proceedings, investigations or claims commenced or to the knowledge of Staccato, threatened or contemplated against Staccato or the Staccato Material Subsidiary in respect of Taxes or any matters under discussion with any Governmental Entity relating to Taxes asserted by any such Governmental Entity;

- (u) The aggregate paid-up capital of all classes of shares of the capital stock of Staccato exceeds the fair market value of its assets, as determined for purposes of the Tax Act, and will do so at the Closing of the Arrangement;
- (v) The amount of non-capital losses available for carry forward by Staccato is accurately reflected in Staccato's most recent income tax return for the taxation year ended April 30, 2009, and Staccato has no reason to believe that the non-capital losses would be denied or otherwise challenged by the Canada Revenue Agency;
- (w) each of Staccato and the Staccato Material Subsidiary is in compliance in all respects with each license and permit held by it and is not in any respect in violation of, or default under, the applicable statutes, ordinances, rules, regulations, orders or decrees (including, without limitation, Environmental Laws) of any governmental entities, regulatory agencies or bodies having, asserting or claiming jurisdiction over it or over any part of its operations or assets, except in all cases above where such would not result in a Material Adverse Effect;
- (x) Except where such would not result in a Material Adverse Effect, each of Staccato and Staccato Material Subsidiary: (i) is in compliance with any and all applicable Environmental Laws; (ii) has received all permits, licenses or other approvals required of it under applicable Environmental Laws to conduct its business as currently conducted; (iii) is in compliance with all terms and conditions of each such permit, license or approval; (iv) confirms that there have been no past, and, to the knowledge of Staccato, there are no pending or threatened claims, complaints, notices or requests for information received by Staccato or Staccato Material Subsidiary with respect to any alleged material violation of any Environmental Law; and (v) confirms that no conditions exist at, on or under any property now or previously owned, leased or occupied by Staccato or Staccato Material Subsidiary which, with the passage of time, or the giving of notice or both, would give rise to liability under any Environmental Law;
- (y) neither Staccato nor Staccato Material Subsidiary, nor to Staccato's knowledge, any other person, has ever caused or permitted hazardous or toxic waste to be placed, held, located or disposed of on, under or at any lands or premises owned, leased or occupied by Staccato or Staccato Material Subsidiary otherwise than in compliance with applicable Environmental Laws and no notice has been received by Staccato of any action or potential liability in respect thereof and, to the knowledge of Staccato, no civil, criminal or enforcement actions or complaints in respect thereof are threatened, pending or have been commenced against Staccato or Staccato Material Subsidiary;
- (z) there are no environmental audits, evaluations, assessments, studies or tests that were commissioned by Staccato respecting the business, operations, properties or facilities of Staccato or Staccato Material Subsidiary;
- (aa) Staccato has filed with the British Columbia Securities Commission and the Alberta Securities Commission all of the technical reports required to be filed under National Instrument 43-101 in respect of each property material to Staccato and all public disclosure made by Staccato regarding its properties complies in all material respects with the requirements of National Instrument 43-101;
- (bb) there is no agreement, judgment, injunction, order or decree binding upon Staccato or any of the Staccato Material Subsidiary that has or could reasonably be expected to have the effect of

prohibiting, restricting or materially impairing any business practice of Staccato or the Staccato Material Subsidiary, any acquisition of property by Staccato or the Staccato Material Subsidiary or the conduct of business by Staccato or the Staccato Material Subsidiary as currently conducted other than such agreements, judgments, injunctions, orders or decrees which would not, individually or in the aggregate, have a Material Adverse Effect on Staccato;

- (cc) the board of directors of Staccato has:
 - (i) determined unanimously that, as of the date of this Agreement, the Arrangement is fair to the Staccato Securityholders and is in the best interests of Staccato;
 - (ii) received an opinion from Paradigm Capital Inc., that as of the date of this Agreement, the consideration to be paid under the Arrangement is fair, from a financial point of view, to Staccato and the Staccato Shareholders; and
 - (iii) determined unanimously, as of the date of this Agreement, to recommend that the Staccato Securityholders vote in favour of the Arrangement;
- (dd) management of Staccato has established and maintained a system of disclosure controls and procedures designed to provide reasonable assurance that information required to be disclosed by Staccato in its annual filings, interim filings or other reports filed, furnished or submitted by it under applicable Securities Legislation is recorded, processed, summarized and reported within the time periods specified in such legislation, laws and rules. Such disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed by Staccato in its annual filings, interim filings or other reports filed, furnished or submitted under applicable Securities Legislation is accumulated and communicated to Staccato's management, including its chief executive officer and chief financial officer (or Persons performing similar functions), as appropriate to allow timely decisions regarding required disclosure;
- except as disclosed in the Staccato Public Record, (i) Staccato maintains internal control over (ee) financial reporting as defined under applicable Canadian Securities Legislation, (ii) such internal control over financial reporting is effective in providing reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles and includes policies and procedures that (a) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the assets of Staccato, (b) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with Canadian GAAP, and that receipts and expenditures of Staccato are being made only in accordance with authorizations of management and directors of Staccato, and (c) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of Staccato's assets that could have a material effect on its financial statements and (iii) Staccato has disclosed, based on the most recent evaluation of its chief executive officer and its chief financial officer prior to the date hereof, to Staccato's auditors and the audit committee of Staccato's board of directors (a) any significant deficiencies in the design or operation of its internal controls over financial reporting that are reasonably likely to adversely affect Staccato's ability to record, process, summarize and report financial information and has identified for Staccato's auditors and Staccato's board of directors any material weaknesses in internal control over financial reporting and (ii) any fraud, whether or not material, that involves management or other employees who have a significant role in Staccato's internal control over financial reporting;
- (ff) The books, records and accounts of Staccato and the Staccato Material Subsidiary, in all material respects, (i) have been maintained in accordance with good business practices and on a basis

consistent with prior years, except as otherwise disclosed in Staccato Public Record), (ii) are stated in reasonable detail and accurately and fairly reflect the transactions and dispositions of the assets of Staccato and the Staccato Material Subsidiary and (iii) accurately and fairly reflect the basis for the Staccato financial statement in paragraphs (j) and (k) of this Section 3.1;

(gg) Except for:

- (i) the transaction contemplated hereby, or
- (ii) any change, condition, event or circumstance disclosed in the Staccato Public Record,

Staccato has not been subject to any Material Adverse Change since April 30, 2009, and since such date, there has not been any damage, loss or other change of any kind whatsoever in circumstances materially affecting the business, assets, rights or capacity of Staccato or the Staccato Material Subsidiary to conduct its respective business, such business having been conducted in the ordinary course;

- (hh) All material contracts and agreements of Staccato and the Staccato Material Subsidiary have been disclosed in Staccato's Public Record and true and correct copies of each or in the case of oral contracts, summaries of the material terms thereof) have been provided to Timberline. Staccato and the Staccato Material Subsidiary are in compliance in all material respects with all material contracts, agreements, indentures, leases, policies, instruments and licences in connection with the conduct of their respective businesses and all such material contracts, agreements, indentures, leases, policies, instruments and licences are valid and binding in accordance with their terms (subject, however, to limitations with respect to enforcement imposed by Law in connection with bankruptcy or similar proceedings, the equitable power of the courts to stay proceedings before them and the execution of judgments and to the extent that equitable remedies such as specific performance and injunction are in the discretion of the courts from which they are sought) and in full force and effect, and no breach or default by Staccato or the Staccato Material Subsidiary or event which, with notice or lapse of time or both, could constitute a material breach or material default by Staccato or the Staccato Material Subsidiary, exists with respect thereto;
- (ii) Staccato maintains, for itself and the Staccato Material Subsidiary, policies of insurance in force as at the date hereof that adequately cover all those risks reasonably and prudently foreseeable in the current operation and conduct of their respective businesses which, having regard to the nature of such risk and the relative costs of obtaining insurance, it is reasonable to seek rather than to provide for self-insurance, and all of the policies in respect of such insurance coverage are in good standing in all respects and not in default in any material respect, except where the failure to obtain such insurance coverage would not reasonably be expected to have a Material Adverse Effect on Staccato;

(jj) Benefit Plans

- (i) Staccato has provided to Timberline copies of all Benefit Plans of Staccato and the Staccato Material Subsidiary;
- (ii) Staccato and the Staccato Material Subsidiary have complied, in all material respects, with all of the terms of their respective Benefit Plans, including the provisions of any collective agreements, funding and investment contracts or obligations applicable thereto, arising under or relating to each Benefit Plan, whether written or oral, which are maintained by or binding upon Staccato or the Staccato Material Subsidiary. All such Benefit Plans have been administered in accordance with the documents governing the Benefit Plans and all reports and filings with Governmental Entities required in connection with each Benefit Plan have been timely made;

(iii) All Benefit Plans of Staccato and the Staccato Material Subsidiary are fully funded and in good standing with such Governmental Entities as may be applicable and no notice of underfunding, non-compliance, failure to be in good standing or otherwise has been received by Staccato or the Staccato Material Subsidiary from any such Governmental Entities. No action has been taken, no event has occurred and no condition or circumstance exists that has resulted in, or could reasonably be expected to result in, any Benefit Plan of Staccato or the Staccato Material Subsidiary, being ordered or required to be terminated or wound up in whole or in part or having its registration under applicable legislation refused or revoked, or being placed under the administration of any trustee or receiver or regulatory authority;

(kk) Employment Matters

- (i) None of Staccato or the Staccato Material Subsidiary:
 - (a) is a party to any collective bargaining agreement or letter of understanding, letter of intent or other written communication with any trade union, council of trade unions, employee bargaining agency or affiliated bargaining agent which may qualify as a trade union, which would apply to any employees of Staccato or the Staccato Material Subsidiary, nor has any trade union, council of trade unions, employee bargaining agency or affiliated bargaining agent applied or threatened to apply for certification as bargaining agent for the employees of Staccato or the Staccato Material Subsidiary, nor is there a threatened or apparent union-organizing campaign for employees not covered under a collective bargaining agreement; or
 - (b) subject to any current, pending or threatened strike or lockout;
- (ii) Staccato or the Staccato Material Subsidiary have been and are being operated in compliance in all material respects with all applicable Laws relating to employees including, but not limited to, employment and labour standards, labour risk prevention measures, social security and other contributions, occupational health and safety, employment equity, pay equity, pay practices, workers' compensation, equal employment opportunity, human rights and labour relations and there are no current, pending or, to Staccato's knowledge, threatened, complaints, charges, orders and its investigations, prosecutions, litigation, proceedings or claims against Staccato or the Staccato Material Subsidiary before any federal, state, municipal or other Governmental Entity, commission, board, bureau, agency or arbitrator, arbitration tribunal or instrumentality, whether domestic or foreign based on, arising out of, in connection with, or otherwise relating to unfair labour practices, the employment application for employment of or termination of employment of any individual by Staccato or the Staccato Material Subsidiary respecting employment or labour standards, including but not limited to employment equity, pay equity, labour relations, workers' compensation or workplace safety and insurance, occupational health and safety, privacy, wrongful dismissal or human rights laws;
- (iii) There are no material notices of assessment, provisional assessment, reassessment, supplementary assessment, penalty assessment or increased assessment or any other communications related thereto which Staccato or the Staccato Material Subsidiary has received from any workers' compensation or workplace safety and insurance board or similar authorities in any jurisdictions where the business is carried on which are unpaid on the date hereof or which will be unpaid at the Effective Date and there are no facts or circumstances which may result in a material increase in liability from any applicable workers' compensation or workplace safety and insurance legislation, regulations or rules after the Effective Date. The accident cost experience of Staccato or the Staccato Material

Subsidiary is such that there are no such material pending or possible assessments and there are no claims or potential claims which may materially affect the accident cost experience of Staccato or the Staccato Material Subsidiary;

(iv) To the knowledge of Staccato, without investigation, no employee of Staccato or the Staccato Material Subsidiary is bound by any non-compete agreement or other agreement or restriction of any kind that would impede in any way the ability of such employee to carry out fully all activities of such employee in furtherance of the business of Staccato or the Staccato Material Subsidiary;

(ll) Mineral Properties

- (i) The disclosure made in Staccato's Public Documents (including without limitation technical reports) concerning Staccato's Mineral Properties is complete and accurate in all material respects and all known facts of a scientific or technical nature are fully and accurately disclosed in Staccato's most recent public technical report, related to such Mineral Properties, and nothing has come to the attention of Staccato to indicate that any of the foregoing statements are or may be inaccurate in any material respect;
- (ii) Staccato has provided to Timberline all material information regarding all Mineral Properties owned, leased, or otherwise held by Staccato or the Staccato Material Subsidiary that are material to the conduct of the business of Staccato or the Staccato Material Subsidiary, and all such information as made available to Timberline is true and correct in all material respects and no material fact or facts have been omitted therefrom which would make such information misleading;
- (iii) Except as disclosed in Staccato's Public Documents, any and all of the agreements and other documents and instruments pursuant to which Staccato or the Staccato Material Subsidiary hold their Mineral Properties and/or their interests and rights therein (including any interest in, or right to earn an interest in, any of their Mineral Properties) are valid and subsisting agreements, documents or instruments in full force and effect, enforceable in accordance with the terms thereof and neither Staccato or the Staccato Material Subsidiary is in default of any of the material provisions of any such agreements, documents or instruments nor has any such default been alleged. All leases, licences and claims pursuant to which Staccato or the Staccato Material Subsidiary hold their Mineral Properties and/or their interests and rights therein are in good standing in all material respects and neither Staccato or the Staccato Material Subsidiary is in default of any of the material provisions of any such leases, licences and claims nor has any such default been alleged;
- (iv) Other than as disclosed in Staccato's Public Documents:
 - (a) all interests and rights in Staccato's Mineral Properties are (i) owned or held by Staccato or the Staccato Material Subsidiary as owner thereof with good and marketable title, (ii) in good standing, valid, subsisting and enforceable under the applicable Laws of the jurisdictions in which the Mineral Properties are located, (iii) sufficient to permit Staccato or the Staccato Material Subsidiary to carry on the business currently carried on by them with respect to the Mineral Properties, and (iv) free and clear of any title defects or Liens, other than Permitted Liens;
 - (b) none of Staccato's Mineral Properties or any mineral rights therein is subject to any option, pre-emption right, right of first refusal or purchase, or acquisition right;

- (c) no royalty or other payment is payable in respect of any of Staccato's Mineral Properties and all work required to be performed in connection therewith has been performed;
- (d) there are no restrictions on the ability of Staccato or its Subsidiaries to use, transfer or otherwise exploit any of their interests and rights in their Mineral Properties, and Staccato does not know of any claim or basis for any claim that may adversely affect such rights or interests; and
- (e) Staccato has no knowledge of any claim or the basis for any claim that might or could materially and adversely affect the right thereof to use, transfer or otherwise exploit their interests and rights in Staccato's Mineral Properties;
- (v) Staccato or the Staccato Material Subsidiary have all surface rights, access rights and other property rights and interests relating to Staccato's Mineral Properties necessary to permit Staccato or the Staccato Material Subsidiary to carry on the business currently carried on by them with respect to those Mineral Properties, and no other property rights are necessary for the conduct of Staccato's or the Staccato Material Subsidiary' business;

(mm) Real Property

- (i) Staccato has provided to Staccato all material information regarding all Owned Real Property of Staccato and its Subsidiaries and all such information as made available to Staccato is true and correct in all material respects and no material fact or facts have been omitted therefrom which would make such information misleading. Each of Staccato or the Staccato Material Subsidiary has good and marketable title to all of its Owned Real Property, free and clear of all Liens, other than Permitted Liens;
- (ii) Staccato has provided to Staccato all material information regarding all Real Property Leases of Staccato and its Subsidiaries. Each such Real Property Lease constitutes is valid and legally binding on the parties thereto, and is in full force and effect. All rents and other sums and charges payable by Staccato and the Staccato Material Subsidiary as tenants under such Real Property Leases are current and no termination event or condition or uncured default on the part of Staccato or the Staccato Material Subsidiary or, to Staccato's knowledge, the landlord, exists under any such Real Property Lease. Staccato and the Staccato Material Subsidiary have a good and valid interest in each parcel of such leased real property, free and clear of all Liens, other than Permitted Liens;

(nn) Condition of Property

All real and tangible personal property of Staccato and the Staccato Material Subsidiary necessary for the conduct of the business currently carried on by Staccato and its Subsidiaries is in good repair and is operational and usable in the manner in which it is currently being utilized, subject to normal wear and tear and technical obsolescence, repair or replacement, except for such property whose failure to be in such condition does not, and could not be reasonably expected to have, a Material Adverse Effect on Staccato;

(oo) Commissions and fees

Except for the engagement agreement dated February 9, 2010 with Kingsdale Shareholder Services Inc. and the engagement agreement dated February 2, 2010 with Paradigm Capital Inc., Staccato has not entered into any agreement that would entitle any Person to any valid claim against Staccato for a broker's commission, finder's fee, expense reimbursement or any like payment in respect of the Arrangement or any other matter contemplated by this Agreement;

(pp) Winding-up

There is no bankruptcy, liquidation, winding-up or other similar proceeding pending or in progress or, to the knowledge of Staccato threatened against Staccato or any of its Subsidiaries before any Governmental Entity;

(qq) Solvency

There are reasonable grounds for believing that (i) Staccato is able to pay its liabilities as they become due, (ii) the realizable value of the assets of Staccato are not less than the aggregate of the liabilities thereof and the stated capital of all classes of shares thereof, and (iii) no creditor of Staccato will be prejudiced by the Arrangement;

(rr) Shareholder Rights Plan

No shareholder rights plan is in force in respect of Staccato;

(ss) Auditors

The auditors of Staccato are independent public accountants as required by applicable Laws and there is not now, and there has never been, any reportable disagreement with the present or any former auditors of Staccato;

(tt) Intellectual Property

None of Staccato or the Staccato Material Subsidiary own or license any patents, patent rights, trademarks, trade names, service marks, copyrights, know how or other proprietary intellectual property rights that are material to the conduct of the business of Staccato or the Staccato Material Subsidiary and, with respect to its mineral property databases, Staccato or the relevant Staccato Material Subsidiary holds all such rights as are necessary to enable it to continue to use these databases, consistent with their past use or in the ordinary course of its business;

(uu) Restrictions on Business

None of Staccato or the Staccato Material Subsidiary is a party to or bound by any non-competition agreement or any other agreement, obligation, judgment, injunction, order or decree which purports to (i) limit the manner or the localities in which all or any material portion of the business of Staccato or the Staccato Material Subsidiary are conducted, (ii) limit any business practice of Staccato or the Staccato Material Subsidiary in any material respect, or (iii) restrict any acquisition or disposition of any property by Staccato or the Staccato Material Subsidiary in any material respect;

(vv) Cease Trade Orders

No order (or the equivalent) ceasing or suspending the trading of its securities or prohibiting the sale of securities by Staccato or the Staccato Material Subsidiary has been issued and is in force as of the date hereof and, to the knowledge of Staccato, no proceedings for this purpose have been instituted or are pending, contemplated or threatened;

(ww) Indebtedness to Directors, Officers and Securityholders

None of Staccato or the Staccato Material Subsidiary is indebted to any of its directors or officers or any of their associates, to Staccato's knowledge, to any Staccato securityholder;

(xx) Indebtedness of Directors, Officers and Securityholders

None of the directors or officers of Staccato or any of their associates or, to Staccato's knowledge, Staccato securityholders is indebted or under obligation to Staccato or to any of its Subsidiaries on any account whatsoever;

- (zz) Staccato and the Staccato Material Subsidiary do not hold assets located in the United States (other than investment assets, voting or nonvoting securities of another person, and assets included pursuant to Section 801.40(d)(2) of the HSR Act) having a total value of over USD\$63,400,000, and Staccato and the Staccato Material Subsidiary have not made aggregate sales in or into the United States of over USD\$63,400,000 in its most recent fiscal year, all within the meaning of the HSR Act;
- (aaa) None of Staccato, the Staccato Material Subsidiary or, to Staccato's knowledge, any directors or officers, agents or employees of Staccato or the Staccato Material Subsidiary, has (i) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses related to political activity; (ii) made any unlawful payment to foreign or domestic government officials or employees or to foreign or domestic political parties or campaigns or violated any provision of the Foreign Corrupt Practices Act of 1977, as amended; or (iii) made any payment in the nature of criminal bribery;

(bbb) Acknowledgment of SEC Review

Staccato acknowledges that the Arrangement will be subject to the review and approval of the SEC upon the filing of the Timberline Circular with the SEC under cover of Schedule 14A as a preliminary proxy statement under Regulation 14A of the U.S. Exchange Act;

(yy) Information

All written information, data and materials made available by Timberline to Timberline in connection with the transactions contemplated by this Agreement is true, complete and correct in all material respects as at the respective dates of which it was prepared;

and

(zz) none of the representations, warranties or statements of fact made in this Section contain any untrue statement of a material fact or omit to state any material fact necessary to make any such warranty or representation not misleading;

4.0 COVENANTS

Consultation

- 4.1 Each Party agrees to consult with the other Party, and to provide the other Party with a reasonable prior opportunity to review and comment on, any press release or public statement or any filing to be made with any Governmental Entity with respect to this Agreement or the Arrangement (including any filing with any securities administrator or stock exchange with respect thereto) prior to the release or submission thereto; provided that, if a Party is required by applicable Laws to make a public announcement with respect to this Agreement and/or the Arrangement, such Party will provide as much prior notice (including the proposed text of the announcement) to the other Party as is reasonably possible. All requests and enquiries from any Governmental Entity with respect to the Arrangement shall be dealt with by Timberline and Staccato in consultation with each other, and Timberline and Staccato shall promptly co-operate with and provide all necessary information and assistance reasonably required by such Governmental Entity upon being requested to do so by such authority. Furthermore:
 - (i) each Party shall promptly notify the other Party of written communications of any nature which it receives from any Governmental Entity with respect to the Arrangement and provide the other Party with copies thereof;
 - (ii) each Party shall permit the other Party (or, where appropriate, the other Party's counsel) to review and comment on in advance any proposed written communications of any nature with Governmental Entities with respect to the Arrangement and provide the other Party (or, where appropriate, the other Party's counsel) with final copies thereof; and
 - (iii) neither Party shall participate in any meeting or discussion which it knows will be substantive (whether in person, by telephone or otherwise) with any Governmental Entity in respect of any filings, investigation or inquiry concerning the Arrangement unless it consults with the other Party in advance and gives the other Party the opportunity to attend and participate thereat (except to the extent that in any such case the Governmental Entity expressly requests that the other Party should not be present at the meeting or discussion or part or parts of the meeting or discussion).

Each Party agrees to consult with the other Party prior to issuing any press release or making any announcement regarding its mineral reserves or resources and, subject to applicable Laws, each Party agrees that it will not issue any press release or make any announcement regarding its mineral reserves or resources without the prior written consent of the other Party, such consent not to be unreasonably withheld

Covenants of Timberline

- 4.2 Timberline hereby covenants and agrees that it shall take such steps and do all such other acts and things, as may be necessary or desirable in order to give effect to the transactions contemplated by this Agreement, subject to shareholders' and regulatory approval, and, without limiting the generality of the foregoing, shall:
- (a) use its commercially reasonable efforts to, prior to the completion of the Arrangement, obtain listing on the Amex of the Timberline Shares to be issued pursuant to the Arrangement;
- (b) use its commercially reasonable efforts to obtain a conditional listing of its shares of common stock on the TSXV, including the Timberline Shares to be issued pursuant to the Arrangement, subject to standard conditions of the TSXV;

- (c) prepare, in consultation with the Staccato, and, subject to obtaining the Interim Order, file the Timberline Information Circular (which shall be form and substance satisfactory to Staccato, acting reasonably), together with any other documents required by applicable Laws (which shall be in form and substance satisfactory to Staccato, acting reasonably), in all jurisdictions where such Timberline Information Circular is required to be filed, including but not limited to, filing the Timberline Information Circular with the SEC under cover of Schedule 14A as a preliminary proxy statement under Regulation 14A of the U.S. Exchange Act, using commercially reasonable efforts to clear comments from the SEC, if any, and thereafter, filing the Timberline Information Circular with the SEC under cover of Schedule 14A as a definitive proxy statement under Regulation 14A of the Exchange Act, and, subject to obtaining the Interim Order and clearance of the Timberline Information Circular with the SEC and filing of the definitive proxy statement under Regulation 14A of the Exchange Act, mail the Timberline Information Circular in accordance with all applicable Laws to each Timberline Shareholder in all jurisdictions where the Timberline Information Circular is required to be mailed
- (d) subject to receipt of the Interim Order, duly call, give notice of, convene and hold the Timberline Meeting as soon as reasonably practicable and, in any event, provided that Staccato has provided to Timberline in a timely matter all information reasonably required by it for the preparation of the Timberline Information Circular, no later than May 21, 2010 or such later date as may be agreed to in writing;
- (e) subject to fiduciary duties of the board of directors, not take any steps to change or withdraw Timberline's board of director's recommendation of the Arrangement in a manner adverse to Staccato or which would impede the completion of the Arrangement, and not make a recommendation to Timberline Shareholders not to vote in favour of the Arrangement and will use all commercially reasonable efforts to obtain all other necessary securityholder approval;
- (f) ensure that the Timberline Information Circular complies in all material respects with all applicable Laws on the date of the mailing thereof and in the form and contain the information required by all applicable Laws, including all corporate and securities law requirements, and do not contain any misrepresentation (other than with respect to any information relating to and provided by or on behalf of Staccato or its subsidiaries or any third party that is not an affiliate of either Party);
- (g) not adjourn, postpone or cancel (or propose adjournment, postponement or cancellation of) the Timberline Meeting, without Staccato's prior written consent except as required by applicable laws or, in the case of adjournment, as may be required by Timberline Shareholders as expressed by majority resolution; cooperate in the preparation of the Interim Order, Final Order and Staccato Information Circular and ensure that the Staccato Information Circular shall contain prospectus-level disclosure respecting Timberline and the information and consolidated financial statements related to Timberline and the *pro forma* financial statements to be contained in the Staccato Information Circular shall be true, correct and complete in all material respects as they relate to Timberline and shall not contain any untrue statement of any material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein not misleading in light of the circumstances in which they are made and shall comply with applicable Laws;
- (h) carry out the terms of the Interim Order and the Final Order applicable to it and use its reasonable efforts to comply promptly with all requirements which applicable Laws may impose on Timberline or its subsidiaries with respect to the transactions contemplated hereby and by the Arrangement;

- (i) defend all lawsuits or other legal, regulatory or other proceedings to which it is a party challenging or affecting this Agreement or the consummation of the transactions contemplated hereby;
- (j) use its commercially reasonable efforts to have lifted or rescinded any injunction or restraining order or other order which may adversely affect the ability of the Parties to consummate the transactions contemplated hereby;
- (k) use commercially reasonable efforts to deliver or cause to be delivered to Staccato all certificates and legal, tax and other opinions necessary or, in the reasonable opinion of Staccato and its advisers, desirable, to support the disclosure contained or to be contained in the Staccato Circular;
- (l) subject to applicable Laws, Timberline will use its commercially reasonable efforts to cause the Supporting Shareholders to execute Support Agreements;
- (m) not request any amendment of, or request or accept any waiver of rights under, any of the Support Agreements, without the prior written consent of Staccato;
- (n) obtain all required certifications and consent of the auditors of Timberline in respect of the Timberline financial statements to be provided in the Staccato Information Circular;
- (o) make arrangements for the prompt delivery of certificates representing Timberline Shares, Timberline Options and Timberline Warrants to the Staccato Securityholders, as provided in the Plan of Arrangement;
- (p) if applicable, use commercially reasonable efforts to make and clear a filing required to be made in relation the HSR Act;
- (q) use commercially reasonable efforts to obtain all necessary waivers, consents and approvals required to be obtained by Timberline or a subsidiary of Timberline from other parties to material agreements of Timberline;
- (r) expressly assume and agree to perform the employment agreement between Staccato and Gary Edmondo dated May 2, 2008;
- (s) reserve sufficient shares of common stock for issuance of Timberline Shares upon completion of the Arrangement;
- (t) Timberline will use commercially reasonable efforts to negotiate with the holders of Staccato Options for the conversion of those Staccato Options into Timberline Options under Timberline's Benefit Plan with the same terms and conditions as the Staccato Options for which they are exchanged, subject to the adjustment of the exercise price thereof and adjustment to reflect the exchange rate of the Arrangement;
- (u) to use commercially reasonable efforts to file a registration statement or supplement with the SEC regarding the issuance of Timberline Shares underlying Timberline Warrants issued in exchange for Staccato Warrants pursuant to the terms and conditions of the Plan of Arrangement no later than the day before the Staccato Meeting and to use commercially reasonable efforts to bring such registration statement or supplement effective prior to the Effective Date; and
- (v) promptly notify Staccato if at any time it becomes aware that the Staccato Information Circular or Timberline Information Circular contains any misrepresentation or otherwise requires an amendment or supplement to the Staccato Information Circular or Timberline Information Circular or any related application and promptly deliver written notice to Staccato setting out full

particulars thereof. In any such event, Timberline shall cooperate with Staccato in the preparation of any required supplement or amendment to the Staccato Information Circular or Timberline Information Circular or such other document, as the case may be.

Covenants of Staccato

- 4.3 Staccato shall take such steps and to do all such other acts and things, as may be necessary or desirable in order to give effect to the transactions contemplated by this Agreement and, without limiting the generality of the foregoing, shall:
- (a) use its commercially reasonable best efforts to apply for and obtain such consents, orders or approvals as counsel for Timberline may advise are necessary or desirable for the implementation of the Arrangement and, without limiting the generality of the foregoing, to:
 - (i) apply for and obtain the Interim Order and the Final Order as provided in Section 2.2 hereof; and
 - (ii) obtain written consents from any persons who are parties to agreements (including without limiting the foregoing, any property agreements, option agreements, warrant agreements or warrant certificates) with Staccato where consents to the transactions contemplated by the Arrangement are required under those contracts or agreements;
- (b) in a timely and expeditious manner, file the Staccato Information Circular in all jurisdictions where the same is required in accordance with applicable law and provide the same to the Staccato Securityholders in accordance with applicable law or as required under exemption orders granted by appropriate regulatory authorities;
- (c) ensure that the Staccato Information Circular shall contain prospectus-level disclosure respecting Staccato and the information and consolidated financial statements related to Staccato contained in the Staccato Information Circular and any related documentation regarding Staccato to be distributed in connection with the solicitation of proxies by the management of Staccato in connection with the Staccato Meeting shall be true, correct and complete in all material respects and shall not contain any untrue statement of any material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein not misleading in light of the circumstances in which they are made and shall comply with applicable securities laws and the rules of the TSXV;
- (d) give Timberline timely opportunity to review and comment on the Initial Order, Final Order, Staccato Information Circular and other materials relating to the Staccato Meeting and all such documentation will be reasonably satisfactory to Timberline before it is filed or distributed to Staccato Securityholders, incorporation therein all reasonable comments made by Timberline and its counsel;
- (e) obtain all required certifications and consents of the auditors of Staccato in respect of the Staccato financial statements to be provided in the Staccato Information Circular;
- (f) convene and use commercially reasonable best efforts to hold the Staccato Meeting in accordance with the Interim Order for the purpose of considering the special resolutions to approve the Arrangement;
- (g) not adjourn, postpone or cancel (or propose adjournment, postponement or cancellation of) the Staccato Meeting, without Timberline's prior written consent except as required by applicable laws or, in the case of adjournment, as may be required by Staccato Securityholders as expressed by majority resolution;

- (h) subject to fiduciary duties of the board of directors, not take any steps to change or withdraw Staccato's board of director's recommendation of the Arrangement in a manner adverse to Timberline or which would impede the completion of the Arrangement, and not make a recommendation to Staccato Securityholders not to vote in favour of the Arrangement and will use all commercially reasonable efforts to obtain all other necessary securityholder approval;
- (i) cooperate in the preparation of the Timberline Information Circular and ensure that the Timberline Information Circular shall contain prospectus-level disclosure respecting Staccato and the Staccato Material Subsidiary and the information and consolidated financial statements related to Staccato and the *pro forma* financial statements to be contained in the Timberline Information Circular shall be true, correct and complete in all material respects as they relate to Staccato and shall not contain any untrue statement of any material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein not misleading in light of the circumstances in which they are made and shall comply with applicable Laws and any such financial statements that are to be included in the Timberline Information Circular shall be prepared in accordance with the requirements of Regulation 14A under the U.S. Exchange Act, including, if applicable, reconciliations to U.S. GAAP and auditor review and qualification standards under the U.S. Exchange Act and the Public Company Accounting Oversight Board;
- (j) use its commercially reasonable efforts to assist Timberline in obtaining a conditional listing of Timberline's shares of common stock on the TSXV, including the Timberline Shares to be issued pursuant to the Arrangement, subject to standard conditions of the TSXV;
- (k) carry out the terms of the Interim Order and the Final Order applicable to it and use its reasonable efforts to comply promptly with all requirements which applicable Laws may impose on Staccato or its subsidiaries with respect to the transactions contemplated hereby and by the Arrangement;
- (l) defend all lawsuits or other legal, regulatory or other proceedings to which it is a party challenging or affecting this Agreement or the consummation of the transactions contemplated hereby;
- (m) use its commercially reasonable efforts to have lifted or rescinded any injunction or restraining order or other order which may adversely affect the ability of the Parties to consummate the transactions contemplated hereby;
- (n) use commercially reasonable efforts to deliver or cause to be delivered to Timberline all certificates and legal, tax and other opinions necessary or, in the reasonable opinion of Timberline and its advisers, desirable, to support the disclosure contained or to be contained in the Timberline Information Circular;
- (o) subject to applicable Laws, Staccato will use its commercially reasonable efforts to cause the Supporting Shareholders to execute Support Agreements;
- (p) not request any amendment of, or request or accept any waiver of rights under, any of the Support Agreements, without the prior written consent of Timberline;
- (q) use commercially reasonable efforts to make and clear and filing required to be made in relation the Competition Act;
- (r) use its reasonable efforts to obtain all necessary waivers, consents and approvals required to be obtained by Staccato or a subsidiary of Staccato from other parties to material agreements of Staccato;

- (s) provide to holders of Staccato Options any necessary notice under Staccato's Benefit Plans regarding the Arrangement and the effect of the Arrangement on such holder's options;
- (t) provide to holders of Staccato Warrants any necessary notice under the Staccato Warrant certificates regarding the Arrangement and the effect of the Arrangement on such holder's Warrants;
- (u) use commercially reasonable efforts to obtain a fairness opinion from Paradigm Capital Inc. stating that that as of the date of this Agreement, the consideration to be paid under the Arrangement is fair, from a financial point of view, to Staccato and the Staccato Shareholders;
- (v) use commercially reasonable efforts to permit Staccato Options that are not part of the Plan of Arrangement pursuant to Section 2.1 above to be accelerated and to permit net exercise (cashless exercise) of such Staccato Options at the in the money amount for Staccato Shares;
- (w) promptly notify Timberline if at any time it becomes aware that the Staccato Information Circular or Timberline Information Circular contains any misrepresentation or otherwise requires an amendment or supplement to the Staccato Information Circular or Timberline Information Circular or any related application and promptly deliver written notice to Timberline setting out full particulars thereof. In any such event, Staccato shall cooperate with Timberline in the preparation of any required supplement or amendment to the Staccato Information Circular or Timberline Information Circular or such other document, as the case may be
- (x) not dispose of an interest in any of its material properties or otherwise enter into any material transaction with, or incur any material liability to, any other corporation or person or agree to do any of the foregoing or perform any act or enter into any transaction or negotiation which interferes or is inconsistent with the completion of the transactions contemplated hereby, other than as contemplated in this Agreement, without the written consent of Timberline thereto;

(y) at Closing, have:

- (i) an authorized capital of an unlimited number of common shares without par value of which 99,842,793 common shares are duly issued and outstanding as fully paid and non-assessable (save as may be increased by the issuance of common shares upon the exercise of outstanding convertible securities in (ii)); and
- (ii) 48,122,059 common shares issuable on exercise of all outstanding convertible securities of Staccato (save as may be decreased by the issuance of common shares upon the exercise of such outstanding convertible securities), including Staccato Options and Staccato Warrants of which there are no common shares issuable upon exercise of in-themoney convertible securities;

save as may be altered by the exercise of outstanding convertible securities, the exercise of dissent rights under the Plan of Arrangement, or consented to in writing by Timberline; and

- (z) not incur between the date of this Agreement and the Closing Date, any expenses or liabilities otherwise than in the ordinary course of its business (which ordinary course shall include obligations under existing employment or management agreements) or in connection with its obligations under this Agreement and without limiting the generality of the foregoing, not without prior written consent of Timberline, such consent not to be unreasonably withheld:
 - (i) other than in the normal course of business or except in accordance with existing agreements, sell, pledge, lease, dispose of, grant any interest in, encumber or agree to sell, pledge, lease, dispose of, grant any interest in or encumber (or permit any of its

- subsidiaries to sell, pledge, lease, dispose of, grant any interest in, encumber or agree to sell, pledge, lease, dispose of, grant any interest in or encumber) any material assets of Staccato;
- (ii) redeem, purchase or offer to purchase (or permit any of its subsidiaries (if any) to redeem, purchase or offer to purchase) any common shares or other securities of Staccato:
- (iii) other than in the normal course of business or except in accordance with existing agreements, acquire, directly or indirectly (or permit any of its subsidiaries (if any) to acquire directly or indirectly) any material assets, including but not limited to mining properties or interests therein or securities of other companies or enter into any joint venture, earn-in or similar agreements or arrangements; or
- (iv) enter into or modify any employment, severance, collective bargaining or similar agreements, policies or arrangements with, or grant any bonuses, salary increases, severance or termination pay to, any employees, officers or directors of Staccato other than pursuant to agreements in effect (without amendment) on the date hereof.

Preparation of Filings

- 4.4 Staccato and Timberline shall cooperate in:
 - (i) the preparation and filing of any application for the orders and the preparation of any required documents reasonably deemed by Staccato or Timberline to be necessary to discharge their respective obligations under applicable Securities Legislation in connection with the Arrangement and the other transactions contemplated hereby;
 - (ii) the taking of all such action as may be required under applicable Securities Legislation in connection with the issuance of the Timberline Shares in connection with the Arrangement; provided, however, that neither Timberline nor Staccato shall be required to register or qualify as a foreign corporation or to take any action that would subject it to service of process in any jurisdiction where such entity is not now so subject, except as to matters and transactions arising solely from the offer and sale of the Timberline Shares; and
 - (iii) the taking of all such action as may be required under the Laws in connection with the transactions contemplated by this Agreement and the Plan of Arrangement.

5.0 <u>CONDITIONS PRECEDENT</u>

Mutual Conditions Precedent

- 5.1 The parties' obligations to complete the transactions contemplated in this Agreement are subject to satisfaction of the following conditions on or before the Closing Date:
- (a) all necessary approvals of Staccato Securityholders by the requisite majorities will have been obtained in respect of the Arrangement at the Staccato Meeting in accordance with applicable laws, the Interim Order and the Plan of Arrangement by May 21, 2010 or such other date as may be agreed to be Staccato and Timberline;
- (b) the Interim Order, the Final Order and all necessary orders of the Court with respect to the Arrangement will have been obtained and shall not have been set aside or modified in a manner that is not acceptable to any party, acting reasonably;

- (c) all necessary approvals of Timberline Shareholders by the requisite majorities will have been obtained in respect of the Arrangement at the Timberline Meeting in accordance with applicable Amex rules by May 21, 2010 or such other date as may be agreed to be Staccato and Timberline;
- (d) the Timberline Shares to be issued pursuant to the Arrangement will, on or before the Effective Time, be listed on the Amex and Timberline will be in compliance with all rules, regulations and policies of the Amex in all material respects;
- (e) the shares of common stock of Timberline, including the Timberline Shares to be issued pursuant to the Arrangement, will, on or before the Effective Time, be conditionally approved for listing on the TSX-V, subject to standard conditions of the TSX-V;
- (f) all other consents, orders, regulations and approvals, including regulatory and judicial approvals and orders, necessary for the completion of the transactions provided for in this Agreement and the Plan of Arrangement shall have been obtained or received from the persons, authorities or bodies having jurisdiction in the circumstances;
- (g) there shall not be in force any order or decree restraining or enjoining the consummation of the transactions contemplated by this Agreement or the Arrangement;
- (h) no provision of any applicable Laws and no judgement, injunction, order or decree shall be in effect which restrains or enjoins or otherwise prohibits the consummation of the Arrangement or the transactions contemplated by this Agreement;
- (i) there shall not be pending or threatened any suit, action or proceeding by any Governmental Entity or other Person, in each case that has a reasonable likelihood of success, (i) seeking to prohibit or restrict the acquisition by Timberline of any Staccato Securities, or seeking to restrain or prohibit the consummation of the Arrangement, (ii) seeking to prohibit or materially limit the ownership or operation by Timberline or any of its subsidiaries of any material portion of the business or assets of Timberline, Staccato or any of their respective subsidiaries or to compel Timberline or any of its subsidiaries to dispose of or hold separate any material portion of the business or assets of Timberline, Staccato or any of their respective subsidiaries as a result of the Plan of Arrangement, (iii) seeking to impose limitations on the ability of Timberline or any of its subsidiaries to acquire or hold, or exercise full rights of ownership of, any Staccato Securities, including the right to vote the Staccato Shares purchased by it on all matters properly presented to the Staccato Shareholders, (iv) seeking to prohibit Timberline or any of its subsidiaries from effectively controlling in any material respect the business or operations of Staccato and its of its subsidiaries or (v) imposing any condition or restriction that, in the reasonable opinion of the Party seeking to invoke this condition, would be materially burdensome to the future operations of Timberline or Staccato after the Effective Time:
- (j) this Agreement will not have been terminated; and
- (k) the Effective Date will have occurred on or before the Termination Date, unless otherwise agreed to by the Parties.

Conditions solely for the benefit of Timberline

- 5.2 The obligations of Timberline to complete the transactions contemplated in this Agreement are subject to satisfaction of the following conditions on or before the Closing Date:
- (a) no Material Adverse Change will have occurred in the business, affairs, financial condition or operations of Staccato;

- (b) dissent rights to the Arrangement shall not have been exercised prior to the Effective Date by Staccato Shareholders representing in the aggregate 7% or more of the total number of Staccato Shares outstanding at such time;
- (c) the issue of Timberline Shares and the other Timberline Securities to be issued pursuant to the Arrangement will be exempt from the registration requirements of the U.S. Securities Act and the registration and prospectus requirements of applicable securities laws in each of the provinces and territories of Canada in which Staccato Securityholders are resident;
- (d) there shall not have been an amendment to the U.S. Securities Act, a change in the U.S. Securities and Exchange Commission's interpretation of the U.S. Securities Act or a decision of a court which provides that orders of Canadian courts such as the Final Order do not qualify under Section 3(a)(10) of the U.S. Securities Act which results in the Section 3(a)(10) Exemption being not available for any reason to exempt the issuances of Timberline Shares and other securities of Timberline issued on completion of the Arrangement from the registration requirements of the U.S. Securities Act;
- (e) each of the representations and warranties of Staccato under this Agreement, shall be true and correct in all material respects on the date of this Agreement and as of the Effective Date as if made on and as of such date (except for such representations and warranties made as of specified date, which will be true and correct in all material respects as of such specified date);
- (f) Timberline shall have received a certificate of a senior officer of Staccato confirming that the representations and warranties of Staccato set out in Section 3.2 are true and correct in all material respects on and as of Closing;
- (g) Staccato shall have performed or complied with, in all material respects, each of its obligations, covenants and agreements hereunder to be performed and complied with it on or before the Effective Time:
- (h) Timberline shall have received a certificate of a senior officer of Staccato confirming that the obligations, covenants and agreements of Staccato set out in Section 4.3 have been completed as at the Closing Date;
- (i) receipt of the resignations of Grant Ewing, Robert Lipsett, Greg Hryhorchuk, Christopher Pollard and George Brazier as directors of Staccato;
- (j) in accordance with the terms of their employment agreements, receipt of resignations of Grant Ewing, Robert Lipsett and Greg Hryhorchuk as officers of Staccato; and
- (k) receipt of an opinion of counsel for Staccato, dated as of the Effective Date and addressed to Timberline relating to the title of its material properties and such opinion shall be in form acceptable to Timberline and its counsel, acting reasonably.

The foregoing conditions in this Section 5.2 are inserted for the exclusive benefit of Timberline and may be waived by it in whole or in part at any time.

Conditions solely for the benefit of Staccato

- 5.3 The obligations of Staccato to complete the transactions contemplated in this Agreement are subject to satisfaction of the following conditions on or before the Closing Date:
- (a) no Material Adverse Change will have occurred in the business, affairs, financial condition or operations of Timberline;

- (b) each of the representations and warranties of Timberline under this Agreement, shall be true and correct in all material respects on the dated of this Agreement and as of the Effective Date as if made on and as of such date (except for such representations and warranties made as of specified date, which will be true and correct in all material respects as of such specified date);
- (c) Staccato shall have received a certificate of a senior officer of Timberline confirming that the representations and warranties of Timberline set out in Section 3.1 are true and correct in all material respects on and as of Closing;
- (d) Timberline shall have performed or complied with, in all material respects, each of its obligations, covenants and agreements hereunder to be performed and complied with it on or before the Effective Time:
- (e) Staccato shall have received a certificate of a senior officer of Timberline confirming that the obligations, covenants and agreements of Timberline set out in Section 4.2 have been completed as at the Closing Date;
- (f) the issue of Timberline Shares and other Timberline Securities pursuant to the Arrangement will have been approved by all necessary corporate action to permit such shares to be issued as fully paid and non-assessable and the issue of such Timberline Shares and the other Timberline Securities to be issued pursuant to the Arrangement will be exempt from the registration requirements of the U.S. Securities Act and the registration and prospectus requirements of applicable securities laws in each of the provinces and territories of Canada in which holders of securities of Staccato are resident; and
- (g) the board of directors of Timberline will be comprised of seven directors, all of whom will be nominated by Timberline.
- (h) receipt of an opinion of counsel for Timberline, dated as of the Effective Date and addressed to Staccato relating to the title of its material properties and such opinion shall be in form acceptable to Staccato and its counsel, acting reasonably.

The foregoing conditions in this Section 5.3 are inserted for the exclusive benefit of Staccato and may be waived by it in whole or in part at any time.

6.0 AMENDMENT, CLOSING AND TERMINATION

Amendment

- 6.1 This Agreement and the Plan of Arrangement may, at any time and from time to time before the Effective Date, be amended by written agreement of the parties hereto without, subject to applicable law, further notice to or authorization on the part of their respective shareholders. Without limiting the generality of the foregoing, any such amendment may:
- (a) change the time for performance of any of the obligations or acts of the parties hereto;
- (b) waive any inaccuracies or modify any representation contained herein or any document to be delivered pursuant hereto;
- (c) waive compliance with or modify any of the covenants herein contained or waive or modify performance of any of the obligations of the parties hereto; or
- (d) amend the terms of Section 3.01 of the Plan of Arrangement and Subsections 5.1(a), (b), (c), (f) and (g) of this Agreement and the sequence of transactions described in the Plan of Arrangement

subject to any required approval of the shareholders of Staccato, given in the same manner as required for the approval of the Arrangement or as may be ordered by the Court.

6.2 This Agreement and the Schedules hereto may be amended in accordance with the Final Order, but if the terms of the Final Order require any such amendment, the rights of the parties hereto under Sections 5.1, 5.2, 5.3, 6.1, 6.2 and 6.4 shall remain unaffected.

Closing

6.3 The completion of the Arrangement (the "Closing") will be at the offices of Miller Thomson LLP, 840 Howe Street, 10th Floor, Vancouver, British Columbia V6Z 2M1, on June 15, 2010 (the "Closing Date"), or such other place or date as may be mutually agreed by the parties, provided it is not later than September 30, 2010. At the Closing, parties will exchange documents to effect the Closing including documents to confirm the matters set out in Article Five of the Plan of Arrangement and to complete the Arrangement and related matters as contemplated hereby.

Termination

- 6.4 This Agreement may be terminated at any time prior to the Effective Time, whether before or after the requisite approval of the shareholders of Staccato or Timberline:
 - (a) by mutual written consent duly authorized by the Boards of Directors of Timberline and Staccato;
 - (b) by either Staccato or Timberline, if the Arrangement shall not have been consummated by September 30, 2010 for any reason (the "**Termination Date**"); *provided, however*, that the right to terminate this Agreement under this Section 6.4(b) shall not be available to any party whose action or failure to act has been a principal cause of or resulted in the failure of the Arrangement to occur on or before such date and such action or failure to act constitutes a breach of this Agreement;
 - (c) by either Staccato or Timberline, if there shall be passed any Law that makes the consummation of the Arrangement illegal or otherwise prohibited, or if a Governmental Entity shall have issued an order, decree or ruling or taken any other action, in any case having the effect of permanently restraining, enjoining or otherwise prohibiting the Arrangement, which order, decree, ruling or other action is final and nonappealable;
 - (d) by either Staccato or Timberline, if the required approvals of the Staccato Securityholders contemplated by this Agreement shall not have been obtained by reason of the failure to obtain the required vote at a meeting of Staccato Securityholders duly convened therefor or at any adjournment thereof or the Staccato Shareholders exercise dissenter rights in excess of the threshold set forth in Section 5.2(b); *provided, however*, that the right to terminate this Agreement under this Section 6.4(d) shall not be available to Staccato where the failure to obtain the approval of the Staccato Securityholders shall have been caused by the action or failure to act of Staccato and such action or failure to act constitutes a breach by Staccato of this Agreement;
 - (e) by either Staccato or Timberline, if the required approval of the Timberline Shareholders contemplated by this Agreement shall not have been obtained by reason of the failure to obtain the required vote at a meeting of Timberline Shareholders duly convened therefore or at any adjournment thereof; provided, however, that the right to terminate this Agreement under this section 6.4(e) shall not be available to Timberline where the failure to obtain the approval of the Timberline Shareholders shall have been caused by the

- action or failure to act of and such action or failure to act constitutes a breach by Timberline of this Agreement.
- (f) by Staccato, if Timberline is in breach of any of its covenants made in this Agreement, which breach individually or in the aggregate causes or would reasonably be expected to cause a Material Adverse Effect with respect to Timberline or materially impedes or would reasonably be expected to materially impede the completion of the Arrangement, and Timberline fails to cure such breach within 10 business days after receipt of written notice from Staccato;
- (g) by Timberline, if Staccato is in breach of any of its covenants made in this Agreement, which breach individually or in the aggregate causes or would reasonably be expected to cause a Material Adverse Effect with respect to Staccato or materially impedes or would reasonably be expected to materially impede the completion of the Arrangement, and Staccato fails to cure such breach within 10 business days after receipt of written notice from Timberline;
- (h) by Staccato, if the Board of Directors of Timberline or any committee thereof shall for any reason have withheld, withdrawn or shall have amended, modified or changed in a manner adverse to Staccato its recommendation in favor of, the adoption and approval of the Agreement or the approval of the transactions contemplated by this Agreement;
- (i) by Timberline, if (i) the Board of Directors of Staccato or any committee thereof shall for any reason have withheld, withdrawn or shall have amended, modified or changed in a manner adverse to Timberline its recommendation in favor of, the adoption and approval of the Agreement or the approval of the transactions contemplated by this Agreement; (ii) the Board of Directors of Staccato or any committee thereof shall have approved or recommended any Acquisition Proposal; (iii) the Board of Directors of Staccato shall have failed to re-affirm its recommendation in favor of the adoption and approval of this Agreement and the approval of the transactions contemplated by this Agreement promptly following the request of Timberline to do so; (iv) Staccato shall have entered into any letter of intent or similar document or any agreement, contract or commitment accepting any Acquisition Proposal; (v) Staccato shall have breached its obligations under the terms of Section 8.1 hereof; or (vi) a tender or exchange offer relating to securities of Staccato shall have been commenced by a person unaffiliated with Timberline and Staccato shall not have sent to its securityholders pursuant to section 2.17 of Multilateral Instrument 62-104 Take-over Bids and Issuer Bids (and the equivalent provisions in other Canadian provincial securities regulations), within fifteen (15) days after such tender or exchange offer is first published, sent or given, a statement disclosing that Staccato recommends rejection of such tender or exchange offer;
- (j) by Staccato, upon the determination by Staccato's board of directors at the conclusion of the process set out in Section 8.1 that the Acquisition Proposal constitutes a Superior Proposal; *provided that* Staccato is not otherwise then in breach or default of its obligations hereunder and subject to the payment of the Staccato Break Fee payable to Timberline under Section 8.2; or
- (k) by Timberline if the conditions precedent to closing the Arrangement in Section 5.2 have not been satisfied by September 30, 2010, *provided*, *however*, that the right to terminate this Agreement under this Section 6.4(k) shall not be available to Timberline, if Timberline's actions or failure to act has been a principal cause of or resulted in the failure of the condition precedent to be satisfied and such action or failure to act constitutes a breach of this Agreement;

(1) by Staccato if the conditions precedent to closing the Arrangement in Section 5.3 have not been satisfied by September 30, 2010, *provided*, *however*, that the right to terminate this Agreement under this Section 6.4(k) shall not be available to Staccato, if Timberline's actions or failure to act has been a principal cause of or resulted in the failure of the condition precedent to be satisfied and such action or failure to act constitutes a breach of this Agreement.

7.0 INDEMNIFICATION/INSURANCE

Indemnification

- 7.1 Timberline agrees that rights to indemnification for acts or omissions occurring prior to the Effective Time existing as of the date of this Agreement in favour of the directors or officers of Staccato as provided in constating documents in effect of the date of this Agreement, shall survive the Arrangement and shall continue in full force and effect until the earlier of the expiration of the applicable statute of limitations with respect to any claims against directors or officers of Staccato arising out of such acts or omissions and the sixth anniversary of the Effective Date, and Timberline hereby assumes, effective on consummation of the Arrangement, all such liability with respect to any matters arising prior to the Effective Time.
- 7.2 If this Agreement is terminated, each party (the "Indemnifying Party") hereto undertakes with the other parties hereto (the "Indemnified Party") to hold the Indemnified Party fully and effectually indemnified from and against all losses, claims, damages, liabilities, actions or demands (including amounts paid in any settlement approved by the Indemnifying Party of any action, suit, proceeding or claim but excluding lost profits and consequential damages), to which such Indemnified Party may become subject insofar as such losses, claims, damages, liabilities, actions or demands arise out of orany action, suit, proceeding or claim brought by any Third Party ("Third Party Claim") and which are based upon any breach of a representation, warranty, covenant or obligation of the Indemnifying Party contained in this Agreement or any certificate or notice delivered by it in connection herewith, and will reimburse such Indemnified Party for any legal or other expenses reasonably incurred by such Indemnified Party in connection with investigating or defending any such Third Party Claim.

Defence

- 7.3 (a) Promptly after receipt by an Indemnified Party of notice of a possible Third Party Claim referred to in Section 7.1 hereof, such Indemnified Party, if a claim in respect thereof is to be made against the Indemnifying Party under such Section, shall provide the Indemnifying Party with written particulars thereof; provided that failure to provide the Indemnifying Party with such particulars shall not relieve such Indemnifying Party from any liability which it might have on account of the indemnity provided for in this Section 7 except insofar as such failure shall prejudice such Indemnifying Party. The Indemnified Party shall also provide to the Indemnifying Party copies of all relevant documentation and, unless the Indemnifying Party assumes the defence thereof, shall keep such Indemnifying Party advised of the progress thereof and will discuss with the Indemnifying Party all significant actions proposed.
 - (b) An Indemnifying Party shall be entitled, at its own expense, to participate in (and, to the extent that it may wish, to assume) the defence of any such Third Party Claim but such defence shall be conducted by counsel of good standing approved by the Indemnified Party, such approval not to be unreasonably withheld. Upon the Indemnifying Party notifying the Indemnified Party of its election so to assume the defence and retaining such counsel, the Indemnifying Party shall not be liable to the Indemnified Party for any legal or other expenses subsequently incurred by it in connection with such defence other than for reasonable costs of investigation. If such defence is assumed by the

Indemnifying Party, it shall, through the course thereof, provide copies of all relevant documentation to the Indemnified Party, keep such Indemnified Party advised of the progress thereof and shall discuss with the Indemnified Party all significant actions proposed. No Indemnifying Party shall enter into any settlement without the consent of the Indemnified Party, but such consent shall not be unreasonably withheld. If such defence is not assumed by the Indemnifying Party, the Indemnifying Party shall not be liable for any settlement made without its consent, but such consent shall not be unreasonably withheld.

(c) Notwithstanding the foregoing, an Indemnified Party shall have the right, at the Indemnifying Party's expense, to employ counsel of its own choice in respect of the defence of any such Third Party Claim if (i) the employment of such counsel has been authorized by the Indemnifying Party in connection with such defence; or (ii) counsel retained by the Indemnifying Party or the Indemnified Party shall have advised the Indemnified Party that there may be legal defences available to it which are different from or in addition to those available to the Indemnifying Party (in which event and to that extent, the Indemnifying Party shall not have the right to assume or direct the defence on behalf of the Indemnified Party) or that there may be a conflict of interest between the Indemnifying Party and the Indemnified Party; or (iii) the Indemnifying Party shall not have assumed such defence and employed counsel therefor within a reasonable time after receiving notice of such Third Party Claim.

Insurance

5.4 Staccato will be permitted to procure before the Effective Time prepaid non-cancellable run-off directors and officers liability insurance providing coverage (on terms comparable to those contained in Staccato's current insurance policies) for six years from the Effective Time with respect to claims arising from or related to facts or events that occurred before the Effective Time in the same manner as such claims would have been covered if they were before the Effective Time, provided however that Timberline will not be required to pay a premium in excess of \$60,000 in order to procure such insurance coverage, and if such insurance coverage is only available at a premium in excess of \$60,000, Timberline will obtain coverage for the maximum period available at a premium of \$60,000.

8.0 STANDSTILL AGREEMENT AND BREAK FEE PAYMENT

8.1 From the date hereof until the Effective Time or, if earlier, the termination of this Agreement, Staccato shall immediately cease and cause to be terminated any existing discussions with any person other than Timberline (a "Third Party") that relate to any Acquisition Proposal, and Staccato shall not, nor shall it permit or authorize its subsidiaries, or any of their respective officers, directors or employees, or any investment banker, attorney or other advisor or representative retained by any of them to, directly or indirectly: (a) solicit, initiate, encourage or otherwise knowingly facilitate any Acquisition Proposal or any inquiries or proposals relating thereto; (b) engage in discussions or negotiations with, or disclose any non-public information relating to Staccato or its subsidiaries or afford access to the properties, books or records of Staccato or its subsidiaries to, any Person (other than Timberline or any designees of Timberline or as may otherwise be required by law) concerning or in connection with an Acquisition Proposal; (c) withhold, withdraw, modify or change, or publicly propose to do so, in a manner adverse to Timberline, or fail to make its recommendation to vote in favor of the Arrangement or approve, endorse or recommend an Acquisition Proposal; or (d) accept or enter into, or publicly propose to accept or enter into, any letter of intent, agreement, arrangement or understanding related to any Acquisition Proposal; provided, however, that in each case, if (A) after the date of this Agreement and prior to the date of any securityholder approval of the Arrangement, an unsolicited, bona fide written Acquisition Proposal is made to Staccato and is not withdrawn; (B) Staccato's Board of Directors reasonably believes in good faith, after consultation with Staccato's financial advisor, that such Acquisition Proposal constitutes (if consummated as proposed) a Superior Proposal; (C) Staccato's Board of Directors reasonably believes in

good faith, after consultation with Staccato's outside legal counsel, that the failure to engage in such negotiations or discussions, to provide such information, or to withhold, withdraw, amend, modify, or change its recommendation to vote in favor of the Arrangement would be inconsistent with the duties of the Board of Directors of Staccato; (D) prior to furnishing any such nonpublic information to, entering into discussions or negotiations with, any Person, Staccato receives from such Person an executed confidentiality agreement (including "standstill" provisions) no less favorable to Staccato than the Confidentiality Agreement; and (E) at the time of or prior to furnishing any such nonpublic information to such Person, Staccato furnishes such nonpublic information to Timberline (to the extent such nonpublic information has not been previously furnished by Staccato to Timberline), then Staccato may (1) furnish information with respect to Staccato and its subsidiaries to such Person, (2) participate with such Person in negotiations regarding such Acquisition Proposal, (3) enter into discussions or negotiations with, such Person, (4) withhold, withdraw, modify or change in a manner adverse to Timberline, or fail to make, its recommendation to vote in favor of the Arrangement, or (5) accept, recommend, approve or enter into an agreement to implement a Superior Proposal (provided that Staccato shall have complied with the provisions of Section 8.1 and paid the Staccato Break Fee).

Staccato shall promptly after receipt of any Acquisition Proposal provide Timberline with a copy of any written Acquisition Proposal and the identity of the Person making such Acquisition Proposal and a written statement with respect to any non-written Acquisition Proposal received, which statement shall include the identity of the Person making the Acquisition Proposal and a reasonably detailed description of all the material terms thereof. Staccato shall promptly advise Timberline of any material modification or proposed modification thereto. Staccato shall not release or permit the release of any Person from, or waive or permit the waiver of any provision of, any confidentiality, "standstill" or similar agreement (other than as required pursuant to the terms thereof as in effect on the date hereof) under which Staccato or any of its subsidiaries has any rights, or fail to use reasonable best efforts to enforce or cause to be enforced each such agreement at the request of Timberline. Staccato shall use its reasonable best efforts to ensure that its subsidiaries and any of their respective officers, directors or employees or any investment banker, attorney or other advisor or representative retained by any of them are aware of the provisions of this Section 8.1, and shall be responsible for any breach of this Section 8.1 by it and its subsidiaries and any of their respective officers, directors or employees or any investment banker, attorney or other advisor or representative retained by any of them.

In the event that Staccato's Board of Directors has determined that any Acquisition Proposal constitutes, or may reasonably be expected to be a Superior Proposal, Staccato shall promptly provide to Timberline a copy of any such written Acquisition Proposal received by it and Timberline shall have the right of first refusal, exercisable within 72 hours from the date of such notice, to amend this Agreement, and Staccato shall sign any amendment to this Agreement that contains terms that are no less favourable than those contained in such written Acquisition Proposal.

- 8.2 If Staccato terminates this Agreement under Section 6.4(f) or 6.4(h), Timberline will pay Staccato a break fee of \$750,000 (the "**Staccato Break Fee**").
- 8.3 If (a) Timberline terminates this Agreement under Section 6.4(g) or 6.4(i) or, (b) Staccato terminates this Agreement pursuant to Section 6.4(j), Staccato will pay Timberline of a break fee of \$750,000 (the "**Timberline Break Fee**").
- 8.4 If this Agreement is terminated pursuant to Section 6.4(d), Staccato will pay Timberline an expense reimbursement fee of \$100,000 (the "**Timberline Expense Reimbursement Fee**").
- 8.5 If this Agreement is terminated pursuant to Section 6.4(e), Timberline will pay Staccato an expense reimbursement fee of \$100,000 (the "Staccato Expense Reimbursement Fee").
- 8.6 Each Party acknowledges that all of the payment amounts set out in this Section 8.0 are payments of liquidated damages which are a genuine pre-estimate of the damages Timberline or Staccato will suffer

or incur as a result of the event giving rise to such damages and resultant termination of this Agreement and are not penalties. Each Party irrevocably waives any right it may have to raise as a defence that any such liquidated damages are excessive or punitive. For greater certainty, the Parties agree that the payment of any amounts pursuant to this Section 8.0 is the sole monetary remedy of Timberline and Staccato, provided, however, that this limitation shall not apply in the event of fraud, intentional misrepresentation or wilful breach of this Agreement by a Party.

- 8.7 For the purposes of this Agreement, "Superior Proposal" means an unsolicited bona fide offer regarding the direct or indirect acquisition or disposition of all or any of Staccato's securities, or any amalgamation, merger, sale of all of Staccato 's assets or, except in the ordinary course of business, any part of Staccato's assets, take-over bid, tender offer, plan of arrangement, issuer bid, reorganization, dividend or distribution, recapitalization, liquidation or winding-up of, or other business combination or similar transaction involving Staccato, all of Staccato's assets or, except in the ordinary course of business, any part of Staccato's assets or similar fundamental transaction involving Staccato which the board of directors of Staccato considers, in good faith, to be superior to the terms of the Arrangement and must be recommended to Staccato's shareholders in order that the board of directors may discharge its fiduciary obligations. Any good faith determination by the board of directors of Staccato of a Superior Proposal shall only be made after consultation with its financial and legal advisors and a determination by the board of directors of Staccato that the failure to entertain and negotiate such a Superior Proposal or to furnish information concerning Staccato to a Third Party in connection therewith could, in the particular circumstances, result in a finding that the directors had breached their fiduciary duties under applicable law.
- 8.8 Nothing contained in this Agreement shall prohibit or prevent Staccato or its boards of directors or officers from: (a) making any disclosure of or in relation to an unsolicited Acquisition Proposal from a Third Party or, prior to the Effective Time if, in the good faith judgment of the board of directors, after consultation with outside legal counsel, such disclosure is necessary for the directors or officers of Staccato to act in a manner consistent with their fiduciary duties or is otherwise required under applicable laws; (b) responding, within the time and in the manner required by applicable laws, to any unsolicited take-over bid or tender or exchange offer made for Staccato Shares or any other securities of Staccato; or (c) taking any other action in relation to an unsolicited Acquisition Proposal from a Third Party to the extent required under applicable securities laws or orders or otherwise mandated by any governmental entity.
- 8.9 Subject to the foregoing provisions of this Section 8.0, each of Timberline and Staccato shall pay all the fees, costs and expenses incurred by such Party in connection with the Agreement and the Arrangement.
- 8.10 If a fee is payable under Section 8.2, 8.3. 8.4 or 8.5, no other fee will be payable under Section 8.0, except in the case of fraud or intentional misrepresentation by a party, in which case the other party can seek any available legal remedies, whether in law or equity, in relation to such fraud or intentional misrepresentations.

9.0 ORDINARY COURSE

9.1 Until the earlier of the closing of the Arrangement and the termination of this Agreement without completion of the Arrangement, Staccato will not without the prior written consent of the other parties, enter into any contract in respect of its business or assets, other than in the ordinary course of business, and each of the parties will continue to carry on its business and maintain its assets in the ordinary course of business, with the exception of reasonable costs incurred in connection with the Arrangement, and, without limitation, but subject to the above exceptions, will maintain payables and other liabilities at levels consistent with past practice and will not engage in any extraordinary material transactions or agree to do any of the foregoing or perform any act or enter into any transaction or negotiation which interferes or is inconsistent with the completion of the transactions contemplated hereby without the prior written

consent of the other parties.

10.0 PUBLIC DISCLOSURE AND CONFIDENTIALITY

- 10.1 No disclosure or announcement, public or otherwise, in respect of this Agreement or the transactions contemplated herein will be made by any party without the prior written agreement of the other parties as to timing, content and method, provided that the obligations herein will not prevent any party from making, after consultation with the other parties, such disclosure as its counsel advises is required by applicable laws or the rules and policies of the reporting jurisdictions of the party.
- 10.2 Unless and until the transactions contemplated in this Agreement have been completed, except with the prior written consent of the other parties, each party and their respective employees, officers, directors, shareholders, agents, advisors and other representatives will hold all information received from the other parties in strictest confidence, except such information and documents which (i) are or subsequently may become generally available to the public; (ii) are required to be disclosed by applicable law; (iii) are available on a non-confidential basis prior to their disclosure to the other parties; (iv) become available to one party on a non-confidential basis from a source other than the other parties provided that such other source is not bound by a confidentiality agreement with the other parties; (v) are independently developed; or (vi) were available to each party as a result of the relationship of the parties prior to the date hereof.
- 10.3 All such information in written form and documents will be returned to the party originally delivering them in the event that the transactions provided for in this Agreement are not completed.

11.0 <u>ASSIGNMENT</u>

11.1 No party may assign its rights or obligations under this Agreement.

12.0 WAIVER

12.1 Any waiver or release of any conditions of this Agreement, to be effective, must be in writing executed by the party for whom such condition is expressed by this Agreement to benefit.

13.0 GENERAL

- 13.1 The covenants, representations and warranties contained herein will survive the Closing.
- 13.2 Time is of the essence herein.
- 13.3 Each party hereto will, from time to time, at the request of the other parties, do such further acts and execute and deliver all such further documents, agreements and instruments as will be reasonably required in order to fully perform and carry out the terms, conditions and intent of this Agreement.
- 13.4 All references to currency are references to Canadian dollars unless otherwise indicated.
- 13.5 The parties intend that this Agreement will be binding upon them until terminated.
- 13.6 Any notice to be given hereunder to the parties will be deemed to be validly given if delivered, or if sent by facsimile:

if to Timberline, to:

Timberline Resources Corporation 101 E. Lakeside Avenue Coeur D'Alene, Idaho 83814 Attention: Randal Hardy, Chief Executive Officer

Facsimile No.: (208) 664-4860

with a copy to:

Dorsey & Whitney LLP TD Canada Trust Tower Brookfield Place 161 Bay Street, Suite 4310 Toronto, ON Canada M5J 2S1 Attention: Kenneth Sam Facsimile No.: (416) 367-7371

and with a copy to:

Miller Thomson LLP Robson Court 1000-840 Howe Street Vancouver, BC V6Z 2M1 Attention: Peter McArthur Facsimile No.: (604) 643-1200

if to Staccato, to:

Staccato Gold Resources Ltd. 2060-777 Hornby Street Vancouver, BC V6Z 1T7 Attention: Grant Ewing, President Facsimile No.: (604) 484-8254

with a copy to:

Stikeman Elliott LLP Suite 1700 666 Burrard Street Vancouver, BC V6C 2X8 Attention: John E. Stark Fax: (604) 681-1825

and any such notice delivered on a business day in accordance with the foregoing will be deemed to have been received on the date of delivery or facsimile transmission.

- 13.7 This Agreement and the rights and obligations of the parties hereunder will be governed by and construed according to the laws of the Province of British Columbia.
- 13.8 This Agreement will enure to the benefit of and be binding upon the parties hereto and their successors.

13.9 This Agreement may be executed in any number of counterparts with the same effect as if all parties had signed the same document. All of these counterparts will for all purposes constitute one agreement, binding on the parties, notwithstanding that all parties are not signatories to the same counterpart. A fax transcribed copy or photocopy of this Agreement executed by a party in counterpart or otherwise will constitute a properly executed, delivered and binding agreement or counterpart of the executing party.

IN WITNESS WHEREOF the parties hereto have executed this Agreement as of the year and day set out on page 1 hereof.

TIMBERLINE RESOURCES CORPORATION
Per: Jane L. Jane
Name: KANDAL L HARDY
Title: CEO
STACCATO GOLD RESOURCES LTD.
Per: 6D 2
615

SCHEDULE "A"

Plan of Arrangement

ARTICLE ONE Interpretation

Definitions

- 1.01 In this Plan of Arrangement, unless something in the subject matter or context is inconsistent therewith:
- (a) "Agreement" means the Arrangement Agreement dated for reference March 22, 2010, made among Timberline and Staccato including the schedules thereto as the same may be supplemented or amended from time to time;
- (b) "Arrangement" means an arrangement under the provisions of Section 288 of the BCBCA, on the terms and conditions set forth in this Plan of Arrangement as supplemented, modified or amended;
- (c) "BCBCA" means the *Business Corporations Act*, S.B.C. 2002, c. 57, including all regulations made thereunder, as amended;
- (d) "Business Day" means any day on which commercial banks are generally open for business in Vancouver, British Columbia other than a Saturday, a Sunday or a day observed as a holiday in Vancouver, British Columbia under the Laws of the Province of British Columbia or the federal Laws of Canada;
- (e) "Closing Price" or "CP" means the closing price of a Staccato Share on the TSX Venture Exchange on the trading day immediately preceding the Effective Date;
- (f) "Court" means the Supreme Court of British Columbia;
- (g) "**Depositary**" means Computershare Trust Company of Canada;
- (h) "Dissent Rights" has the meaning set out in Article 4.01;
- (i) "Dissenting Shares" means the Staccato Shares held by a Dissenting Shareholder and in respect of which the Dissenting Shareholder has validly exercised Dissent Rights;
- (j) "Dissenting Shareholders" means Holders of Staccato Shares who have duly and validly exercised their Dissent Rights pursuant to Article Four and the Interim Order;
- (k) "Effective Date" such date, as Timberline and Staccato may agree upon, following the satisfaction or waiver of all of the conditions to the completion of the Arrangement as set out in Sections 5.1, 5.2 and 5.3 of the Agreement, or, in the absence of agreement, one Business Day following the satisfaction or waiver of such conditions, provided however the Effective Date shall be no later than September 30, 2010;
- (1) "Effective Time" means 12:01 a.m. (Vancouver time) on the Effective Date;
- (m) **"Exchange Ratio"** means 0.142857143;

- (n) "Exercised Staccato Option" means a Staccato Option that has been duly and validly exercised for Staccato Shares in accordance with its terms but in respect of which the name of the former holder of the Staccato Option has not been added to the register of Staccato Shares as the Holder thereof prior to the Effective Time;
- (o) "**Final Order**" means the final order of the Court approving the Arrangement as such order may be amended or varied at any time prior to the Effective Time or, if appealed then, unless such appeal is withdrawn or denied, as affirmed or as amended on appeal;
- (p) "Holder" means a registered holder of Staccato Shares or any person who surrenders to the Depositary certificates representing such Staccato Shares duly endorsed for transfer to such person in accordance with the Letter of Transmittal;
- (q) "Interim Order" means the order of the Court concerning the Arrangement providing for, among other things, the calling and the holding of the Staccato Meeting, as such order may be amended, varied or supplemented by the Court;
- (r) "Letter of Transmittal" means (i) in respect of Staccato Shares issued as a result of the provisions of Article 3.02(a) or Article 3.02(c), the letter of transmittal providing for the delivery to the Depositary of all documents, certificates and instruments as the Depositary or Timberline may reasonably require; (ii) in respect of all other Staccato Shares, the letter of transmittal forwarded by Staccato to the Holders of Staccato Shares providing for the delivery to the Depositary of certificates in respect of Staccato Shares and all other documents and instruments as the Depositary or Timberline may reasonably require; and (iii) in respect of Staccato Warrants, the letter of transmittal forwarded by Staccato to the Warrantholders providing for the delivery to the Depositary of certificates in respect of Staccato Warrants and all other documents and instruments as the Depositary or Timberline may reasonably require;
- (s) "Liens" means any hypothecs, liens, claims, encumbrances, charges, adverse interests or security interests;
- (t) "Notice of Dissent" means a notice of dissent duly and validly given by a Holder exercising Dissent Rights as contemplated in the Interim Order and as described in Article Four;
- (u) "Original Exercise Price" or "OEP" means, in relation to a Staccato Option, the exercise price per Staccato Share of that Staccato Option immediately prior to the Effective Time;
- (v) "Person" shall be broadly interpreted and includes any natural person, partnership, limited partnership, joint venture, syndicate, sole proprietorship, body corporate with or without share capital, unincorporated association, trust, trustee, executor, administrator or other legal personal representative;
- (w) "Plan of Arrangement" means this plan of arrangement and any amendment or variation thereto made in accordance with Article Six hereof;
- (x) "shareholder" or "holder of shares" means "shareholder" as defined in the BCBCA;
- (y) "**Specified Option**" means a Staccato Option (which is not an Exercised Staccato Option) in respect of which Timberline and the holder of that Staccato Option have agreed in writing, prior to the Effective Date, that Article 3.02(g) will apply;
- (z) "Staccato" means Staccato Gold Resources Ltd., a company incorporated under the laws of the Province of British Columbia;

- (aa) "Staccato Circular" means the management information circular of Staccato to be sent to the Staccato Securityholders in connection with the Staccato Meeting;
- (bb) "Staccato Meeting" means the special meeting of Staccato Securityholders to be held to consider and, if thought fit, to approve the Arrangement, among other things;
- (cc) "Staccato Options" means stock options exercisable to purchase Staccato Shares;
- (dd) "Staccato Option Share Number" or "SOSN" means, in respect of a Staccato Option, the number of Staccato Shares for which that Staccato Option was exercisable immediately prior to the Effective Time (assuming all such Staccato Options are vested immediately prior to the Effective Time);
- (ee) "Staccato Shares" or "Shares" means the common shares without par value in the capital of Staccato;
- (ff) "Staccato Securityholders" means holders of Staccato Shares, holders of Staccato Options and holders of Staccato Warrants;
- (gg) "Staccato Warrants" means warrants exercisable to purchase Staccato Shares;
- (hh) "Substituted Right" means a right that will entitle its holder to acquire, on the exercise thereof, the number of Staccato Shares that is equal to the product obtained when the Staccato Option Share Number of the Staccato Option for which it is exchanged pursuant to Article 3.02(b) is multiplied by the amount, if any, by which one (1) exceeds the quotient obtained when the Original Exercise Price of that Staccato Option is divided by the Closing Price, being, when expressed as a formula, SOSN x 1 (OEP/CP) and, such right shall be otherwise on, and subject to, the same rights and restriction as the Staccato Option for which it is so exchanged;
- (ii) "**Tax Act**" means the *Income Tax Act* R.S.C. 1985 (5th Supp.) c. 1, and the regulations promulgated thereunder, as now in effect and as it may be amended from time to time prior to the Effective Date; and
- (jj) "Timberline" means Timberline Resources Corporation, a company existing under the laws of Delaware;
- (kk) "**Timberline Options**" means options exercisable to purchase Timberline Shares to be issued in exchange for Specified Options;
- (II) "**Timberline Shares**" means the shares of common stock with par value of \$0.001 in the share structure of Timberline;
- (mm) "**Timberline Warrants**" means warrants exercisable to purchase Timberline Shares to be issued in exchange for Staccato Warrants; and
- (nn) "Warrantholder" means a registered holder of Staccato Warrants or any person who surrenders to the Depositary certificates representing such Staccato Warrants duly endorsed for transfer to such person in accordance with the Letter of Transmittal.

Headings

1.02 The division of this Plan of Arrangement into Articles, Sections and Subsections and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Plan of Arrangement. The terms "this Plan of Arrangement", "hereof" and "hereunder" and similar expressions refer to this Plan of Arrangement and not to any particular Article, Section or Subsection hereof and include any agreement or instrument supplemental therewith, and, unless reference is specifically made to some other document or instrument, references herein to Articles, Sections and Subsections are to Articles, Sections and Subsections of this Plan of Arrangement.

Number

1.03 In this Plan of Arrangement, unless something in the context is inconsistent therewith, words importing the singular number only shall include the plural and vice versa, words importing the masculine gender shall include the feminine and neuter genders and vice versa, words importing persons shall include individuals, partnerships, associations, trusts, unincorporated organizations and corporations and vice versa and words importing shareholders shall include members.

ARTICLE TWO Arrangement Agreement

Arrangement Agreement

2.01 This Plan of Arrangement is made pursuant and subject to the provisions of the Agreement.

ARTICLE THREE The Arrangement

Plan of Arrangement

- 3.01 This Arrangement will become effective at, and be binding at and after, the Effective Time on (i) Staccato, (ii) Timberline, and (iii) all registered and beneficial holders of Staccato Shares, Staccato Warrants and Staccato Options.
- 3.02 At the Effective Time, the following events or transactions shall occur sequentially in the order set out unless otherwise noted and shall be deemed to occur without any further act or formality required on the part of any Person, except as expressly provided herein:
 - (a) the holder of each Exercised Staccato Option outstanding immediately prior to the Effective Time will be deemed to be the Holder of, and the name of such holder added to the register of Staccato Shares in respect of, the number of Staccato Shares for which that Exercised Staccato Option was duly and validly exercised;
 - (b) each Staccato Option outstanding immediately prior to the Effective Time that is neither an Exercised Staccato Option nor a Specified Option, will be exchanged for a Substituted Right which shall be exercisable immediately;
 - (c) each Substituted Right shall be deemed to have been exercised by its holder and, subject to Article 3.07, such holder shall be deemed to be the Holder of, and the name of such holder shall be deemed to be added to the register of Staccato Shares in respect of, such number of Staccato Shares (if any) for which such Substituted Right is exercisable;

- (d) the Dissenting Shares shall be deemed to have been transferred to Staccato and cancelled and the Dissenting Shareholders shall cease to have any rights as shareholders of Staccato other than the right to be paid the fair value of their Staccato Shares in accordance with Article Four;
- (e) all Staccato Shares (other than the Dissenting Shares transferred to Staccato in accordance with Article 3.02(d)) will be transferred to Timberline in consideration for 0.142857143 of a Timberline Share and US\$0.0001 for each Staccato Share;
- each Staccato Warrant outstanding immediately prior to the Effective Time will be (f) exchanged for a Timberline Warrant to purchase that number of Timberline Shares as is equal to the number of Staccato Shares that would otherwise have been issuable on exercise of the Staccato Warrant immediately prior to the Effective Time multiplied by the Exchange Ratio, at an exercise price per Timberline Share equal to the exercise price per Staccato Share of such Staccato Warrant immediately prior to the Effective Time divided by the Exchange Ratio, provided that: (A) if the foregoing calculation results in a Timberline Warrant being exercisable for a number of Timberline Shares that includes a fraction of a Timberline Share, then the number of Timberline Shares subject to such Timberline Warrant shall be rounded up to the nearest whole Timberline Share in the case of a fraction representing 0.5 of a Timberline Share and rounded down to the nearest whole Timberline Share in the case of a fraction representing 0.5 or less of a Timberline Share; (B) if the aggregate exercise price payable to acquire the Timberline Shares under the Timberline Warrant includes a fraction of a cent, then such aggregate exercise price shall be rounded up to the nearest cent in the case of a fraction representing more than 0.5 of a cent and rounded down to the neared cent in the case of a fraction representing 0.5 of a cent or less; (C) outside of restrictions on issue and resale under United States federal and state securities laws, all other terms and conditions of such Timberline Warrant, including as to expiry, will be the same as the terms of the Staccato Warrant exchanged therefor; and (D) for greater certainty, this Plan of Arrangement shall not result in any other adjustment to the number or type of shares for which, or the exercise price of, such Staccato Warrants:
- (g) each Specified Option outstanding immediately prior to the Effective Time will be exchanged for a Timberline Option to purchase that number of Timberline Shares as is equal to the Staccato Option Share Number of that Staccato Option multiplied by the Exchange Ratio, at an exercise price per Timberline Share equal to the Original Exercise Price of such Staccato Option divided by the Exchange Ratio, provided that: (A) if the foregoing calculation results in a Timberline Option being exercisable for a number of Timberline Shares that includes a fraction of a Timberline Share, then the number of Timberline Shares subject to such Timberline Option shall be rounded down to the nearest whole Timberline Share; (B) if the aggregate exercise price payable to acquire the Timberline Shares under the Timberline Option includes a fraction of a cent, then such aggregate exercise price shall be rounded up to the nearest cent; (C) outside of restrictions on issue and resale under United States federal and state securities laws, all other terms and conditions of such Timberline Option, including as to vesting and expiry, will be the same as the terms of the Staccato Option exchanged therefor; and (D) section 4.7 of the 2003 Staccato Option Plan shall not apply in respect of the change of control of Staccato effected by this Plan of Arrangement;

provided that none of the foregoing will occur or be deemed to occur unless all of the foregoing occurs.

- 3.03 Upon the exchange, at the Effective Time, of Staccato Options for Substituted Rights pursuant to Article 3.02(b) and as a result of such exchange:
 - (a) the former holders of such Staccato Option shall cease to be the holders of the Staccato Options so exchanged and the name of each such former holder of Staccato Options shall be removed from the register of holders of Staccato Options;
 - (b) no such former holder of Staccato Options shall be entitled to a certificate or other document representing such Substituted Right; and
 - (c) each such Staccato Option shall be deemed to be cancelled.
- 3.04 Upon the exchange, at the Effective Time, of Substituted Rights for Staccato Shares pursuant to Article 3.02(c) and as a result of such Exchange:
 - (a) the former holders of the Substituted Rights shall cease to be the holders of the Substituted Rights so exchanged;
 - (b) no such former holder of Substituted Rights shall be entitled to a certificate or other document representing such Staccato Shares; and
 - (c) each such Substituted Right shall be deemed to be cancelled.
- 3.05 Upon the exchange, at the Effective Time, of Staccato Shares for Timberline Shares pursuant to Article 3.02(e) and as a result of such exchange:
 - (a) each former Holder of Staccato Shares (other than Dissenting Shareholders with respect to Dissenting Shares) shall cease to be the holder of the Staccato Shares so exchanged and the name of each such former Holder shall be removed from the register of holders of Staccato Shares;
 - (b) each such former Holder of Staccato Shares (other than Dissenting Shareholders with respect to Dissenting Shares) shall become the holder of the Timberline Shares exchanged for the Staccato Shares held by such former Holder and shall be added to the register of holders of Timberline Shares in respect thereof; and
 - (c) Timberline shall become the holder of the Staccato Shares so exchanged and shall be added to the register of holders of Staccato Shares in respect thereof.
- 3.06 Upon the exchange, at the Effective Time, of Staccato Warrants for Timberline Warrants pursuant to Article 3.02(f) and as a result of such exchange:
 - (a) each former Warrantholder shall cease to be the holder of the Staccato Warrants so exchanged and the name of each such former Warrantholder shall be removed from the register of holders of Staccato Warrants; and
 - (b) each such former Warrantholder shall become the holder of the Timberline Warrants exchanged for the Staccato Warrants held by such former Warrantholder and shall be added to the register of holders of Timberline Warrants in respect thereof.
- 3.07 For greater certainty, if the quotient obtained when the Original Exercise Price of a Staccato Option exchanged for a Substituted Right pursuant to Article 3.02(b) is divided by the Closing Price is equal to or greater than one (1) the number of Staccato Shares for which a Substituted Right is exercisable

shall be nil and the holder thereof shall not be entitled, under Article 3.02(c) or otherwise, to a Staccato Share in exchange therefor, and such Substituted Right shall be deemed to be surrendered and cancelled for no consideration.

3.08 Any transfer of any securities pursuant to the Arrangement shall be free and clear of any Liens.

ARTICLE FOUR Rights of Dissent

Rights of Dissent

- 4.01 A Holder of Staccato Shares may exercise dissent rights ("**Dissent Rights**") conferred by the Interim Order in connection with the Arrangement in the manner set out in Section 242 of the BCBCA, as modified by the Interim Order, provided that the Notice of Dissent is received by Staccato by no later than 4:30 p.m. (Vancouver time) on the date which is two Business Days prior to the date of the Staccato Meeting. Without limiting the generality of the foregoing, Holders who duly exercise such Dissent Rights and who are ultimately determined to be entitled to be paid fair value for their Staccato Shares shall be deemed to have transferred such Staccato Shares (free and clear of all Liens), as of the Effective Time, without any further act or formality, to Staccato in consideration of their entitlement to be paid the fair value of the Staccato Shares by Staccato under the Dissent Rights. In no case shall Staccato or Timberline be required to recognize such Holders as shareholders of Staccato at and after the Effective Time, and the names of such Holders shall be removed from Staccato 's central securities register as of the Effective Time.
- 4.02 In the event a Holder gives a Notice of Dissent but is not ultimately entitled, for any reason, to be paid the fair value of the Staccato Shares in respect of which the Notice of Dissent was given as contemplated in Section 242 of the BCBCA and the Interim Order, such Holder shall be deemed to have participated in the Arrangement on the same basis as non-dissenting shareholders of Staccato and shall be entitled to receive only the Timberline Shares that such non-dissenting Shareholders are entitled to receive, on the basis set out in Article 3.02(e).
- 4.03 For greater certainty, no holder of Staccato Options or Staccato Warrants shall be entitled to exercise Dissent Rights with respect to Staccato Options or Staccato Warrants.

ARTICLE FIVE Certificates and Documentation

Effect of Arrangement

5.01 After the Effective Time, certificates formerly representing Staccato Shares and Staccato Warrants prior to the Effective Time shall cease to represent such shares and warrants and shall represent only the right to receive the consideration therefor specified in Article Three in accordance with this Plan of Arrangement, subject to Article 5.06 and to compliance with the requirements set forth in this Article Five; and for greater certainty, no former Holder will be entitled to receive any interest, dividends, premiums, or other payment in connection therewith.

Entitlement of holders of Specified Options to Timberline Option Certificates

5.02 At the Effective Time, each and every certificate, document, agreement or other instrument, if any, formerly representing Staccato Options shall be and shall be deemed to be cancelled, void and of no further force and effect without any further authorization, act or formality. Each former holder of a Specified Option shall be entitled to receive instruments representing the Timberline Option to which

each such former holder of a Specified Option is entitled pursuant to the provisions of this Plan of Arrangement as soon as practical after the Effective Time. Timberline shall register each former holder of a Specified Option as the holder of the Timberline Options to which such former holder of a Specified Option is entitled pursuant to the provisions of this Plan of Arrangement and make available to each former holder of Specified Options, or send by first class mail to the last address of each such former holder of a Specified Options as shown on the books and records of Staccato, instruments representing such Timberline Option.

Letter of Transmittal

5.03 Staccato shall cause a Letter of Transmittal to be sent to each holder of Staccato Shares or Staccato Warrants, at the address of such holder as it appears in the register maintained by or on behalf of Staccato in respect of such holders, or if there is no such register, then as it appears in other records maintained by or on behalf of Staccato, with the Staccato Circular.

Right of Holder to Receive Timberline Shares and Cash

5.04

- (a) At or prior to the Effective Time, Timberline shall deposit with the Depositary, for the benefit of the Holders of Staccato Shares (including each Holder whose name is deemed to be entered on the register of Staccato Shares pursuant to Article 3.02(a) or Article 3.02(c)) and Exercised Staccato Options, a certificate or certificates representing the aggregate number of Timberline Shares which the Holders of Staccato Shares are entitled to receive hereunder (calculated without reference to whether any Holders have exercised or may exercise Dissent Rights) and cash equal to US\$0.0001 times the aggregate number of Timberline Shares to which such Holders are entitled pursuant to Article 3.02(e).
- (b) At the Effective Date, Timberline will cause the Depository to forward or cause to be forwarded by first class mail (postage prepaid) to the former Holders whose names are deemed to be entered on the register of Staccato Shares pursuant to Article 3.02(a) or Article 3.02(c), at the address on record at Staccato for such former Holders, certificates representing the number of Timberline Shares, if any, issuable to such former Holder as determined in accordance with the provisions hereof and a cheque payable to the order of such former Holder in an amount equal to US\$0.0001 times the aggregate number of Timberline Shares to which such former Holder is entitled (if any) pursuant to Article 3.02(e), subject to any withholding obligation under applicable tax laws.
- (c) Subject to Article 5.04(b) and Article 5.06, as soon as practicable following the later of the Effective Date and the date of deposit by the Holder with the Depositary of a duly completed Letter of Transmittal, documents, certificates and instruments contemplated by the Letter of Transmittal, the certificates representing Staccato Shares (if any) and such additional documents and instruments as the Depositary may reasonably require, Timberline shall cause the Depositary to:
 - (i) forward or cause to be forwarded by first class mail (postage prepaid) to the former Holder at the address specified in the Letter of Transmittal; or
 - (ii) if requested by the former Holder in the Letter of Transmittal, make available at the Depositary for pick-up by the Holder; or

(iii) if the Letter of Transmittal neither specifies an address nor contains a request as described in (ii), forward or cause to be forwarded by first class mail (postage prepaid) to the former Holder at the address of such former Holder as shown on the share register maintained by Staccato immediately prior to the Effective Time,

certificates representing the number of Timberline Shares, if any, issuable to such former Holder of Staccato Shares as determined in accordance with the provisions hereof and a cheque payable to the order of such former Holder in an amount equal to US\$0.0001 times the aggregate number of Timberline Shares to which such former Holder is entitled (if any) pursuant to Article 3.02(e), subject to any withholding obligation under applicable tax laws.

Subject to Article 5.04(b), until such time as a former Holder of Staccato Shares deposits with the Depositary a duly completed Letter of Transmittal, the documents, certificates and instruments contemplated by the Letter of Transmittal, the certificates representing Staccato Shares (if any) and such additional documents and instruments as the Depositary may reasonably require, all certificates representing Timberline Shares and cash to which such former Holder is entitled in respect of such Staccato Shares, if any, shall, subject to Article 5.06, to be held in trust for such former Holder for delivery to the former Holder, upon delivery of the Letter of Transmittal, the documents, certificates and instruments contemplated by the Letter of Transmittal, the certificates representing the Staccato Shares (if any) and such additional documents and instruments as the Depositary may reasonably require.

Right of holder of Staccato Warrants to Receive Timberline Warrants

5.05

- (a) At or prior to the Effective Time, Timberline shall deposit with the Depositary, for the benefit of Warrantholders, a certificates representing the Timberline Warrants which the Warrantholders are entitled to receive hereunder.
- (b) Subject to Article 5.06, as soon as practicable following the later of the Effective Date and the date of deposit by the Warrantholder with the Depositary of a duly completed Letter of Transmittal, documents, certificates and instruments contemplated by the Letter of Transmittal, the certificates representing Staccato Warrants (if any) and such additional documents and instruments as the Depositary may reasonably require, Timberline shall cause the Depositary to:
 - (i) forward or cause to be forwarded by first class mail (postage prepaid) to the former Warrantholder at the address specified in the Letter of Transmittal; or
 - (ii) if requested by the former registered holder of Staccato Warrants in the Letter of Transmittal, make available at the Depositary for pick-up by the former Warrantholder; or
 - (iii) if the Letter of Transmittal neither specifies an address nor contains a request as described in (ii), forward or cause to be forwarded by first class mail (postage prepaid) to the former Warrantholder at the address of such former Warrantholder as shown on the share register maintained by Staccato immediately prior to the Effective Time;

certificates representing the number of Timberline Warrants, if any, issuable to such former Warrantholder as determined in accordance with the provisions hereof, subject to any withholding obligation under applicable tax laws.

Until such time as a former Warrantholder deposits with the Depositary a duly completed Letter of Transmittal, the documents, certificates and instruments contemplated by the Letter of Transmittal, the certificates representing Staccato Warrants (if any) and such additional documents and instruments as the Depositary may reasonably require, all certificates representing Timberline Warrants to which such former Warrantholder is entitled in respect of such Staccato Warrants, if any, shall, subject to Article 5.06, to be held in trust for such former Warrantholder for delivery to the former Warrantholder, upon delivery of the Letter of Transmittal, the documents, certificates and instruments contemplated by the Letter of Transmittal, the certificates representing the Staccato Warrants (if any) and such additional documents and instruments as the Depositary may reasonably require.

Surrender of Rights

5.06 Notwithstanding any other provision hereof, any certificate which, immediately prior to the Effective Time, represented Staccato Shares or Staccato Warrants that has not been surrendered with all other documents, certificates and instruments required by Article 5.04(c) or Article 5.05(b), as the case may be, on or before the sixth anniversary of the Effective Date shall cease to represent any right or claim of any kind or nature and the right of the former Holder to receive Timberline Shares and cash and the Timberline Shares issued and cash payable to such former Holder and the right of the former Warrantholder and the Timberline Warrants issuable to such former Warrantholder shall, on the sixth anniversary of the Effective Date, be deemed to be surrendered to Timberline, together with all dividends or distributions thereon held for such Holder or Warrantholder, for no consideration.

Distribution

5.07 All dividends paid or distributions made in respect of the Timberline Shares for which a certificate formerly representing Staccato Shares has not been deposited with all other documents, certificates and instruments as provided in Article 5.04 hereof, shall be paid and delivered to the Depositary to be held (subject to Article 5.06) in trust for the former Holder of such unsurrendered certificate, for delivery to such former Holder upon delivery of the certificate and other documents, certificates and instruments required by Article 5.04(c).

Fractional Shares

5.08 No fractional Timberline Shares shall be issued and any Holder of Staccato Shares in connection with the Arrangements. If a Holder of Staccato Shares would otherwise be entitled to receive a fractional Timberline Share, the number of Timberline Shares issuable to such Holder under the Arrangement will be rounded up to the nearest whole Timberline Share in the event that such Holder is entitled to a fractional share representing 0.5 or more of a Timberline Share and rounded down to the nearest whole Timberline Share in the event that such Holder is entitled to a fractional share representing 0.5 or less of a Timberline Share.

Withholding Rights

5.09 Notwithstanding any other provision of this Plan of Arrangement, Timberline, Staccato and the Depositary shall be entitled to deduct and withhold from the consideration otherwise payable to any former Holder under this Arrangement such amounts as Timberline, Staccato or the Depositary is required or entitled to deduct and withhold with respect to such payment under the Tax Act, the United States Internal Revenue Code of 1986 or any provision of any applicable federal, provincial, state, local or foreign tax law, in each case, as amended. To the extent that amounts are so withheld, such withheld

amounts shall be treated for all purposes hereof as having been paid to the former Holder in respect of which such deduction and withholding was made, provided that such withheld amounts are actually remitted to the appropriate taxing authority.

Lost Certificates

5.10 In the event any certificate, which immediately prior to the Effective Date represented one or more outstanding Staccato Shares that were transferred pursuant to Article 3.02(e) or Staccato Warrants that were surrendered pursuant to Article 3.02(f) has been lost, stolen, or destroyed, upon making of an affidavit of that fact by the Person claiming such certificate to be lost, stolen or destroyed and delivery of such other documents as may be required by the Depositary or Timberline, including a bond satisfactory to Timberline and Staccato and their respective transfer agents and the Depository in such sum as Timberline may direct, or otherwise providing an indemnity in favour of Timberline, Staccato and the Depositary in a manner satisfactory to Timberline and Staccato against any claim that may be made against Timberline or Staccato or the Depositary with respect to the certificates alleged to have been lost, stolen or destroyed, Timberline will issue will (or will cause to be issued) in exchange for such lost, stolen or destroyed certificates, appropriate the consideration to which the holder is entitled pursuant to the Arrangement.

ARTICLE SIX Amendment

Plan of Arrangement Amendment

- 6.01 Staccato and Timberline reserve the right to amend, modify and/or supplement this Plan of Arrangement at any time and from time to time, provided that any amendment, modification or supplement must be contained in a written document which is filed with the Court and, if made following the Staccato Meeting, approved by the Court and communicated to Staccato Securityholders in the manner required by the Court (if so required).
- 6.02 Any amendment, modification or supplement to this Plan of Arrangement may be proposed by Staccato and Timberline at any time prior to or at the Staccato Meeting with or without any other prior notice or communication and, if so proposed and accepted by the persons voting at the Staccato Meeting, shall become part of this Plan of Arrangement for all purposes.
- 6.03 Any amendment, modification or supplement to this Plan of Arrangement which is approved or directed by the Court following the Staccato Meeting shall be effective only if it is consented to by Staccato and Timberline (acting reasonably) and, if required by the Court, approved by Staccato Securityholders voting in the manner directed by the Court.
- 6.04 This Plan of Arrangement may be withdrawn prior to the Effective Time in accordance with the terms of the Agreement.

Schedule "B"

Outstanding Staccato Options, Staccato Warrants and Staccato Securities

Shareholders of Staccato Gold Resources Inc. as at March 19, 2010

MOLUCIA				COLCIACO		
DENTERER	FULLNAME	DOODSDA	MICHAGO	TOTAL REG ADDRESS LINE 1	REG ADORESS LINE 2	REG ADDRESS LINE 3
C00000000094	EDWARD DIONES & CO	63141-5824	USA	40000	700 MARYVILLE CENTER DR	STIGUIS MO 63141-5894
0,0000	RICK LEWIT	600 NVA	CAN M	4000 RICK LEVAT	#306-BUILDING 102	15910 FRASH R HETHWAY
000000000000000000000000000000000000000	PERSHING LLC	730	7303 UNA	30000 PERSHING LLC	PO BOX 2050	JERSEY CITY NJ 07303
	ROYTOR & COTR					
CC0000000728	A/C T12058611	MSV 3U3	CAN.	CO & HOTYCH DRODGE	TR A/C T12058511	RBC DEXIA INVESTOR SERVICES
00000001597	MOHAMED'S NIZAR	MOC 1AT	Ž	20000 MOHAMED S NIZAR	1415 MILITARY THAIL	SCARBOROUGH DN M1C 1A7
00000001945	DOMALD J DECKER	.0868	YSN TOBES	666667 DONALD J DECKER	3716 IDAHO ST	STEA
00000002381	DONALO J DECKER	93/J TDK98	200	666667 DONALD J DECKER	3716 E IDAHO ST SUIT - A	ELKO NV SOBOL
00000002429	CYNTHIA DOUTHARD	PSU TZMOR	5	67500 CYNTHA DOUTHARD	SO WOLF ROAD	BAILEY CO ROAZI
C0000002437	MANK INNERFELD	80232 UKA	2 Usya	55500 HANKINNERFELD	10691-0 WEST FLORIDA AVE	DAKEWOOD CO 80232
00000003255	SCOTT CHOSED	10010 USA	150	20000 SCOTT CHOSED	30 WATERSIDE PLAZA #20D	NEW YORK NY 10010
	DEPORTUNITIES			FORUM GLOBAL		
016E0000000	MASTER FUND LTD		ACB.	571429 OPPORTUNITIES MASTER	FUND LTD	C/O FOC COMPORATE SERVICES
00000003875	SEPANATE PROPERTY CHRIS HUMN &	78802-1647	Ē	20000 SEPARATE	PROPERTY	PO BOX 1647
C0000003913	GREG OLSON &	T2E 8K9	Ç.	30000 CHRISTILLIAN &	TOBIE HUHIN TEN COM	311 16TH AVE NE
CI000003921	VALERIE HIOKSONUT TEN HARE & CO FBO GEOLOGIC	82509-3217	VSV	16000 GREG OLSON & HARE & CD FBO GEOLDGIC	VALERIE HOLSON JT TEN	1100 DONEGAL
C00000004111	RESOURCES FUND LP	VSN 9820T	5 USA	742857 RESCURCES	FUND &	C/O THE BANK OF NEW YORK
CDQDQQQ4138	WILLIAM O STONEBURNER	76513-5792	ASU	SCODO WILLIAM D STONEBURNER	P 0 H0X 1792	BELTON TX 76513-5792 THE EXCHANGE TOWER
C0000004154	AUGEN CAPITAL CORP	MSX 1A6	CAN	8600DD AUGEN CAPITAL DORP	130 KING STREET WEST	SUITE 720
00000004162	AUMERCO LTD	SWT XSW	CM	S949000 AUMERCO LTO	C/O J DAVID MASON	130 KING ST WEST SUITE 720
5TC0C0000018	CDS & CO (NCI)	MSW 1GS	CAN	87634173 CDS & CD [NCI)	PD BOX 1D38 STN A	28 THE ESPLANADE

REG_ADDRESS_UNE, 6 REG_ADDRESS_UNE_S REG ADDRESS LINE 4 SURREY BC V4N 0X9 SECURTIES DAGE UNII ED STATES UNITED STATES

LORGINTO ON MISY BLS 155 WELLINGTON ST W 2ND FLOOR

UNITED STATES ELKO NV 89801

UNITED STATES UNITED STATES UNITED STATES UNIT ED STATES

BRITISH VIRGIN ISLANDS 125 MAIN STROBOX 1/4 ROAD TOWN TORTOLA

UNITED STATES UVALDETX 78802-1647

CALGARY AN TZE 3K9 PO BOX 52034

UNITED STATES NEW YORK NY 10286 UNITED STATES CASPER VYY 82609-3217 PO 80X 11203

TORONTO DN MSK 1A6 P G & GX 137

UNITED STATES

TORONTO ON MSX 146 TORONTO ON MSW 195

Stack options - continuity schedule Standaro Gold Resources Ltd.

													-	-	ı		
1,400,000									100,000				600,000	300.000		144041	1400-00
100,000									357					1000	\$0.600		100400
1,200,000								50.000	60,000	20,000		40,000	485,000	485.CC0	50600	18-40-11	19-401-06
1,400,000						0.000.00					50,000	50,000	895.000	575 000	B09.08	1-Fab-11	1+35-36
800,000							15,000						400,000	362,000	\$0,400	5-Jan-11 90,400	Flands
COUTCE												50,500			\$0.150	1-020-10	1-120-06
20,000							20,000				300000				\$0.100	18-01-0-10	10-Aug-06
180,000			-	-							180,000				\$0.100	NA WAYN	25-Nov-04
Elenu Musikhage Edgi	Entherne Ho	AND BA	Estualente	Dhamas Dhamas	Rive sees	ACCUMANTAL SANGE	Skotav Skotav	NACARI VARA	COO CO VIII	Charles .	Solar Solar	Capacia	Hydrachus	CESSOR	St. Pito	Engin	Giant date

2 37	Expired May 80 Expired	_	10.Dec-10	2-401-18	33-May-08	BO-May-OR	72-May-06 12-May-13	27-May-08	28-00-07	18-00-07		1-400-00	10240	19-4gr-05	14.00-08	Clarco.	1-1260-06	St-Duk-01	The section
	Con Am	de la banke	PERMIT!	2.4pr-14	String 13	SE-MAR-193	12-MMp-13	97-487-13	28-00-12	18-00-12	8-Cd-12	14ddv11		19-40-11	1-F0b-11	5-Jan-1)	1-000-10	18-Aug-10	TALES 40. 100
		Consultant of the last	Mary N	\$5.100	# 100				DOM US	\$0.800	080 08	\$0 600	\$0.600	\$0,600	B08.00	\$0,400	\$0.150	\$0.100	40.000
240,000	(8,490,000)	-		240,000				S		10000	760,000	300,000	5000	485,000	575 000	362,000			
240,000	12,810,0001			240,000							750,000	600,000		485,000	695,000	400,000			
815,000											E75,303			40,000	50,000		80,800		
815.000 100,000	(100,000)			100,000							.75 000				50,000			300000	0,00,00
000 000	163,000)		7	100,000				200,000						20,000					
375,000 60,000		04,000					-		-		176,000	100,000	2018	50,000					
60,000														50.000					
136,000		ANTHE	-							51000	60,000					19,000		20,000	
90,000															0.000.00				
00000 B0000										250,000									
400,000	p.acc.occ			\$60,00¢				100	COCCOURT	į									
900,000 IG0,000		00000	2000						1,000,000 100,000,1										
600,000		781,187					600,000												
000,000		50,000	200			60,000					1	ď							
DOD NO		CCO/OC																	
200000				A Specific	200,000								-						
3,655,600	9630000 (000008136 (000008136	450,000	400000000000000000000000000000000000000	100000	KURK	90,000	400,000	300,000	120000	250,000	2,075,000	1,400,000	100,000	1,200,000	1,400,000	800,000	900,000	20,000	1900,081

Altin Morris expired April 2008. Scott Getrom expend July 2008, Bruss Yearman Aspered Sept 2007, Grandish original Sept

Predict options produce that accessive on New 15-19.

8.0 m % ot aptions ram. 89,842,753 9,884,279 6,989,279

Oliter Conferent Property in TURY CAY Supported

excise - Persec 1,004,000 exercises (60.10 to \$0.10 pints price)

Santad - Green: 1,640,000 paraelled (\$6.25 - 40.50)

Sussion - Blue 96,000 options - κεγ transfer to TLN, Imperoptions pay not be to the moving, and helders are Stateday director (George Brash Α, Lasted of Louiside (properly (64 % Huselmon, could add to dissection 0;17), and NY amployees that TLN may retain (Gery Edmondo, Cyriline Nagley).

REGISTER OF WARRANTS

NAMB	CERTAUMBEK	DATEGRANT	DATISSIUM	HOPWARRANTS - EXERCISE		RICE 'EXPORYDATE	AMENDEDEXPIRY	HALANCE	NOTES
ALTEIA CAPITAL ANSTRUT PRADAPANT 7 FURSTENTUMS 9190 VADUZ, LUICHTENSTEIN	JULX07-411	JULY 4407	JULY 4007	1,071,429	050	JULY ANN	JULY 4717	L091.429	
W.D. LATIMER & CO. (THAUGEN CAPITAL CORP. ACCOUNT 3/047A-Z 100 WELLINGTON ST. WEST SUITE 600, P.O. BOX 96, TD CENTRE TORONTO, UNITARIO, MSK 1G8	TD-CIMCICIE	JULY 4007	JCLX 4007	8wa,000	άτρ	MAY ATOR	THE ATTHE	₩a,pco	
DOUGLAS R. CASEY 202 CLEVEDON-XAWAKAWA ROAD CLEVEDON, NEW ZEALAND	אנדאת ווג	THE ATH	11T.Y 47T?	270,000	De.O	JRILY 4KW	TICK VALLE	270,000	
CRESCENT INTERMATIONAL LTD. CYD CANTARA (SWITZERLAND) SA 84 AV. JOUIS-CASAL CH 1216 COINTEIMOENEVA SWITZERLAND	JULY25-04	אונץ אמד	ЛП У 4-ЮТ	900,000	05-0	2004 ATION	נוו+ גיות	900009	
DKR SOUNDSTORE OASIS HOLDWA RUND LTD, CLD CODAN TRUST COMPANY (CAYMAN) LTD, CRECKET SQUARE, DUTCHINS DRIVE GBORGE TOWN, TRAND CAYMAN KYJ-LLII, CAYMAN ISLANDS	אסייטאר	JULY 9/07	ארווי איזטוב	714286	Q di	1CTX 409	אחדה ייור	714,286	
PETTAL KULEY 3503 - 1438 RICH ARDS STREET YAMOUDVER, R.C., V6Z 388	90-03XTHT	JULY 4KD	JULY 4/07	70,000	0.50	TULY 4/09	TICA ATHLE	70,000	
FORUM GLOBAL OPPORTONITES NASYER FUND, LTD 125 MAIN STREET, TORTOLA, BY1	JULY67-07	JULY 1/80	ли у 4807	571,429	P.S.	ли. у 409	JULY 4/1)	571.729	

NAME	CRETNUMBER	DATEGRANT	DATEISSUED	#OFWARRANTS	EXERCISEPROCE	EXPLANDATE	AMENDODEXMIN	BALANCE.	NOTES
TOWN TOWNS THE WAR THE	*** COATIN	NUT ATM	100 X 1011	441,787	0.58	HALV 4/09	JULY A/11	۵	REALEGISTERED TO
RTRSHINW & CO. (145RRRL LYNCH INTH 1, CHTY BANK, 18 TH BLOCK, 124 YRONT ST. W.EST. BLOCK, 124 YRONT ST. W.EST.	## TOTAL	AULY 476	HUN YOU	2,857,143	**	140 KW	HILY 4711	c	RE RECUSTERED TO
JAMES MALCOLM 1116H JAMESON AVENUE N.E. CALCARY, ALBERTA, 125 OLS	JUEN07-10	JULY 4/00	JULY 4/07	UCOLISE.	0.50	JULY 4/09	JULY 4/33	257,000	
BRANT INVESTMENTS LIMITED AIC 109423109 SLLEVEL, ROYAL BANK PLAZA SOUTH TOWER, 200 BAY STREET TORONTO, DHYARM, MELLEL	אועי אוניין ר	Mr Ali	JELY 407	450,000	0.50	JULY 4014	JUEY 47??	+10,000	70
BMO-NESBITT BURNS TE NINERALERI (XVENERO YEISLDS, NUETUSER ES FIRM DO, ACCHUNT ROS-2000-17 CAO C. CLANK, I PIRST CANADIAN FLACE BMO NESBITT BURNS, 47 TH PLODE PLODE	JULY97 12	ןרד∧ ז והט	MAY ATTIC	714.285	0.50	10TX 7400	ሀቱ አጎስና	7 [A ₁ 285	
MONARCH CARITAL FUND LTD. 2° FLOOR, HARBOHK HOUSE WATERFRONT DRIVE, KOAD TOWN TORNOLA, BY]	VIEVO 13	JULY 4007	JULY 4/00	\$42,563	0.50	JULY 400	11.0r.ATDr	542363	
GUMDYCTI ON PINETREE RESOUNCE PARTNERSHIP (ACCIA MINJUUS-25) IOI BAY STREET, JO ^{TA} FLOOR TORONTO, ONTARIO, MSI 238	3UEN07-34	JULY 4kp	JULY 4/07	1,546,000	050	JULY 409	JULY 411	1,700,000	
HAYWOOD SECURITIES INC. ITF PLATORD RIVESTMENTS LTD. 2000 - AND RURRARD STREET VANCHUVER, D.C., V6C 3A6	31.7807-15	TULY 4007	JULY 4:07	130,000	0.50	JULY 409	יחרג יוור :	150,000	

ROYTOR & CO. A/C PIZEJTAJJ, ROYAL BANK PLAZA, 200 BAY STREET SUITE 2500, NORTH TOWER TYRIGHTE 2500, NORTH TOWER TORONTO, INTARIO, MSJ 215 ROYAL BANK PLAZA, 200 BAY STREET SUITE 2500, NORTH TOWER TORONTO, ONTARIO, MSJ 215 WILLIAM M. SHERIFF 64 SHORELINE CHICLE PORT MODDY, B.C., VHTHB'S WILLIAM M. SHERIFF 64 SHORELINE CHICLE PORT MODDY, B.C., VHTHB'S WILLIAM M. SHERIFF 64 SHOREMARES LITD., P.O., BOX 43764 35 MODREMATE, LONDON BOHTHA WILLIAMSON BOUCHERVILLE, QUENUCL THE TZS THE TZS THE TZS	JULY07-16 JULY07-19 JULY07-20	JULY 407 JULY 407 JULY 407 JULY 407	TOTA 400. TOTA 400. TOTA 400. TOTA 400.	#OFWARRANTS 1,070,000 620,000 14,908,600	EXERC.	0.50 0.50 0.50	150 JULY 409 150 JULY 409 150 JULY 409 150 JULY 409		JULY 409 JULY 409 JULY 409
WONTO: CIVITARIO, MSJ 235 OVAL BANK PLAZA, 200 BAY FREET THE MOD, NORTH TOWER BROKTO, ONTARIO, MSJ 235	TETAON-15	JULY AND	JIGLY 4017	620.000	0.50		JULY 400y		JIEX 4311
WILLOW M. SHERIFF SG SHORELINE CHALLE FORT MODDY, B.C., VIH 183	NEXECTIVE	JULY 407	JULY 4/07	112/213	11.50		30FX 409	- !	JULY 4/II
BCGD-SAA 12 WOODCVLE TOVDON WOODSCALE TOVDON WOODSCALE TOVDON	#EX07-10	#III v AKE	1611-1407	++1000-1000	تبدي		MUP X THE	35	## ***********************************
ELENA TOJERNYCHEVA 1183 DE BLOIS BOUCHERVELLE, QU'ERRC. TAB 723	JULY07-20	JULY 407	200-X10r	90006:	Q.S.D		JCT:Y 409		ייםרג אַננ
PENSON FINANCIAL SERVICES OF CANADA BIC. ITF HARBORY IEW MASTER HUND LP. 369 ST. MACQUES OUBSE, SUITE 1000, MONTREAL QUEBEL; HZY 1695	JULYG7:21	JULY 4:07	ALKIINT 2407	441.781	0.50	ļ	JULY 409	TITA ATZU 600 K-TITU	
RONTOR & COSOLEMENT A WHILLIAM SOM NONDERS LTD., REC DRIVE DRIVES OF SERVICES TRUST SHILLIAM, SOLINI TOWER MOYAL RANK PLAZA, SOLINI TOWER MINAY STREET TORIGHTA, CMTARIO MS1316	ML VIOLES	100th Adlita	AMEMAY	007806'	620	<u></u>	IIII AASA IIII	TUEN AMO JUEN AMA	

NAME	CERTNUMBER	DATEGRANT	DATIONNUM	STEWARKARIN.	SOMEOISIONNOS	EXPRYPATE	AMENDEDEKPIRY BALANCE	BALANCE
ROYTOR & CO POR SMITH & WILLIAMSON NOMENES LID. REIDENIA INVESTOR SERVICEN TRUST SL LEVIL. ROYAL BANK PLAZA NOTHI TOWER 200 BAY STREET TORONTO, ONTARIO MSI 215	אביבייטאוטנ	ייטיא אינטיר	OCTOBER OCTOBER	992.500	0.50	ANT ATIN	נוף אדולור	992.900
ROWNON & CO POIT EMETT & TOWNS HAND FOR THE HERE ROWN FLATANA ROWN STEEDER ROWN STE	महर्ग व्यक्ताता	THE SAME	9889 98900120	+154,000	P.	TUDY 4899	The ATCH	٥
CONTINE WITH WHEN WHITH PARTY STREET, TORONIO, CONTINE WITH WAS AND WHITH WAS AND WAS	STEE HONTEIN	mr.v.agi	OCTOBER 688	0007118	100	MTX-4009	ttit x III u	0
MST 11:1 SLVLION LOMONIA: CMLIVER CONNECT COUNT HARITY VALUE F. L.D. E.O. BOX 611	(1.08/1/1	the Albi	POSTOR WENCHOOL	8,857,143	86	WO APPL	Hrr-Khit	
CHEVROT CAPITAL (NOMINIES) LÜMITED 90 LONG ACRE LONDON, WCZE 9RA	MLMmazeu	JULY 4/07	60'8 Alvanyer	25,000	0.80	MTX 4/09	ሀንታ አፕግ፣	25,000

	PHOTON GLOBAL LIMITED HESTON BUSINESS COCIET 19 CAMP ROAD, WINGLESON SW19 4UTV, UNITED KINGSOM	WOYTOR & CO FOR SMITH & WILLIAMSON NOMINEES LITD. REC DEXIA INVESTOR SERVICES TRUST SL LEVEL, ROYAL BACK PLAZA SOUTH TOWER, 200 DAY STREET TORONTO, ONTARIO MSJ 213	PONTOR & CO POR SMITH & WELLAMSON NOMBRES LTD., REC DESAM DOMESTOR SERVICES FRIET SELLEVEL ROYAL BANK PLAZA SOUTH TOWERLOO BAY STREET TURDATO, CRITARIO MSJ 295	POPPEND ONTARIONER SERVICES TORONTO, ONTARIONES SERVICES TORONTO	NAMES CERTINOSCIER I
ያሁት አፒጋ፣	JLT.Y 9/07	JULY 403	144 - 141 A	JULY AMI	DATECRANT
SEPTEMBER 23/09	60% SILL IC	JANUARY 1279	жатит	JANUARY Bug	DATKINKING
23,500	2,857,142	75.1600	UNITY	900,68F	SLAVENAMACA
950	0.50	ەر ۋ	9#6	£	EXERCISEINICE
זחרג פנה	JULY 409	1015 AUD	#### 4899	nov Aller	EXPREDATE
TUP ATIO	וויף אבוו	ערג איוו	The second	#2EN 4/33	ANTAKAGRUNDEA
INIS'NZ	2,857,[4]	35.00u	=	•	BALANCE
REMINISTERED FROM JULYIN-2003	RE-REGISTERED FROM JULY07 23	SPLIT PROM	SPLIT FROM JULYOF-22R RE-REGISTERED TO JULYOF-27	SPLIT FROM JULYOT-22C AND THEN RES REGISTERED TO JULYOT-24 AND JULYOT-26	KOTES

NAME	CHRITALINGER	DATEGRANT	DATEISSUED	#OFWARRANTS	EGRUSORICE	EXTRYDATE	ANICYDEDENTIRY	DALANCE	NOTES
ROYTON & COLIUN SMITTE & WILLIAMSON NOMINGES LTD. RECIDEXIA INVISTOR SERVICES TRUST TRUST SECURITIES CAGE 155 VELLIMOTON STRUST WEST, GROUND FLOOR TORONTO ONTARKO MAY XLS	T)T,Y01-26	LTA 400	SEPTEMBER 2309	IDCERT	E.	66r veir)CLY-WI	757.500	RE-RECISTERED FROM JULYOS-23C.2
IAYVER & CO COMMERCE CHIRT WEST INVERTY STREET SIKLIKITIES LEVEL TORKNITO, ON MSLIGS	JULY07 27	JULY 4/05	JANUARY 1910	311,000	0.50	JUEV 4/09	UTA 4.111	30,000	RE-REQUETERED FROM JULY07-23R.1
SCOTT CHOSED 30 WATERSIDE PLAZA MODD, NEW YORK, MEW YORK, 10010	លខា	JLL& 4/00	JCLY 4007	20,000	0.00	301.Y 40199	10157-4/11	20.000	
MANCHISTER, NE 0304 09	10-02	JLLY 4007	AUP ATIN	60000	050	JOLY 4/09	ነነ ም አተገቦ	60,000	
KELLASIG CAPITAT, TIROOTI LLC SS BROADWAY, 4TH FLOOR NEW YORK, NEW YORK, 10006	8	JULY 407	TULY 400	1,714,286	0.50	JULY 4409	JLLY #017	1,714,286	
D. JOSEPH KKARACHER 201 N. MILL. ST., #201 ASPEN, CULDRADO, M10]]	пени	TULY 4407	100k knor	30,000	DS:0	JULY 4/59	ToF.X3nd	100,00	
KUDC PARTNERS, LP. 1900 COUNTY ROAD 124 HESPERUS, CO. 81926	115-()5	JULY 4/07	JULY 400	142,000	0.90	JULY 4009	ELP A TOU	142,000	
MST YEATHERS LLC 32% ORANOR STREET WILMINGTON, DG 19801	NS 0K	Trick avoz	JULY 4:07	260,000	0.90	JULY 4/09	TOTA ALIS	200.000	

MANGE	CORTNUMBER	DATEGRANT	DATEKSLED	NUMBER	EXERCISERRICE	PARTHADALE	AHLICHOMOMOREA	BALANCE	MUTES
AVVOCATANTE PER PROPERTY AND THE PER PROPERTY AND T	18-10ATM8F	7081 X10E	WH ATM	1999981	#59	AULY 19109	ACX 15:73	=	RE-REGISTERED TO
HAYWOOD SECURITIES INC. ITE DOROTHY ATKINSON ACCOUNT # BC2-1369C # BC2-1369C YANCOUVER, B.C., V6C 3A6	20-LOATONRE	PULY 1807	FULY 1990	між	l)SCII	MYRI A'ICIT	אמלא ושנוו	50.000	
HAYWOOD SECTRITIES BYO, THE GARY HALDWIN ACCOUNT & BCC-4405C #2000 - 400 HIRRARD STREET VANCOUVER, H.C., VAC 3A6	tBJDLY29.d3	JŲLY (3/0)	30% LATOR	00FK	950	שרורג ואטא	אווזרג נואוו	36,500	
HAYWOOD SECTIONES INC. ITE SYDNEY BASS ACCOUNT & BCI- 3986 C #2000 - 400 BURBARD STREET VANCOUVER, B.C., VAC 146	1800LX07-04	JULY 18/07	אוטובץ ואפוז	30,000	0.50	ያበተአ ነውው	וואראווונ	30,000	
HAYWOOD SECURITIES INC. ETY 150 BELISLE ACCOUNT # BRI . 3905-C 92000 - 400 BURRARD STREET VANCOUVER B.C., VECTAA	50-20A: B138:	TULY 1847	UULY 18400	42,000	D.50	MARI ATEL	AUY 1811	42,000	
HAYWOOD SECURITIES INC. THE CURTIS BEER ACCOUNT # BCI- 4821-C 42000 - 400 BURRARD STREET VANCOLYER, B.C., VAC 346	INJULYDZAK	JOSE A TITL	JULY 18407	100,000	0.50	TOEK LANN	TCEN LATE	100,000	
HATWOOD SECURITIES INC. THE BESNARDINA AND RIDENTS ROUCHARD ACCOUNT # BML- 4538-C VANCOUVER, B.C., VAT 3A6	ואינוראטי-פט	ערא ואלין	ארג וויסו	60,000	0.50	MAN KIÖL	זמרג ואיוו	÷n,000	

NAME	CERTNUMBER	DATEGRANT	DA DASSUKO	KINAMBAWKIK	KUNDANGKAN CHIMANSUMANN	KYPIEYD4'IX	ANTIONOMORCHERY	HALANCK	SHLON
JAYWOOD SECURITES INC. 1119 VICTOR BISHOP ACCIDINT WISCH- 4000C 4200C—400 BURKARD STREET VANCORVER, B.D., VOC 346	MP-UNCITIBLE	JULY 1907	20%1 AJOI	J00,000	050	30813300	JOLY 1811	000,001	
HAYWOOD SECURITIES INC. ITE DOLDER OPPORTUNITIES I LE. ACCOUNT # BM7 2226-C 4000- 400 BURBARD STREET VANCOUVER, J.C., VCC 3A6	160-707-7031	20/81 X:ICIL	1017 1807	750,000	0.50	JUST ATOL	ואנו אורא ואנו	IMMP1152	
IIAYWOOD SECURITES INC. ITF DON BOSSERT ACCOUNT #BCL 6231C #2000 - 400 BURRARD STREET WANCOUVER, B.C., MCC JA6	JSJULY07-10	אחדא ואיוע	JULY TRATE	30,000	05.0	6681 ATH	ומדג ואוו	30,000	
HAYWOOD SECURITIES INC. ITP MICHAEL BREMER ACCOUNTS BCL-1786 C. ADOW - 100 BUJUNAKU STREET, MANCOUNER, D.C.: NACJAE	HINDLE STATE OF THE STATE OF TH	THAN A BIT	THE TAKE	20,000	ŧ	WHE ATHE	TOST X UN	٥	REJECUSTERED TO
AVVOCATIONES SUPPLIES OF THE STATE OF THE ST	THEMSTANA	THE PARTY AND A	2008-1802	toillar	PPR.	WEN 1889	TIAR ADRI	Þ	RG-REGISTERED YO
NANACOGONIA BECARTERNA STREET NOOG AWARDELL VECCOUNTA NOOG AWARDELL VECCOUNTA NANACOGONIA	FT 20X7E115#	tarat Artor	that and	50,000	4	60/81-XTΩ#	T181 ATC	¢ 	RE REGISTERED IO
IIAYWOOD SECURITIES INC. ITE PAT CAMPLING ACCOUNT # BCL 9715-C #2000 = 400 BHRR ARD STREET VANCOUVER, B C., V6C 3A6	ISJULX07-14	MTA ISAU	3081 XIDE	500.06	0.50	JULY 1809	מברא ויאוו	YII,000	

NAME	CKRTNUMBER	DATIOURANT	DATRISSUED	DATEISSUED #OFWARRANTS	EXERCISEPROCE	EXPERVUATE	AMENDEDKEMMY	RALANCE.	NOTES
HAYWOOD SECURITIES INC. ITE FAR CARELLA ACCOUNT # BC1- 5730-C #2000-400 BURRARD STREET FANCOUNTR. D.C., Voc NAA	38000XB-15	זנו א זומוז	JULY 3807	25,000	0.50	זחרא ואאי	ווארענונ	25.000	
TAYWOOD SECURITIES INC. ITS C. CHANNING BUCKLAND ACCOUNT # BR2 4693 C #2000 - 400 BURVARD STREET VANCOUVER, B.C., V6C 3A6	191-00ATIN/R1	JUNE ATDI	דיונץ נפגעד	250,000	38	TOLY INTO	אורא ואינו	IMNFOCZ.	
HAPPOOD SECURITIES INC. THE COLIN COUMBES ACCIDENT # BRI-1958C #2000 = 400 BRIER ARD STREET YANKSLYER, BIG. YOC 346	נו האלידות איני	PULY 18/0/	אוורא ואטט	30000	0.50	חנדג ושש	Jמרא ואוו	0000/07	
HAYWOOD SECURITIES INC. ITF DMINATEL MIDDICAL DAC. ACCOUNT WIDCU 5853C #0000-400 BURBARD STREET VANCOUVER, D.C., VAC 346	וארויאני-ו8	10LY 18/07	JULY 1840	30,000	0.50	ארצה אנאוו	HIST X DIE	59,000	
HAYWOOD SECURITIES INC. DIV VERCHIDALA DAS ACCOUNT # BC1-4426C #2000 400 BURDARD STREET VANCOUVER, B.C., V6C 3A6	18.711.727-19	7,07.2 7,07.	JULY 18100	211,10001	0.प्य	JULY 1809	LURL KTIN	20,000	
HAYWOOD SECURITIES INCLITY BARBARA DEGLAH ACCOUNTY BC1-17604C WANKUUVER, B.C., V6C \$46	DE-CONTENIE	TOTA TAYOO	JULY 18/07	20.000	020	שנא ואלא	11:81 2700	59,000	
HAYWOOD SECURITES INC. FTP J. DEGLAD ACCOUNT & BM1- 1262C A2000 - AM BORBARD STREET VANCOUVER, B.C., Voc. NA6	ו אוווראנה יבו	וטבצ נאס?	סרג ואונ	50.000	0.50	<i>ከ</i> ቤተ ተመፅ	IIST X TOL	MON	
HAYWOOD SECURITES INC. THE SUBBARAO BYANI ACCOUNT # BC1-4250C # M2100 - 400 BURRARU STREET WANCOUVER, B.C., VKC 3A6	1 6 3ULX07:22	1011 A 1803	JOLY 1807	000.001	0.50	ורג זאאו	HSTATOL	100,000	

NAME	CERTNUNDER	DATEGRANT	DATEMENTAL	HIPWAREANTS	КХЖИЗКИРВЭСК	ALVELANIANS.	AMENDEDKAPIRY	BALANCE	SKLON
BAYWOOD SECURITIES UNG ITE 42190, U.C. LTD. MICTORNITHECH \$1194-C HORRORD STREET VANCOUVER, B.C., VGC 3A6	18,000,007-23	DLY 1807	JORI JANI	25,000	0.50	JULY 18/09	JULY 18:11	25.000	
HAYWOOD SECURITIES INC. ITF WOHN PRIFINAN ACCOUNT # BCJ- 41VIC: #200 - 400 BÜRMAKU STRILET VANCOUVER, B.C., V(C) 340	15-00.4-10181	JULY 1807	LOVRI ATINC	₩	0.50	JULY LB/09	עורג ואוו	40,000	
IGYWDOD SECLRITHES INC. THE BODGLANGAIN ACYDUNT # BCI- 4485C VANCOUVER, B.C., VM: YAO	1817.797-25	JCHLY LOOS 1 A-HI JC	7081 7207	CONTE	ţ	GORL ATOR	HZEL KITIF	15,000	
HAYWOOLDER BYCTAY BENT 10000 BYL 10000 WHERE WOOD OF THE BYTAN ON THE BY THE	*C 2004 WINTE	tâât A'Itif	ESPE A TEN	45,000	o.do	10081 X12k	41:31:3-TE-14	٥	REREGISTEAED TO
HAYYOOD SECURITIES LYCLITY LAD GORDON ACCOUNT # BKZ 1986-C VANCOLVER, B.C., V6C 3A6	18JULY07-27	JULY 1807	JULY 1960	400,000	0.50	ULY 1809	לוובץ גאניו	440,000	
HAYWOOD SECURITIES INC. ITF RRIAN GUZE ACCOUNT # BCT- MC07-C #2007-400 BURRARD STREET VANCHUVER, B.C., V6C 3A6	R2-2037L7881	JUEY 18/07	AOFA TRIOS	5,000	0.50	JULY 18/09	בונא ואטו	5,000	
HAYWOOD SECURITIES INC. ITP BOBILINDA HAMM ACCOUNT # BC1-55920 #2000 - 400 BURRARD STREET YANCOUVER, B.C., V60 546	67-JOA-TUIB1	אָראָרו אַ וְחָלֵּ	אטרא וגיים.	.10,000	050	SQRI ABOF	70EX 18/11	50,000	
HAYWOOD SECURITIES INC. ITE MONICA HASSETT ACCOUNT # BC1-4384C #2000 400 BURRARD STREET VANCOUVER, B.C., V6C 3A6	ISIL LYDY-NI	אמרא ווייטע	JULY 1807	47MH+	0.50	אררא ואטא	11/81 A70x	42,1MN1	

NAME	CERTNUMBER	DATEGRANT	DATEMENTA	SINVABUALINE	EXERCISEPRICE	EXPERYDATE	AMEMBEDEXPIRS	BALLANCE	
HAYWOOD SECURITIES INC. ITF KIRLYEN HERINGA ACCOUNT # BCT-4113C 9XKO-400 BURRARD STREET YANCOCYER, B.C., Y6C 3A6	IBIUCY9744	JULY 1860	AULY INDO	40,000	0 70	JULY 1809	JARI AJOL	INNESIC	
THE TAXABLE TAXABLE TO TAXABLE	ऽहें कीक्ष्रमताव ी	UNI ATOL	tent tran	\$C,000	P ₅	dibility Actin	11/81 - KT0#	٥	RE-RECUSTERED TO
ITAYTYOOD SECURITIES INC. ITP INCRED HEBBARD ACCOUNT & DCI 4605C M2000-400 BURRARD STREET VANCOUVER, B.C., VCC JAG	18117780-33	אטרג וצשנ	ADAI ATH	ro,000	0.50	JULY 1509	JULY 1811	10,000	
HAYWOOD SHOWER HER HAS THE STANLEY HOLSTAD ACCOUNTS # BCH 4029C	# CUA WERT	THE SECTION	BJE V LBACE	43,000	8-5-9	KORL KTAF	tt/m-x-31/f	Ģ	AU-RECUSTRAND TO
JANNE HINTZ ACCOUNT # BC1- 4139-C VANCOUVER, B.C., V6C 3A6 VANCOUVER, B.C., V6C 3A6	DE-SILATIONE	ZORU A'JÜL	JORN LATOR	21000	930	AUSI ATIOL	זרגזאנו	25,000	
MAYYUGO SHIJIR TIRSINC, ITE MISUNJIUNG ACCOUNT # BC2- 5280-C #2000 - 400 BURRARIT STREET VANCOUVER B.C., VCC 3A6	9E-03ATH781	JCLY 1807	JOLY 1807	19,000	0.50	חברא וגאטפ	Hist Arm	000YI	!
#AYCHOOD SOCUMENTS ONC. FIT MARIA ILIQ AYCHUNEN BCI 4823 Q #AANCOUNEN B C., VAC JA6	14.00V.2000 8:2	1981 Xmm	DAST A'BU	coolee	##	SOTE ATIC	HISH ATEL	۰	RD-REGISTERED TO
HAYWOOD SECURITIES INC. ITE MAXIALLAD ACCOUNT # 852- 5275-C #2000 - 400 BURLAND STRIFFT VANCOUVER, B.C., VOCTAK	181ULY107-3K	JULY 1880	105K JAOL	DHNFC	0.50	JULLY LYON	TOTA 1897	1.000	

NAME	CERTINUMBER DATEGRANT	DATEGRANT	DATEISSUED	SENVARIVADOR	SCHECKSONICE EXPERSIONE	EXPIREDATE	AMENDEDESCRIPT	BALANCE	NOTES
HAYWOOD SECURITIES INC. ITS LIBERYLIUM ACCOUNT # BCT- 41MC #2000 - 400 BURRARD STRJET VANCOLVER, B.C., Velt 3A6	ISULTURE	JULY 1800	101.Y 1807	35.000	œر	זהרג נושש)GLA 18/11	25,000	
JIA YWOOD SECURITES INC. ITE M. KALAIRAIAH ACCOUNT A BM1-5459-C M2000 - 400 BLERARD STREET VANCOUVER, B.C., VIC. 346	1800270740	ICLY 1807	JULY JWIZ	GRIUNZE	930	BON ATO	ICTA IBILL	1000,000	
HAYTAGO SECURIUS INC. ITA STAN KASON ACCOUNT * TRL 4275 C. 1000 — 400 BURKARD STREET, VANCOUVER, B.C., 1460 346	th cox muss	ebăt x tin	DILT TANG	130,001	F	00081 X 1117	11/81:472#	.	OT CONTRIBETS TO
HAYWOOD SECURITIES INC. ITE KENSHAW ENTERPRISES LTD. ACKIONNI U DCI-3 183-C ACKIONNI U DCI-3 183-C VANDDI VER, B.C. V6C 3A6 VANDDI VER, B.C. V6C 3A6	18000742	AULY 13/07	JULY 18/0/	оори	0.50	JULY 1809	JULY (8/1)	:скать:	
HAYWOOD SECURITIES INC. ITP KLASSIC-FORE INVESTMENT INC. ACCOUNT # BMT-4-212C #21887 400 BURRARD STREET VANCOUVER D.C., VOCTAM	1847.CIBU	עורא ואסט	JULY INIT	100,000	0.50	101X 1809	13/81 A700	100,000	
HANWOOD SECURITIES INC. UP LAMOND INVESTIMENTS LTD. ANTICKNE # BCI-56720 VANCOUVER, B.C., VAC 3A6 VANCOUVER, B.C., VAC 3A6	**************************************	JIII Y 18407	TOLK 18407	CREMIEL	D.50	MAR ATHE	11/81 ATOL	190,000	
IIAYWOOD RECURITIES INC. ITY MICHAEL LATERCEE ACCOUNT & BCI-6107 C #2000 400 BURRARD STREET VANCOUVER, B.C., V6C 340	TRICOLYB7-45	JULY 18KI	JULY 18/07	71,100	05.0	AULY 1809	LIASL K'IIIT	71,000	
HAYWOOD SECURITIES INC. ITE MARK LEATHER ACCOUNTS BICH-4793-C #ZINO - 400 BURBARD STREET VANCOUVER, B.C., VKC 3AA	18101707-46	TOPE LEVOL	JULY 1967	160,000	0.50	KILY IXIO	TINSE ATIOU	Jerman	

DAYN	CERTAUMEER	DATEGRANT	DATESECTAC	SOFWARRANTS - EXERCISE	BUCE	EXPOSSIBLE	ANDENDEDEXPORY	BALANCE	MILIDA
HAYTWOOD SECCRETIES INC. THE JORD LEBLANC ACCOUNT A BOL 4162 C. TURO — 400 BURBARD ATT TOTAL VANCOUVER, B.C. 1460 346		TOTAL STATE	DOWN ATHE	40,000	95.86	6051-X308	1981 A 1981	=	RE-REGISTERED TO
HAMMOOD SECURITIES INC. ITE KEN LEK ACCOUNT MBR2:2622C #2000 - 400 BURRARD STRUET WANCOUVER, B.C., VIC 3A6	*F-20A7D481	JULY 1907	JULY 1887	10,000	050	JULY 1809	אחרג ואור	111,000	
AVANCOTATE TO ACT AND TOTAL TO ACT AND TOTAL TOT	UP-LÜKYLYINI	1017. 1500	POST ATH	HHTE	P-\$6	6681.XTM	11/81 3/10/	۰	RE-REGISTERED TO
DAYWOOD SECURITIES INC. IN- TERRY LEG ACCOUNT # BRJ. 524NC 924NC - 400 BURDARD STREET VANCOUVER, B.C., V6C JA6	IN-20A'TEBSE	2031.ATH	JULY 18X7	39,000	05.20	50/E1 J.TINE	נני\$נ אמטנ	CRNTGE	
WANTOUNER, B.C. VEG 3A6 WANTOUNER, B.C. VEG 3A6 WANTOUNER, B.C. VEG 3A6	HIGHADIT	iost Amir	WEL KITH	3A,000	#	60%1-KHR	THANK! ATMIT		REREMEDIAND TO
HAYWOOD SOCIETIES THE FITE ALEXANDER LIND ACCOUNT # BCT 623940 VANCOUVER, B.C., V6C 3A6 VANCOUVER, B.C., V6C 3A6	18001207-42	JCCY 1807	JULY 1807	25,000	040	JULY 1809	ברא זופוו	25,900	
BAYWOOD SECURTIES INC. ITP JAN MACDONALD ACCOUNT # BCJ-4830 C #2000 400 BURRAND STREET YANCOUVER, B.C., Y6C JA6	i&ruLY07-53	JULY 1880	YULYY 18405	MONOR	0.50	JULY 1809	חומומוו	50,000	4
HAYWYOO SECURITIES INC. THE MARLENE MACDONALD ACCOUNT HECLARD STREET 42000 400 BUSHAMD STREET VANCOUVERLESC, V6C 3A6	18TULYU7-54	FULY 1807	TULY 1807	SHARKI	0.50	SONE ATTUE	JULY 18:11	50,000	

NAME BAYWOOD SECURITES INC. THE ELIZABETH MACKLEZIE ACCOUNT & BIM LOSSE ACCOUNT & BIM LOSSE YANCOUNTER, B.C., V6C 3A6 BAYWOOD SECURITIES INC. ITF PAUL MCKENZIE ACCOUNT & BCC, 400 BURRARD STREET VANCOUVER, B.C., V6C 3A6 MAYWOOD SECURITIES INC. ITF	CERTNUMIKE	DULY 1807	DATEGRAPHED JULY 1800	\$0,000 \$0,000	DAZRCIMETRICE 0.50	JULY 18409	JUCA 1840) JUCA 18411 JUCA 1840) JUCA 18411	\$0,000
PAUL MCKENZIE ACCOUNTS BYLL MUKENZIE ACCOUNTS BYLL MARCHANIC WANCOUVER, B.C. V6C 3A6	EUCLY07-56	TOTA 18904	MOLVIUM	50,000	0.50	6081 JUL	JOEA LEAL	50,000
AVACAGAMENT BECT AND TAXABLE WOODD THE BITTER WAS TAXABLE WATTWEEN THE WOODD SECRETARY WOODD SECRETARY WOODD SECRETARY WATTWEEN THE WATTWEEN THE WOODD SECRETARY WOODD SECRETARY WOODD SECRETARY WATTWEEN THE WOODD SECRETARY WOODD SECRETARY WATTWEEN THE WOOD SECRETARY WATTWEEN THE WOODD SECRETARY WATTWEEN TH	787CLY97 57	WATE ATOM	TELL YES	880°FR	95-0	6681 21.21	ינורא ואירו	Ф
HAYWOOD SECTIFITIES INC. ITS RICHARD MAKK ACCOUNT A DCI 4168C P2000 - 400 BUHRAND STREET VANCOUVER, BAL, YMCIAA	182ULY05-58	NCTA 18002	JULY 1807	20,000	0.50	ANS LATE	ווינדא ואטו.	20,000
IIAYWOOD SECURITIES INC. THE JOSEPH MARTIN ACCOUNTY BCI \$720C \$720C +2000 - 400 BURRARD \$18 EET VANCOLVER, B.C., VACCIAA,	65-20AThrei	JULY 1807	זענע ואסד	20,000	0.50	זטרג ואאו	ויטר צוווע	20,000
HAYWOOD SECURITIES INC. IT FENNIFER MORPORD ACCIDINGS BRI-1685C 400 BURRAND STRIFF VANCOUVER, BAC, VAC 546	18077.407.60	יוטדא ואינע.	זטנא נגשי	oot,n;	0.50	TOLY THUM	שנא ואָנו	50,000
HAYWOOD SDOURTIES INC. 178 BARRY MUDI ACCOUNT Y BR2 4932C 420W6-400 BURRARD STREET YANCOUNTER, B.C., Voc 3A6	19-003-17036	JORI ATOR	JULY 1897	100.000	0.50	60/83 ATOL	אמרא ואָרו	100,000
MAYWOOD EMCURTHANDS, HY KEN MURDHY ACCOUNT REQ 88640 43000 - JOU MLARARD STREET VANCOUVER, R.C., V6C 3A6	to and the	2008 HATCH	test Afor	enembys	94-9	WOM! ATOM	HASI ATOR	

KAND	CERTNUMBER	DATEGRANI	COUNCIPALING	MOPWAKRANTS	EXERCISEPRICE	EXPERIENT	AMENDEDEXPORT	BALANCE	SELLON
HAYWOOD SECURITES INC. IT RICHARD OLIVER ACCOUNT # BCI-5317C *2009 - 400 BURBARD STREET VANCOUVER, B.C., Voc.346	IRAILYCTAS	JULY 18917	FUR 1 8/07	25,000	0.90	JULY 1809	IULY 1811	25,000	
HAYWOOD SECURITIES INC. ITE JAME O'NEIL ACCOUNT # BEL 2902C #2000 - 400 BURDARD STREET VANCOUVER, B.C., V6C 3A6	187ULY07-64	ANT ATOM	JULY 1807	9 0,000	0.50	It ID.Y. 18400	ואנא ואנו	50,000	
HAYWOOD SECTRIFIES INC. IN MARCIE ONEILL ACCOUNT & BCL-5217-C #2000 - 400 BURRARD SINGET YANCOUVER, B.C., V6C 3A6	THTC/TAIL-VI	THE YOUR	JULY 1807	мироо	orno	301.7 ISOS	YOOY IBIT	90,000	
ITAYWOOD SECURITIES INC. ITS PINK HOLDINGS INC. ACCOUNT # BCI-2748-C WANCOUVER, B.C., YEC \$46	18 CTAID-04	JORI KIOL	וטנץ ואַסי	30,000	0.50	SENT ATTI	ומרג ואנוו	MOOLIK	
HAYWOOD SECURITIES INC. ITP PEPPY HOLDINGS LTD: ACCOUNT: # BCL-4060C m2000 - 400 BURKARD STREET YANGOUVER, B.C., V6C 3-46	75-FOYLINKE	RULY 18/07	אנבא ואמנ	25,000	0.50	אורא ואספ	JIHAY KATI	25,000	
KAYWOOD NECURITIES INC. ITP LOKNE PUTNAM ACCOUNT (BK L44004) VANCOUVER, B.C., V6C 3A6	SP-CHATCIFIT	JULY 1860	20R1 ATEC	100,000	0.99	אונץ נאגא	YUN YUN	100,000	
HAYWOOD SECURITIES INC. ITP GUIDMER KARDLER ACCOUNT # DCI-3384.C #2007 - 400 BURRARD STREET VANCQUYER, B.C., V6C SA6	1810/1707-49	אטבץ נופוז	MAI ATOL	EVANO	050	MUN 1809	אברג זומור	25,000	2
HAYWOOD SECURITIES INIT IT DIV RANDRAWA ACCOUNT # BCT-1499C #2000 - 400 BURKARD STREET VANCOUVER, B.C., V6C 1A6	18ULYAT-70	JUCK 18407	1081 3301	GRRITIE	050	111.7 R09	אברג זאווו	50,000	

NAME	CERTINUMBER	DATEGRANT	DATEISSUED	миктинимия	X HARDEL HEATH	ВТУСЛЯНИХН	VMENDEDENGEN	BALANCE	STLOK
HAYWOOD SECHRITISSING ITS KIVERSPILL RESPONSE LED ACCOUNT # BCT 4511 C #2000 - 4HI BL REARD STRBET VARCHUVER, B.C., VGC A66	1870LY07-71	JULY 1807	JULY ISIT	47,000	D 50	TULY 18709	JETA 18-CC	47,000	
HAYWOOD SECURITIES INC. (ITS GARY BUSS ACCOUNT # BCI 4836C 9ANOCUMER B.C., V6C JAJ VANCOUMER B.C., V6C JAJ	38.AULY07-72	JIRY 1807	ATA JAQU	28,000	0.50	MARL ATIDE	וואָראנונ	0000'872	ĺ
HAYWOOD-SECURITIES INC. HX 6156-05 GAMADA-BED, ACCOUNT #-BC1-4304C #-800100 BURBARD STREET #-APPCOUVER, B.C., V6C3AA	12.111.12.7X	THE PARTY HAR	DDF81 XTIDI	M00'C11	F	W 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	HIRLAND	۰	NE ALGESTERED TO INCLLYOF-107
HAYWOOD SECURITIES INC. TO 47883 B.C. LITE, ACCOUNTY BCI 28310 28310 - ANI BURBARD STREET VANCULIVER, B.C., VAC 346	18JULY07-74	60/81 KTIOL	ረሰጫ ሊገበር	Barros	0.80	100% F898	JULY 18/11	515,000	
HAYWOOD RECHETERS INC. 878 977-167 B.C. LID. ACCOUNT # BG1-307.K. PAREGUNER, B.C., VGC JA6 PAREGUNER, B.C., VGC JA6	ACCOUNTED	TULY TROST	20 81 x 101	dajana	g.	DONE ATOM	LOTA 1841	e	REREGISTERED TO 18/ULV97-107
HAYWOOD SWILWITHSKING THE SUREL SAMILACCULAR FIRET SUASC MANDULYER, R.C., 146C3A6	18JULY 24	1981 XJU	All V-1827	34,000	ŧ	9081-XTIV	NTX 1611		OH-46ATD#ST AL GRBBLSIOGHBB
IIAYWOOD SECURITES INC. TU ROLAND SCHEMEL ACCOUNTY BCI-2733-C 92000 - AN BURRARD STREET VANCOLIVER, B.C., V6C JAD	いるというない	JULY 1807	JULY 1807	300,000	010	60/81 JATOF	THEN KARL	100,000	
HAYWOOD SECURITIES INC. 109 KEN SCHEPREK ACCOUNT V BUI. 40400 YANDUIVER, B.C., V6C 346 VANDUIVER, B.C., V6C 346	E4-40ATOTRI	מורא ואסג	JULY IMB	47,000	0.00	JULY 18/09	אזויא ואינו	42,000	

NAME IIAYWOOD SECURITIES INC. ITE SHREEPADA HOLDINGS INC.	CERTNEMBER	DATEGRAPH	DATHASUED	ROETWARRANTS	EXEBCISEPRICE	ENHKYDATE	AMENDENDEXPIEN	ABIAXO
IIAYWOOD NECURITHNING ITF SHREEPADA HOLDINGNING. AGCOUNT # HOL41720 COXIO—AINE BURRARD STREET VANCOUVER, B.C., V6C 3A6	181017402-79	תונץ נאסי	JULY 1807	2000,000	05a	668) ATN	2	IINI ATIN
HAYWOOD SECURITIES INC. ITT HOMEST SEK ACCOUNT A BRI- 2776-C #2000 - AGUBURKARD STREET VANCOUVER, B.C., Y6C 3A6	DB:40ACTIRE	JOLY IMP	JI 2.2 1897	30,000	0.50	JULY 1809	<u>s</u>	וושראווא
HAYWOOD SHOURT IS SNO, IT SOUTH EAST ASIA FIRED LITE. ACCOUNT A BIOL-SIGH. ACCOUNT A BIOL-SIGH. VANCOUVER, BALL, NGC 346	18-CACTARE	ነውያር አገባር	אלורא נאטב	190,000	0.50	MOLE VALUE	<u>_</u>	וניאו איזו
RBC DEXIA BIVESTOR SERVICES TRUST FIF ACCOUNT 110 455-C40 SOUTH TOWER, ROYAL BANK PLAZA 200 BAY STREET, SL LEVEL TOKONTO, ONTARRA, MAI 215	18JULY07-62	200EL ATION	אמרא ואינט	47.100	0.50	ארורג ושנפ	₌	הרג ויינו
RICOLIXIA INVESTOR SERVICES TRUST ILIVACCOLDIT 111-440-001 SOUTH TOWER, KOYAL BANK PLAZA 200 BAY STREET, SULLEVEL TORONTO, ONTAKRI, MSI 235	ER-docturer	JCLY JB07	JILET YEAR	50,990	0.50	פאשר עונונ	2	MLX \$811
RRC DEXIA BYSESTOR SAMPLES TRUST ITS ACCOUNT (10 ASSESS) SOUTH TOWER BOYAL BANK PLAZA 200 MAY STREAM, ST. LEVEL TORONTO, ONTARTO, M.SJ. 235	# tBJULYIN-H	FULY 18407	ANTA 1900	1,124,901	0.50	ADVRI ATINI	≧	ארג נאזז
COMARK SECURITIES NO. SUITE 2000, SOUTH TOWER ROYNL BANK PLAZA TORONTO, ONTAKIO, MILI 202	נפוניר גניישני	70/81 A.U.M.	YJUK 18407	716,400	0.50	JOLY 18WN	JGE.	JULY 18/13

NAME	CERTAUMORIA	DATEGRANT	DATIGNSHIDS	SINVARVABOR	EXERCISEPRICE	EXPERYDATE	AMENDEDEXPURY	BALANCE	NOTES
RECIDENTA INVESTOR SERVICES TRUST THE ACCOUNT 086-284-001 SOUTH TOWER, ROYAL BANK PLAZA 200 BAY STREET, SLUEVEL TORONTO, ONTARBO, MSI 215	発売がいない	70LY 18/07	אטנא נאשף	9,400	0LD	ecubi XII.II	ንወርሃ ነው! ነ	9,400	
RHC DEXIA JUNESTOR SERVICES TRUST HE ACCOCNTT 12315721 SOUTH TOWER, ROYAL BANK PLAZA 200 BAY STREET, SL LEVEL TORONTO, ONTAKIO, MSJ 216	181/LYW:87	JOLY 1907	4.0431 XTOP	22,300	0.50	साम राज	IIMI KICIE	22,300	
HAYWOOD SECURITIES INC. ITH WARKEN STANYER ACCOUNT # BRJ 423M. 420M - 400 BIRK 4 RD STREET VANCOUVER, B.C., VIC 3 A6	18.FULYOO 88	שני זאה	DYR A HIL	50,000	0.50	JULY 1869	וואו אווונ	50.000	
HAYWOOD SECURITIES INC. ITE GLEN AND/OR CAROLE AN NE STOCKER, ACCOUNT # BC 5272-C #2443-400 BURRARD STREET YANCKHIVER, B.C., V6C 3A6	64 t9A7nir	JULY 1907	10FX 18/01	10,550	li Si	JULY 1809	נודא זאיז	terfot	
HAYWOOD SECURITIES INC. FIT BRENDA STEUNENBERG ACCOUNT # BC1 4336C #200 400 BURRAND STRBET VANCQUYER, B C., Y6C 3A6	46-50ATTRE	טכרג ופעוז	זטנצ ואחד	90000	050	אוורא ואמא	אַנוצ וואַנו	30,000	
HANNOOD SECTERITES DOT, THE BLLGTRATULAT ACCOUNT ARL ADDOCHMENT AND BURELARD STREET, YANDOUVER, S.C., VCC 400	16 20A mrst	SHIRE ATTR	DARL ASH	\$9.490	P-58	WALKER.	HMt xmm	0	RE-REGISTERED TO
HAVEOGREE, B.C., VIC. 376 1980 - 160 BURBARD STREET 1980 - 160 BURBARD STREET 1980 - 160 BURBARD STREET	20 CUX TINER	WITATION	THEX BAGS	AMANKI.	9.50	cost Aftu	HINL LTH	o	RE-REGISTERED TO

NAME	CKETNUMBOO	DATEGRANT	CHUSSIALVO	HOPWARRANTS	EXERCISEPRICS	EXPIRYUATE	ANIMONOMINA	HALANCE
HAY WOOD SET IR THE INC. ITE ALLO TRIMETH ACCOUNT # RC1- 5534C #200— WILL BURRARD STREET VANCOUVER, B.C., V6C 3A6	E4-10.KTMSE	1098 ATOR	JULY 18A07	IMMI, INC.	lk:n	JULY 1309	TIREKTON	90000
HAYWOOD SECURITIES INC. ITS SAINCTURBAT ACCOUNT # BCI- 4299-C 4299-C VANCOUVER, B.C., V6C3A6	18701.707-94	מאו אנת	70LY 1807	25,000	0.00	ADIT INC	אורא ואנו	25.188
HAYWOOD SECURITES INC. ITE WHITE CANADA BITD ACCOUNTA BELLAMBHE WOOD - 400 BHIRBARD STREET YANCOLVER, B.C., V6C3A6	16 נפעראטז 76	חורא ואיני	IDLY 1809	100,000	0.50	JULY (BOD)	אסרג ואָנו	Синстин
HAYWOOD SECURITIES INC. ITE CARSTEN WEISS ACCOUNTS BRI-2397 C #2000 400 BURRARD STREET VANCOUVER, B.C., VAC YAA	18JLZ-Y07-96	TULY 18/0?	100 KT71	000000	0.40	SANL ATEN	1017 18/11	30,000
HAYWOOD SETURITIES INC. ITE JUDITH WIZES ACCOUNT # REJ- 449h-C #2000 - 4CUBURRARD STREET YANCOUVER, H.C., YAC 346	เลมเมชาย	אונאר אבואר –	JIII.Y 18407	90.000	F.78	60/BLA/JOF	זוחרג ואיזו	-00mb:
RAYWOOD SECENTIES INC. JTF FREDERICK WHITE ACCOUNTS BC1-4134 C #2000 - 400 BURKARD STREET VANCOUVER, B.C., VKC 3A6	78-COATORET	JULY 18/07	JULY 1889	25,000	O.V.	60% I KTur	שרג ואנו	25,000
HAY WOOD SECURITIES INC. ITP JOHN WRIGHT ACCOUNT A BC1- 4844C #2009 - AW BORRARD STREET VANCOUVER, B.C., V6C 346	1810LX07-94	אחרג וואט:	JOLY I NIN	65,000	0.80	LCTX 1800	11/81 2/13t	(A), (A)
HAYWOOD SECURITESSING, ITE JAMES XU ITENAKK XU ACCOUNT # BC1 6263 C #2000 - 400 BURRARD STREET YANCOUVER, B.C., V6C 3A6	18112-1417-100	JELY 1807	SURT ATOR	HMILLIE	0.40	6081 KIDE	וואנאזמוו	90,000

NAME	DESCRIPTION BACK	DATEGRANT	DATEISSUED	#OFWARRANTS	EXERCISEPRICE	EXPONUDATE	AMENDEDEXIMIN	BALANCE	SZLOK
KOYTOK & CO.	IBJULYITARAIL.	SINSI ATDI	TULY ISAT	4,285,000	0.50	60.81 XTU	HALAER	4,285,000	
NESRITE BLENS ITE JOHN LEBLAND ACT 225 ZIM 15, ATTN- DOCUMENT MANAGEMENT I PIRT CANADIAN PLACE, 35 ¹⁸ PLACE PLACE PLACE PLACE REGISTO, ONTARIO, MCX, 183	נאטוראטט-וווו	JILY IRET	DBCB/dBeR. 21,97	40,000	0.50	JULY 18/09	JULY 1811	40,000	RE-REGISTERED FROM
PI FINANCIAL CORP CIT STAN KARON, RRSP ACA 049-1079-0 1900-666 BURRARD STREET VANCOUVER, R.C., V6C 3N I	: 201-20 A JIMR3	JULY 1807	VICCLT-VS	IMMFIXM	45.7	SOUTH ATTHE	ALL STUT	400,001	RE-REGISTERED PROM
INVESTIOR COMPANY (TH JOHN) HEWILETT, A/C 301.844E CM) TO WATERIJOUSE CANADA INC. 77 BLDOR 51, WEST, 3 th FLOOR P/J. BOX 3000, STATION "F" TOWONTO, (IN, MAY 2T)	1807.1007.100	אחרג ואַאָא	FERRUALV 2409	80,000	0.50	אבדג וויאא	זמנא נגיוו	80,000	RE REGISTERED PROM (SITIL) VOI-12
ENVESTOR COMPANY ITERLI STRATULAT, AND 1674228 CNO TO WATERHOUSE CANADA ENC. TO BLACK ST WEST, YOU FLOOR PO. BOX 1999, STATION 'P' TORONTO, ON, MAY 271	IBTILCYON-104	JULIA LATUR	MARCH 1009	50,000	E.	7ULY 1800	MINTYLON	. —	RE-REGISTERED FROM
INVESTOR COMPANY ITS MICHAEL BREMILK, AKT 241851 A GOO TO WATLEHKINSE CLANADA INC. IT BLOOK ST. WEST. 3 ⁶⁰ FLOOR PAR BOX 5999, STATION "P" TORDISTO, ON, MAY 2TI	. אטראוא-נטע	70EX 1807	MAY 2009	20,000	0.50	ADEA ENDA	JULY 18/tt	ZHERN	RE KBOISTERED PROM

NAMO	CENTYUMBER	DATEGRANT	DAYMSKIND	SLANDANAMER	SON ARRANTS - EXERCISEFRECE	EXPORTEDING	VPGATAGEMENT	BALANCE	KHIDA
INVESTOR COMPANY ITE MAJA REPMER A/C 475H5ZA C/O TO WATTERHOUSE CANADA INC. TO BLOOK 51, WEST, 3 ^{MI} FLOOR P.O. BOX 5999, STATION F TORONTO, ON, M4Y 2T1	18TULY07-106	אמז אשני	NINE 1900	30,000	0.50	JULY 1909	זמרא ואָנו	20,000	RE REGISTERED PROM
CANACCORD CAPITAL COREGRATION PO BOX JOST PACIPIC CENTRE 2200 - 609 GRANVALLE STREET VANCOCVER, BC VYV JH2	ופוטרגטי-זטז	JUCY 1807	AUS! ATDE	excited	03.0	11/81 2/2/1	N.	191,000	RE-REGISTERED PROMI (SULTY)7-73, DXIULTY(7- 75 AND (SUILTY)7-72
CANACORD CAPTAL CORPORATION THE DAVID LICK CORPORATION THE DAVID LICK ACM 198-8325-1 PO BOX 10337 PACIFIC CENTRE 2000 - 609 GRANVILLE STREET VANCOUVER, BC VVV 193	וויובדאסט וואן	JULY 1896	DULK 1909	26,000	0.50	TEST ATON	YıK	26,000	RE-REGISTERED FROM
CANACCORD CAPITAL COMPORATION OF MUTRAY GAVEL ACCIDENCIPAL PO BOX 10357 PACIFIC CENTER 1200 - 405 GRANVILLE STREET VANCOUPER, BC VOY 11Z	18ULYW-189	JULY 1808	DILY 1509	40.000	Deal	JULEA 18/11	NA	HOLLOR	R5-REGISTERED FROM 18JULY07-26
CANACTORIO CAPITAL CORPCIRATION ITE SUBILL SAMJI ACY 1901-1631 PO BEX HEST PACIFIC CENTRE 2001-691 GRANVILLE STREET VANCKUUVER, BC YOY 1112	18,07,797-410	זחרג ואטט	60ST ATDR	34.000	050	10% L XTDC	NIA); (x)	RE-REGISTERAD FROM 1800LY97-76
CANACCÓRD CAPITAL. CHEPORATION ITP MARION LIGE AC# 198-1945-1 PO BOX 10357 PACIPIC CENTRE 2200 - 659 GRANVILLE STREET VANCOUVER, BC. VIY 1H2	18JULYOT LII	JELY 1807	ALGUST 1209	45,000	16.50	נואן געע	N/A	tý,oðu	RD REGISTERED FROM 187011707-49
RDC DOMINION SECURITIES INC. ITT, IAN CAMPRELL, 3821501511 PO BOX 50, ROYAL BANK FLAZA TORONTO, ONTARIO MSI 2W7	1601112/07-112	SCLY 1847	ALXHUST 1209	100.000	0.50	ננשו אמת	Nia	100,000	NEWSTREED FROM

NAME	CERTINIMBER	INVESTME	DATEMENTED	#OFWARRANTS	EXERCISEIPLOX	ENTRYDATE	AMENDEDERGERY - HALANCE	BALANCE	NOTES
MIC DOMINION SECURITIES INC. ITF, KAREN CAMPBELL, 1824105216 PÜBOX MI, ROYAL ÜANK PLAZA TOMONTO, UNTARIO MISTANI	1811/7/27-113	אטנג וצעס	ACOUST 12003	900008	or po	TOWE A TOTAL	N'A	50,400	RE-REGISTERED FROM
CANACCORD CAPITAL CORPORATION ITP STANLEY HILSTAD AKT 198-0027-1 PO BOX 10657 PACIFIC CENTRE 2200 - 469 GRANVILLE STREET VANCOVERS, 8C - VTY 1 H2	18117702-114	JULY 1807	SEPTEMBER 809	42,000	0.50	II ISE ATIN	4,K	42,000	RE REGISTERED PROM
NVESTOR COMPANY JU- KENDALL R. MIRPHY A/C SAE267F COTTO WATERHOUSE CANADA INC. TO BLOOK STREET W. 3 ^{FD} FLOOK PO DOX 5999, STATION 'F' TOBONTO, ON MSS 13/C	SL-2UN INTH	200s XIII	KARICICOO	2000/116	H.H.	וואו צוווע	AW	90000	REALDONSTERRID FROM DRITTLYNT-AZ
INVESTIGE COMPANY ITE DAVID R. MALARCHIK AJC 738446E C/O TO WATERHOUSE CANADA INC. 77 BLOOK STREET W. 3 ³⁰ FLOOR PO BOX 5994 STATION "P" TORONTO, ON MAS IM2	911-05X1/0/8E	אופאר ופאר	OCTOBER 2209	Deut'sz.	07.0	יונידג 1941	N/A	26,000	RE-REGISTERED FROM 1870LY07-57
CANACCORD FINANCIAL LTD, ITF GERALD ADAMS ANC 19D-916S 2 PO HOX 10337, PACIFIC CENTRE 2XIO - 6/9 GRANVILLE STREET VANCOLVER BC VYY 1H2	211-20171081	תונצ זאמי	DECEMBER	толуни	0.50	JULY 1841	* FIR	160,000	RE-REGISTERED FROM 12UE/1970)
PASSPORT MATERIALS MASTER FUND, CI* C/O I*ASSPORT MANAGEMENT. LLC XI HOTALING PLACE, STB. XKI SAN FEADYCISCO, CA. VALIT	PASSPORT t	AUGUST 9/07	AUGUST 9/07	оосринге	050	ALKILET 900	AUGUST9/11	exitooris	

MAMR	CEKTAUMOER	DATECKANT	DATEMENTER	HIDWARRANTS	EXPROSEDENCE	EXPERIDATE	AMENDEDECORY	BALANCE
GOLIMAN, SACIS & CO., AS NOMINEE FOR THE REVISIT OF ORDIT OF CROLOGIC RESOURCE FUND LP 30 HUDADIN STREET 128NSEY (2TY, NJ. 07302	OS-AUGODOL	ALKIUST 1600	AUGUST TAND	742,857	0.50	WINDOW 1800 IV	AUGUSTIWII	742.817
ROYTOR & CO. ACCOUNT RTI 20585]] RBC PLAZA, SOUTH TOWER SERVICE LEVEL, 200 BAY STREET TORONTO, ONTAKIO, MY 215	LS-ACKOTAD	AUGUST 1607	AHRHET IANG	3,000.000	020	AUGUST 1609	אטטנקד ואיון	3,000,000
GOLDMAN, SACHS ACCT., AS NOMINEE FOR THE BENEFIT OF GEOLOGIC RESOLUTED HIND, LTD. 30 FILIOSON SCHEET JERSEY CITY, NJ. 07302	l6AUGII7-DI	VACTEL	WHAT LENDAY	2,114,286	930	AUQUST 1609	ACCUST 1811	2,111,286
SCOTIA NCLEGO ETFLIMINYO MARKET DEALER INC. ACCOUNT 439-0825-13	10-AY		JULY 400	21,A2H	0.50	JULY 4409	N/A	٥
TANCHTARBUNIAL TANCHTAR TANCHT	AW-02		JULY 407	943,714	0.50	JULY 4/09	A.M	0
BOLDRE (NYBSTMENT PARTNERS LTD.	10-WA-cing		JULY (BAD)	*75,590	IESI	MILX 1909	WIN	0
MERIDAN CAPITAL INTERNATIONAL	2ND-AW-03		LONE ATOL	257,100	0.50	MTA 1809	NA	d
TEMOLENSHING TVLIATO NYMERING	10-conty6-aky		VIO MIL	\$40,000	0.50	AUG 9/09	NIA	
MERIDIAN CAPITAL INTERNATIONAL	AW-36ALIGGS- UL		AUG 1607	351,429	080	AUG 1600	NIA	0
Total								

Schedule "C"

Outstanding Timberline Options, Timberline Warrants and Timberline Securities

TIMBERLINE RESOURCES CORPORATION SUMMARY OF OPTIONS AND WARRANTS BY GRANT DATE AS OF MARCH 22, 2010

Options

Grant Date	Strike Price	Number of Options	Expiration Date
3/15/2006	0.75	333,333	3/14/2011
8/15/2006	0.75	95,000	8/14/2011
5/1/2007	3.00	214,584	5/1/2012
10/24/2007	3.40	180,000	10/24/2012
1/14/2008	4.05	165,000	1/14/2013
8/22/2008	2.48	400,000	8/22/2013
9/3/2008	2.05	175,000	9/3/2013
12/19/2008	0.33	1,936,665	12/19/2013
4/24/2009	0.36	100,000	4/24/2014
8/19/2009	0.43	200,000	8/19/2014
8/31/2009	0.57	1,071,500	8/31/2014
		4,871,082	Total Options

Warrants

Grant Date	Strike Price	Number of Warrants	Expiration Date
11/23/2009	1.50/1.75 *	1,697,638	3/25/2011

^{*} Exercisable at \$1.50 until 5/31/2010, then \$1.75 thereafter until expiration

SUPPORT AGREEMENT

This Support Agreement (this "Agreement"	") is made and entered into as of March 22, 2010, by
and among Timberline Resources Corporation, a	a company organized under the laws of the state of
Delaware ("Timberline"), and	(the "Shareholder"), with reference
to the following:	

RECITALS

- A. Timberline and Staccato Gold Resources Ltd., a company organized under the laws of the province of British Columbia ("Staccato"), are concurrently entering into that certain Arrangement Agreement, dated as of the date hereof (such agreement without further amendment or alterations being hereafter referred to as the "Arrangement Agreement"), pursuant to which Timberline will issue one share of common stock of Timberline ("Timberline Shares") and a cash payment, as stated in the Arrangement Agreement, in exchange for each seven common shares of Staccato ("Staccato Shares") held by the shareholders of Staccato (the "Arrangement").
- B. The Shareholder is an officer or director of Staccato and the beneficial holder of outstanding Staccato Shares or securities convertible into or exercisable to acquire Staccato Shares.
- C. As a condition to its execution of the Arrangement Agreement, Timberline has required that the Shareholder enter into this Agreement.
- D. The board of directors of Staccato has advised the Shareholder that it believes that the transactions contemplated by the Arrangement Agreement are in the best interests of the shareholders of Staccato.

NOW THEREFORE, on the basis of the foregoing recitals and in consideration of the mutual representations, warranties, covenants and agreements contained herein, the parties hereto agree as follows:

- 1. <u>Certain Definitions</u>. Capitalized terms used but not otherwise defined herein shall have the meanings ascribed thereto in the Arrangement Agreement. For all purposes of and under this Agreement, the following terms shall have the following respective meanings:
 - (a) "Acquisition Proposal" shall mean any tender offer or exchange offer or proposal to Staccato (including, without limitation, any proposal or offer to shareholders of Staccato) with respect to a Business Combination or involving the purchase of 20% or more of the outstanding voting securities of Staccato.
 - (b) "<u>Business Combination</u>" shall mean (a) a merger, consolidation, acquisition, scheme or other analogous arrangement in which Staccato is a constituent corporation or party and pursuant to which voting securities of Staccato are or may be exchanged for cash, securities or other property or (b) a sale of all or substantially all of the assets of Staccato and its subsidiaries.
 - (c) "Constructive Sale" means with respect to any security a short sale with respect to such security, entering into or acquiring an offsetting derivative contract with respect to such

- security, entering into or acquiring a futures or forward contract to deliver such security or entering into any other hedging or other derivative transaction that has the effect of materially changing the economic benefits and risks of ownership.
- (d) "Expiration Date" means the earlier to occur of (i) such date and time as the Arrangement Agreement shall have been validly terminated pursuant to the terms thereof or (ii) the Effective Time.
- (e) "Transfer" means, with respect to any security, the direct or indirect assignment, sale, transfer, tender, pledge, hypothecation, or the grant, creation or sufferage of a lien or encumbrance in or upon, or the gift, placement in trust, or the Constructive Sale or other disposition of such security (including transfers by testamentary or intestate succession or otherwise by operation of law) or any right, title or interest therein (including, but not limited to, any right or power to vote to which the holder thereof may be entitled, whether such right or power is granted by proxy or otherwise), or the record or beneficial ownership thereof, the offer to make such a sale, transfer, Constructive Sale or other disposition, and each agreement, arrangement or understanding, whether or not in writing, to effect any of the foregoing.
- 2. <u>Transfer Restrictions</u>. At all times during the period commencing with the execution and delivery of this Agreement and expiring on the Expiration Date, the Shareholder shall not, except in connection with the Arrangement or as the result of the death of the Shareholder, Transfer any of the Staccato Shares, or discuss, negotiate, make an offer or enter into an agreement, commitment or other arrangement with respect thereto, unless each person to which any of such Staccato Shares, or any interest in any of such Staccato Shares, is or may be Transferred shall have: (i) executed a counterpart of this Agreement, and (ii) agreed in writing to hold such Staccato Shares (or interest in such Staccato Shares) subject to all of the terms and provisions of this Agreement.

3. Right to Vote.

- As of the date hereof and for so long as this Agreement remains in effect (including as of (a) the date of the Staccato Meeting, which, for purposes of this Agreement, includes any adjournment or postponement thereof), except for this Agreement or as otherwise permitted by this Agreement, the Shareholder has full legal power, authority and right to vote all of the Staccato Shares then owned of record or beneficially by him, in favor of the approval and authorization of the Arrangement, the Arrangement Agreement and the other transactions contemplated thereby (collectively, the "Proposed Transaction") without the consent or approval of, or any other action on the part of, any other person or entity. Without limiting the generality of the foregoing, the Shareholder has not entered into any voting agreement (other than this Agreement) with any person or entity with respect to any of the Staccato Shares, granted any person or entity any proxy (revocable or irrevocable) or power of attorney with respect to any of the Staccato Shares, deposited any of the Staccato Shares in a voting trust or entered into any arrangement or agreement with any person or entity limiting or affecting his legal power, authority or right to vote the Staccato Shares on any matter. If the Shareholder is the beneficial owner, but not the record holder, of the Staccato Shares, the Shareholder agrees to take all actions necessary to cause the record holder and any nominees to vote all of the Staccato Shares in favor of the approval and authorization of the Proposed Transaction.
- (b) From and after the date hereof, except as otherwise permitted by this Agreement or prohibited by order of a court of competent jurisdiction, the Shareholder will not commit

any act that could restrict or otherwise affect his legal power, authority and right to vote all of the Staccato Shares then owned of record or beneficially by him. Without limiting the generality of the foregoing, except for this Agreement and as otherwise permitted by this Agreement, from and after the date hereof, the Shareholder will not enter into any voting agreement with any person or entity with respect to any of the Staccato Shares, grant any person or entity any proxy (revocable or irrevocable) or power of attorney with respect to any of the Staccato Shares, deposit any of the Staccato Shares in a voting trust or otherwise enter into any agreement or arrangement with any person or entity limiting or affecting the Shareholder's legal power, authority or right to vote the Staccato Shares in favor of the approval of the Proposed Transaction.

4. <u>Agreement to Vote Shares</u>

- (a) Prior to the Expiration Date, at every meeting of the stockholders of Timberline called, and at every adjournment or postponement thereof, and on every action or approval by written consent of the stockholders of Timberline, the Shareholder (in the Shareholder's capacity as such) shall appear at the meeting or otherwise cause the Staccato Shares to be present thereat for purposes of establishing a quorum and vote (i) in favor of approval of the Proposed Transaction, (ii) against the approval or adoption of any proposal made in opposition to, or in competition with, the Proposed Transaction, and (iii) against any of the following (to the extent unrelated to the Proposed Transaction): (A) any merger, consolidation or business combination involving Timberline or any of its subsidiaries other than the Proposed Transaction; (B) any sale, lease or transfer of all or substantially all of the assets of Timberline or any of its subsidiaries; (C) any reorganization, recapitalization, dissolution, liquidation or winding up of Timberline or any of its subsidiaries; or (D) any other action that is intended, or could reasonably be expected to, impede, interfere with, delay, postpone, discourage or adversely affect the consummation of the Proposed Transaction.
- 5. <u>No Solicitation</u>. The Shareholder, in his capacity as a Shareholder, shall not directly or indirectly, (i) solicit, initiate, encourage, induce or facilitate the making, submission or announcement of any Acquisition Proposal or take any action that could reasonably be expected to lead to an Acquisition Proposal, (ii) furnish any information regarding Staccato to any person in connection with or in response to an Acquisition Proposal or an inquiry or indication of interest that could reasonably be expected to lead to an Acquisition Proposal, (iii) engage in discussions or negotiations with any Person with respect to any Acquisition Proposal, (iv) approve, endorse or recommend any Acquisition Proposal or (v) enter into any letter of intent or similar document or any contract contemplating or otherwise relating to any Acquisition Proposal.
- 6. <u>Action in the Shareholder Capacity Only</u>. The Shareholder makes no agreement or understanding herein as director or officer of Staccato. The Shareholder signs solely in his capacity as a record holder and beneficial owner of Staccato Shares, and nothing herein shall limit or affect any actions taken in his capacity as an officer or director of Staccato.
- 7. <u>Additional Representations and Warranties of the Shareholder.</u> The Shareholder hereby represents and warrants to Timberline as follows: (i) the Shareholder is the beneficial or record owner of the Staccato Shares indicated on the signature page of this Agreement free and clear of any and all pledges, liens, security interests, claims, charges, restrictions, options or encumbrances; (ii) the Shareholder does not beneficially own any securities of Staccato other than the Staccato Shares and rights to purchase Staccato Shares set forth on the signature page of this Agreement; and (iii) the Shareholder has full power and authority to make, enter into and carry out the terms of this Agreement; and (iv) this

Agreement has been duly and validly executed and delivered by the Shareholder and constitutes a valid and binding agreement of the Shareholder enforceable against him in accordance with its terms.

- 8. <u>Conflicting Instruments</u>. The execution and delivery of this Agreement and the performance by the Shareholder of his agreements and obligations hereunder will not result in any breach or violation of or be in conflict with or constitute a default under any term of any agreement, judgment, injunction, order, decree, law, regulation or arrangement to which the Shareholder is a party or by which the Shareholder (or any of his assets) is bound, except for any such breach, violation, conflict or default which, individually or in the aggregate, would not impair or adversely affect the Shareholder's ability to perform his obligations under this Agreement or render inaccurate any of the representations made by him herein.
- 9. <u>Exchange of Shares; Waiver of Rights of Appraisal</u>. If the Arrangement is consummated, the Staccato Shares shall, pursuant to the terms of the Arrangement Agreement, be exchanged for the consideration provided in the Arrangement Agreement. The Shareholder hereby waives any rights of appraisal with respect to the Arrangement, or rights to dissent from the Arrangement, that such the Shareholder may have.
- 10. <u>Termination</u>. This Agreement shall terminate and be of no further force or effect whatsoever as of the Expiration Date.

11. Miscellaneous Provisions.

- (a) <u>Amendments, Modifications and Waivers</u>. No amendment, modification or waiver in respect of this Agreement shall be effective against any party unless it shall be in writing and signed by each party.
- (b) Entire Agreement. This Agreement and the Arrangement Agreement constitute the entire agreement among the parties to this Agreement and supersede all other prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof.
- (c) <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the Province of British Columbia.
- (d) WAIVER OF JURY TRIAL. EACH OF THE PARTIES IRREVOCABLY WAIVES ANY AND ALL RIGHTS TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING BETWEEN THE PARTIES ARISING OUT OF OR RELATING TO THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT.
- (e) <u>Attorneys' Fees</u>. In any action at law or suit in equity to enforce this Agreement or the rights of any of the parties hereunder, the prevailing party in such action or suit shall be entitled to receive a reasonable sum for its attorneys' fees and all other reasonable costs and expenses incurred in such action or suit.
- (f) <u>Assignment and Successors</u>. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, including, without limitation, the Shareholder's estate upon the death of the Shareholder, provided that except as otherwise specifically provided herein, neither this Agreement nor any of the rights, interests or obligations of the parties hereto

- may be assigned by any of the parties hereto without prior written consent of the other parties hereto.
- (g) <u>No Third Party Rights</u>. Nothing in this Agreement, express or implied, is intended to or shall confer upon any person (other than the parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- (h) <u>Cooperation</u>. The Shareholder agrees to cooperate fully with Timberline and to execute and deliver such further documents, certificates, agreements and instruments and to take such other actions as may be reasonably requested by Timberline to evidence or reflect the transactions contemplated by this Agreement and to carry out the intent and purpose of this Agreement.
- (i) <u>Severability</u>. If any provision of this Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Agreement will remain in full force and effect. Any provision of this Agreement held invalid or unenforceable only in part or degree will remain in full force and effect to the extent not held invalid or unenforceable.
- (j) <u>Time of Essence</u>. With regard to all dates and time periods set forth or referred to in this Agreement, time is of the essence.
- (k) Specific Performance; Injunctive Relief. The parties hereto acknowledge that Timberline shall be irreparably harmed and that there shall be no adequate remedy at law for a violation of any of the covenants or agreements of the Shareholder set forth in this Agreement. Therefore, the Shareholder hereby agrees that, in addition to any other remedies that may be available to Timberline, as applicable upon any such violation, such party shall have the right to enforce such covenants and agreements by specific performance, injunctive relief or by any other means available to such party at law or in equity without posting any bond or other undertaking.
- (l) Notices. All notices, consents, requests, claims, demands and other communications under this Agreement shall be in writing and shall be deemed given if (a) delivered to the appropriate address by hand or overnight courier (providing proof of delivery), or (b) sent by facsimile or e-mail with confirmation of transmission by the transmitting equipment confirmed with a copy delivered as provided in clause (a), in each case to the parties at the following address, facsimile or e-mail address (or at such other address, facsimile or e-mail address for a party as shall be specified by like notice): (i) if to Timberline, to the address, e-mail address or facsimile provided in the Arrangement Agreement, including to the persons designated therein to receive copies; and (ii) if to the Shareholder, to the Shareholder's address, e-mail address or facsimile shown below the Shareholder's signature on the last page hereof.
- (m) <u>Counterparts</u>. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument, and shall become effective when counterparts have been signed by each of the parties and delivered to the other parties; it being understood that all parties need not sign the same counterpart.

- (n) <u>Headings</u>. The headings contained in this Agreement are for the convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.
- (o) <u>Legal Representation</u>. This Agreement was negotiated by the parties with the benefit of legal representation and any rule of construction or interpretation otherwise requiring this Agreement to be construed or interpreted against any party shall not apply to any construction or interpretation thereof.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the parties hereto have duly executed this Shareholder Agreement as of the day and year first above written.

TIMBERLINE	RESOURCES	CORPORATION
By:		
SHAREHOLDER		
Ву:		
Name:		
Address:		
Telephone: ()		
Facsimile: ()		_
E-Mail Address:		
Staccato Shares E	Beneficially Own	ed:
	Staccato Shares	
Options to Acquir	re	_Staccato Shares
Other Securities Convertible Into or Exercisable for		
Staccato Shares		

PRESS RELEASE



Timberline Agrees to Acquire Staccato Gold with Potential Near-Term Nevada Gold Production at Lookout Mountain

March 23, 2010 – Coeur d'Alene, Idaho – **Timberline Resources Corporation (NYSE Amex: TLR)** ("Timberline") and **Staccato Gold Resources Ltd. (TSX.V: CAT)** ("Staccato") are pleased to announce that they have entered into a definitive agreement (the "Agreement") whereby Timberline will acquire, through a court-approved plan of arrangement, all of the issued and outstanding common shares of Staccato by means of a share exchange (the "Arrangement"). Under the Agreement, Staccato shareholders will receive one share of common stock of Timberline and \$0.0001 for every seven Staccato common shares held (the "Exchange Ratio"). This represents a value of approximately CND\$0.17 per Staccato common share based on the closing price of Timberline shares on the NYSE Amex on March 22, 2010, an 82% premium to the closing price of Staccato shares on the date prior to announcement of the Arrangement and a 76% premium to Staccato's 20-day volume weighted average closing price.

Executives of Timberline and Staccato will hold a conference call to discuss this acquisition at 4:30 pm Eastern Time (1:30 pm Pacific Time) tomorrow afternoon, March 24, 2010. Details regarding the conference call are provided below.

With the completion of this Arrangement, Timberline will acquire Staccato's South Eureka property which includes a drill-tested exploration portfolio and the advanced-stage Lookout Mountain project, located along Nevada's Battle Mountain – Eureka trend, and approximately \$5 million in cash.

The South Eureka property is comprised of several projects included within one of the largest exploration land packages in the Battle Mountain / Eureka Trend – approximately 15,000 acres. The property has identified exploration potential evidenced by historic workings and gold anomalies throughout the district. The flagship Lookout Mountain project features a defined section of mineralized material within a large scale structural corridor and numerous high-priority targets to test. Timberline's Butte Highlands Gold Project is funded to production by its joint venture partner; therefore, Timberline intends to focus its exploration efforts at Staccato's Lookout Mountain project.

Benefits for Timberline Shareholders

- Low cost acquisition of significant mineralized material base with near-term gold production potential at the Lookout Mountain project
- Large, drill-tested, highly prospective project portfolio on Nevada's Battle Mountain Eureka gold trend
- Cash infusion of approximately \$5 million for exploration and development, primarily at the Lookout Mountain project
- Addition of strategic assets in Nevada while maintaining an attractive share structure

Benefits for Staccato Shareholders

- Strong premium to recent Staccato market price and a fresh start toward realization of shareholder value
- Anticipated near-term gold production at the Butte Highlands joint venture where Timberline owns a 50% carried-to-production interest
- A vertically-integrated company with in-house capability, expertise, and strategic alliances to explore, drill, permit, develop, and produce and the corporate strength to advance Staccato's projects

- Two core drilling subsidiaries with positive cash flow and annual revenue over US\$17 million in the fiscal year ended September 30, 2009 will provide economical drilling for exploration and development
- Attractive corporate vehicle and share structure to accommodate additional acquisitions and growth

Timberline CEO Randal Hardy stated, "We are very pleased with this acquisition opportunity and the value it represents for our shareholders. We are focused on projects that offer near-term production potential with significant exploration upside, and we believe that Staccato's South Eureka properties provide an excellent fit. Combined with our development-stage Butte Highlands project and our core drilling businesses, the Staccato assets will advance our goal of positioning Timberline as an emerging junior gold producer."

Mr. Hardy added, "Following a thorough technical review, we believe that the Lookout Mountain project has potential for near-term open pit production with relatively modest capital expenditures. We believe our team is well-suited to advance the project. Our V-P Exploration, Paul Dircksen, has extensive gold exploration and mine development experience in Nevada and has contributed to the discovery of several gold deposits that later became operating mines. Our newest board member, Bob Martinez, adds extensive metallurgical knowledge and experience managing heap leach operations in Nevada. Paul, Bob, and our entire team are enthusiastic about acquiring Staccato and look forward to advancing the Lookout Mountain project."

Timberline's Chairman and V-P Exploration, Paul Dircksen, stated, "Staccato's South Eureka package lies within the Battle Mountain / Eureka Trend in Nevada, just a few miles from Barrick's two-million ounce Archimedes / Ruby Hill mine. Our internal studies indicate that the Lookout Mountain project hosts significant oxide gold mineralization amenable to low-cost, heap-leach recovery techniques, along with a smaller, high-grade, sulfide resource. Our plan is to drill in-fill and step-out holes to further define and expand known mineralized zones, conduct additional metallurgical testing, and complete an economic scoping study for a near-term production decision. We also plan to launch field programs at Windfall and other earlier-stage, high-potential targets. We are very excited to acquire the South Eureka property and believe that it is an excellent fit for our business plan and our team."

Staccato's President and CEO, Grant Ewing, said, "We are very pleased with the premium offer Timberline has made. Timberline is a near-term producer that has the expertise and the knowledge base to move our advanced stage Lookout Mountain project towards production - in addition to their own projects. We are impressed with the Timberline management team and believe that the combined assets of Staccato and Timberline represent enhanced value creation potential for Staccato shareholders."

Summary of Arrangement

The proposed arrangement between Timberline and Staccato is expected to be completed by way of a court-approved plan of arrangement whereby one share of common stock of Timberline and \$0.0001 will be exchanged for seven (7) Staccato common shares and Staccato will become a wholly-owned subsidiary of Timberline. Upon completion of the Arrangement, Timberline will have approximately 56 million common shares issued and outstanding, with 73 million shares on a fully diluted basis. Current Timberline shareholders will own approximately 74% of the shares outstanding and former Staccato shareholders will own approximately 26%. On a fully diluted basis, Timberline will be owned 71% by current Timberline shareholders and 29% by current Staccato shareholders. In conjunction with this Arrangement, Timberline will seek a listing of its shares of commons stock on the TSX Venture Exchange with the intent to have a listing in Toronto in addition to the NYSE Amex listing.

Pursuant to the Arrangement, the holders of a portion of the outstanding Staccato options and the holders of all Staccato warrants will be entitled to receive Timberline options and warrants at the Exchange Ratio and with corresponding changes to the exercise price based on the Exchange Ratio. The balance of the outstanding Staccato options will either be exchanged for Staccato shares prior to the closing or cancelled at closing.

Completion of the Arrangement is subject to regulatory approvals and a favorable vote of at least two-thirds of the holders of Staccato common shares voted at a special meeting of shareholders to be held on May 14, 2010 and a favorable vote of a majority of the holders of shares of Timberline common stock voted at a special meeting of shareholders to be held in May 2010.

The Arrangement is subject to customary closing conditions including the receipt of all necessary court and regulatory approvals, including the approval of the NYSE Amex and the TSX Venture Exchange. The Agreement provides for the payment of reciprocal break fees under certain conditions, and the Arrangement is expected to close by June 15, 2010.

The Arrangement has been unanimously approved by the Board of Directors of both Staccato and Timberline, and the management and directors of both companies have entered into support agreements. The Agreement contains a commitment from Staccato not to solicit or initiate discussions concerning alternative transactions to the proposed Arrangement.

Management Team and Board of Directors

There will be no change in the management or Board of Directors of Timberline following the completion of the Arrangement.

Conference Call

The conference call to discuss the Arrangement will be hosted by Paul Dircksen, Chairman and VP Exploration of Timberline, Randal Hardy, CEO of Timberline, and Grant Ewing, CEO of Staccato. They will be available for questions immediately following a presentation highlighting the value of the Arrangement for both companies' shareholders. The presentation that will be followed during the call will be available at Timberline's web site at www.timberline-resources.com.

Conference call details:

Toll free: (800) 214-0694 International: (719) 955-1425

Passcode: 713423

A link to a replay of the call will be available for two weeks on the same Timberline web site.

About Timberline Resources Corporation

Timberline is a diversified gold company with three complementary business units: a mine in development with anticipated gold production, an active exploration division, and two contract core drilling subsidiaries. Timberline is focused on the evaluation and the acquisition of advanced-stage exploration opportunities, with the potential for near-term development and production. Timberline formed a 50/50 joint venture with Highland Mining, LLC, an affiliate of Small Mine Development, at its royalty-free Butte Highlands Gold Project which commenced development in the summer of 2009 and has gold production targeted in the near term. The Management team has experience, depth, and a solid track record of achievement in building successful companies and discovering economic ore bodies. Timberline is listed on the NYSE Amex and trades under the symbol "TLR".

About Staccato Gold Resources Ltd.

Staccato Gold is a well funded junior gold exploration company with resources and advanced stage exploration assets located exclusively in Nevada. Staccato has assembled a portfolio of prospective gold prospects in the dominant gold trends in Nevada.

Contact Information

For Timberline: Randal Hardy, CEO Phone: 208.664.4859

Website: www.timberline-resources.com

Additional Information and Where to Find It

In connection with Timberline's and Staccato's solicitation of proxies with respect to the meeting of shareholders of each of Timberline and Staccato to be called with respect to the proposed Arrangement, Timberline will file a proxy statement with the Securities and Exchange Commission (the "SEC") and Staccato will file an information circular with certain regulatory authorities in Canada.

SHAREHOLDERS ARE ADVISED TO READ THE PROXY STATEMENT/INFORMATION CIRCULAR WHEN IT IS FINALIZED AND DISTRIBUTED TO SHAREHOLDERS BECAUSE IT WILL CONTAIN IMPORTANT INFORMATION.

Shareholders will be able to obtain a free-of-charge copy of the proxy statement/information circular (when available) and other relevant documents filed with the SEC from the SEC's website at http://www.sec.gov and with Canadian regulatory authorities from SEDAR at http://www.sedar.com. Shareholders of Timberline will also be able to obtain a free-of-charge copy of the proxy statement and other relevant documents (when available) by directing a request by mail or telephone to Timberline Resources Corporation, 101 E. Lakeside, Coeur d'Alene, Idaho 83814 or (208) 664-4859, or from Timberline's website,

www.timberline-resources.com. Shareholders of Staccato will also be able to obtain a free-of-charge copy of the information circular and other relevant documents (when available) by directing a request by mail or telephone to Staccato Gold Resources Ltd., Suite 2060, 777 Hornby Street, Vancouver, British Columbia, Canada V6Z 1T7 or (604) 682-8789 or from Staccato's website, www.staccatogold.com.

None of the securities anticipated to be issued pursuant to the Arrangement have been or will be registered under the Securities Act of 1933, as amended (the "U.S. Securities Act") or any state securities laws, and are anticipated to be issued in reliance upon available exemptions from such registration requirements under Section 3(a)(10) of the U.S. Securities Act and applicable exemptions under state securities laws. This press release does not constitute an offer to sell or the solicitation of an offer to buy any securities.

Certain statements in this press release relating to the proposed Arrangement and Timberline's exploration activities, project expenditures and business plans are "forward-looking statements" within the meaning of securities legislation. These statements include statements regarding completion of the Arrangement, the near and long term value of Timberline following the Arrangement, future production by Timberline, value to be realized by utilizing Timberline's drilling subsidiaries for anticipated exploration activities, Timberline's future exploration activities following the Arrangement, and future development at Timberline's and Staccato's properties. Timberline does not intend, and does not assume any obligation, to update these forward-looking statements. These forward-looking statements represent management's best judgment based on current facts and assumptions that management considers reasonable, including that the required approval will be obtained from the shareholders of Timberline or Staccato, that all third party regulatory and governmental approvals to the Arrangement will be obtained and all other conditions to completion of the Arrangement will be satisfied or waived. Timberline makes no representation that reasonable business people in possession of the same information would reach the same conclusions. Forward-looking statements involve known and unknown risks, uncertainties and other factors which may cause the actual results, performance or achievements of Timberline to be materially different from any future results, performance or achievements expressed or implied by the forward-looking statements. In particular, fluctuations in the price of gold could prevent Timberline from achieving its targets. Other factors are disclosed under the heading "Risk Factors", "Risks and Uncertainties" and elsewhere in Timberline's documents filed with the NYSE Amex and the United States Securities and Exchange Commission.